

Positivity of characteristic forms via pointwise universal push-forward formulæ

Dipartimento di Matematica "Guido Castelnuovo" Dottorato di Ricerca in Matematica – XXXIV Ciclo

Candidate Filippo Fagioli ID number 1842334

Thesis Advisor Prof. Simone Diverio

May 2022

Thesis defended on 12 May 2022 in front of a Board of Examiners composed by:

Prof. Gian Pietro Pirola (chairman) Prof. Arvid Perego Prof. Matei Toma

Positivity of characteristic forms via pointwise universal push-forward formulæ Ph.D. thesis. Sapienza – University of Rome

 $\ensuremath{\mathbb{C}}$ 2022 Filippo Fagioli. All rights reserved

This thesis has been typeset by ${\rm L\!AT}_{\rm E\!X}$ and the Sapthesis class.

Author's email: filippo.fagioli@uniroma1.it

Abstract

Given a Hermitian holomorphic vector bundle over a complex manifold, consider its flag bundles with the associated universal vector bundles endowed with the induced metrics. We show that the universal formula for the push-forward of a homogeneous polynomial in the Chern classes of the universal vector bundles also holds pointwise at the level of Chern forms in this Hermitianized situation.

As an application, we obtain the positivity of several polynomials in the Chern forms of Griffiths semipositive vector bundles not previously known. This gives new evidences towards a conjecture proposed by Griffiths, which has raised interest in the past as well as in recent years. This conjecture can be interpreted as a pointwise Hermitianized version of the Fulton–Lazarsfeld theorem on numerically positive polynomials for ample vector bundles.

Acknowledgments

There are many people to whom I owe a great deal of gratitude for having supported this project in different yet very important ways.

I felt and still feel honored for the unparalleled opportunity to engage with the study of Complex Geometry in my PhD programme at Università di Roma "La Sapienza" from such a unique venue and with an outstanding advisor, Simone Diverio. My deepest gratitude goes to him for his invaluable advice, constant support, and the time spent together on these stimulating topics. I find it difficult to imagine my PhD without his supervision.

I sincerely thank Ping Li and Christophe Mourougane for the careful review of my thesis and the very helpful comments.

I also wish to express my gratitude to my tutor, Domenico Fiorenza, for his generous support, and to Daniele Angella for his interest in my work and for having involved me in several events he organized.

I am very grateful to all the people that I have met during these beautiful years in Rome, especially my colleagues of the Department of Mathematics. In particular, I want to thank Andrea, Ángel, Fabio, Giuseppe, Massimiliano, and Nicola for their friendship. Having been part of all that has been simply superb.

Sono infinitamente grato ai miei genitori Andrea e Cecilia, che mi hanno sempre sostenuto, supportato (e sopportato) nelle difficoltà, a mio fratello Paolo, per il bellissimo legame che ci unisce, e a Tommaso, per la nostra storica amicizia.

Un ringraziamento speciale a zio Giovanni, che non mi ha mai fatto mancare i suoi preziosi consigli, a nonna Tosca e a tutta la mia famiglia per il continuo sostegno e incoraggiamento.

Infine, un grazie di cuore a Sara ... per tutto.

Contents

In	trod	action	vii
1	Pre	iminary notions	1
	1.1	Introductory notation	. 1
	1.2	Flag bundles	. 3
		1.2.1 Universal vector bundles	. 4
		1.2.2 Universal line bundles and their curvature	. 7
		1.2.3 Splitting of the tangent bundle	. 9
	1.3	Schur polynomials	. 9
		1.3.1 Generalized Schur classes and forms	. 11
	1.4	Positivity notions	. 11
		1.4.1 Positivity notions for differential forms	. 11
		1.4.2 Positivity notions for vector bundles	. 14
2	Cur	vature of universal bundles and push-forward formulæ	17
	2.1	Gysin formulæ à la Darondeau–Pragacz	. 17
	2.2	Push-forward formulæ for universal line bundles	. 20
		2.2.1 Curvature intrinsic expression: universal line bundles	. 20
		2.2.2 $$ Gys in formulæ for the curvature of universal line bundles $$.	. 22
	2.3	Push-forward formulæ for universal vector bundles	. 27
		2.3.1 Curvature of universal vector bundles	. 28
		2.3.2 Curvature intrinsic expression: universal vector bundles \ldots	. 32
		2.3.3 Gysin formulæ for the curvature of universal vector bundles .	. 34
	2.4	Alternative expression of the push-forward formulæ	. 37
3	Pos	tivity of characteristic forms for positive vector bundles	41
	3.1	About Griffiths' conjecture on positive polynomials	. 41
	3.2	A family of strongly positive characteristic forms	. 43
		3.2.1 Examples	. 46
	3.3	Other positive characteristic forms	. 49
		3.3.1 Positivity of c_2 in any rank $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$. 49
		3.3.2 Positivity of $c_1c_2 - c_3$ in rank 3	. 50
	3.4	State of the art, remarks and open questions	. 52
		3.4.1 A comparison on the positivity of Schur forms $\ldots \ldots \ldots$	53
		3.4.2 Concluding remarks	54
		3.4.3 Further related questions	. 55

A PARI/GP computations	57
Bibliography	59

Introduction

The main purpose of this thesis is the study of the characteristic differential forms associated to Hermitian holomorphic vector bundles over complex manifolds. More precisely, this thesis addresses two issues. The first one deals with the extension to the level of differential forms of certain push-forward formulæ for flag bundles valid in cohomology. The second one concerns the positivity of the characteristic differential forms arising from the Chern curvature of positive vector bundles.

The connection between these two issues consists in the use of the pointwise push-forward formulæ obtained to deduce the positivity of several characteristic forms associated to positive vector bundles.

Preliminaries

Let X be a complex manifold of dimension n and let $E \to X$ be a holomorphic vector bundle of rank $r \ge 2$. Once fixed a sequence ρ of dimensions

$$0 = \rho_0 < \rho_1 < \dots < \rho_{m-1} < \rho_m = r,$$

one can consider the (incomplete if m < r, or complete if m = r) flag bundle $\pi_{\rho} \colon \mathbb{F}_{\rho}(E) \to X$, which is naturally endowed with a filtration

$$(0) = U_{\rho,0} \subset U_{\rho,1} \subset \cdots \subset U_{\rho,m-1} \subset U_{\rho,m} = \pi_{\rho}^* E$$

of m + 1 tautological vector bundles where, for $0 \le l \le m$, the rank of $U_{\rho,l} \to \mathbb{F}_{\rho}(E)$ is ρ_l . Out of this filtration, one can form the universal quotient bundles

$$U_{\rho,l} \Big/_{U_{\rho,\ell}} \to \mathbb{F}_{\rho}(E), \quad 0 < \ell < l \le m.$$

Both tautological bundles and universal quotients are called *universal vector bundles* over $\mathbb{F}_{\rho}(E)$. Let $\mathcal{E}_1, \ldots, \mathcal{E}_N$ be an enumeration of all the universal vector bundles, where $N = \binom{m+1}{2}$. We denote by r_1, \ldots, r_N the corresponding ranks. Finally, for $1 \leq j \leq N$, consider the corresponding Chern classes $c_1(\mathcal{E}_j), \ldots, c_{r_j}(\mathcal{E}_j)$ in the cohomology of $\mathbb{F}_{\rho}(E)$.

Now, given a homogeneous polynomial F in $r_1 + \cdots + r_N$ variables of degree $d_{\rho} + k$, where d_{ρ} is the relative dimension of the proper holomorphic submersion π_{ρ} and $0 \le k \le n$, the proper push-forward

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1),\ldots,c_{\bullet}(\mathcal{E}_N))$$

gives a cohomology class in $H^{2k}(X)$ which of course needs to be a characteristic class for E. It is then a natural issue (which has been considered and settled by several authors in different degree of generality) to try to determine a closed formula to express this class more or less explicitly as a polynomial, call it Φ , in the Chern (or Segre) classes of E.

There is a vast literature concerning push-forward formulæ (which are also called $Gysin\ formulæ$) for flag bundles, and the approaches followed are several. For instance, Gysin formulæ for flag bundles are given by:

- [Qui69; Dam73; AC87] by using Grothendieck residues;
- [BS12; Tu17; Zie18] by using residues at infinity;
- [Bri96; PR97] by using symmetrizing operators;
- [JLP81] by using Schur functions.

See also the books [FP98; Man98].

In this thesis, we shall consider for our purposes the incarnation of such a formula given by Darondeau and Pragacz in [DP17]. For similar formulæ see [Ilo78]. We also shall use a Gysin formula for Grassmann bundles given in [KT15].

Pointwise push-forward formulæ for flag bundles

Alongside the cohomological situation described above, one can ask the analogue in the Hermitian setting as follows. Suppose that E is moreover endowed with a smooth Hermitian metric h. From the Chern curvature tensor $\Theta(E, h)$, let us consider the corresponding Chern forms on X defined, for $0 \le s \le r$, as

$$c_s(E,h) = \operatorname{tr}_{\operatorname{End}(\Lambda^s E)}\left(\bigwedge^s \frac{i}{2\pi}\Theta(E,h)\right).$$

By the Chern–Weil theory the form $c_s(E, h)$ represents the Chern class $c_s(E)$ of the vector bundle E.

The universal vector bundles \mathcal{E}_j 's considered before inherit, being sub-bundles of $\pi_{\rho}^* E$ or quotients of them, natural Hermitian metrics H_j 's. Thus, the classes $c_1(\mathcal{E}_j), \ldots, c_{r_j}(\mathcal{E}_j)$ now have special representatives $c_1(\mathcal{E}_j, H_j), \ldots, c_{r_j}(\mathcal{E}_j, H_j)$ given by the Chern forms of their induced Hermitian metrics.

Given the homogeneous polynomial F above, one can formally compute it using the $c_{\bullet}(\mathcal{E}_j, H_j)$'s as variables to get a closed $(d_{\rho} + k, d_{\rho} + k)$ -form on $\mathbb{F}_{\rho}(E)$, which can be pushed-forward on X via integration along the fibers to obtain a (k, k)-form

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N))$$

on X. Such a form is of course a special representative for the class

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1),\ldots,c_{\bullet}(\mathcal{E}_N)) = \Phi(c_{\bullet}(E)),$$

where Φ is the polynomial given by universal Gysin formulæ mentioned above.

Certainly, one can also formally evaluate Φ in the Chern forms of (E, h), thus obtaining another special representative, namely $\Phi(c_{\bullet}(E, h))$, for the cohomology class $\Phi(c_{\bullet}(E))$. Therefore, a priori, the forms $(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1, H_1), \ldots, c_{\bullet}(\mathcal{E}_N, H_N))$ and $\Phi(c_{\bullet}(E, h))$ differ by an error term which is an exact 2k-form.

The first central result of this thesis (see Theorem 2.18) can be now summarized in the following.

Theorem A. We have the equality of differential forms

 $(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N)) = \Phi(c_{\bullet}(E,h)).$

So, in fact, there is no error term at all. In other words, the universal Gysin formulæ to compute the push-forwards in cohomology from the flag bundle can be now used *verbatim* to compute pointwise the push-forwards for differential forms constructed from the Chern–Weil theory in the Hermitian situation.

A remarkable special case of Theorem A, for which we dedicate a separate description due to its applications in positivity as we explain below, occurs when we consider the m line bundles

$$Q_{\rho,l} := \det \left(\frac{U_{\rho,m-l+1}}{U_{\rho,m-l}} \right) \to \mathbb{F}_{\rho}(E), \quad 1 \le l \le m,$$

which are the determinants of the successive quotients of the tautological bundles.

The Chern classes $c_1(Q_{\rho,1}), \ldots, c_1(Q_{\rho,m})$ have special representatives denoted by $\Xi_{\rho,1}, \ldots, \Xi_{\rho,m}$, which are the Chern curvatures of the determinants of the natural Hermitian metrics induced by h mentioned before. Equivalently, these representatives are the first Chern forms of the Hermitian metrics induced on the successive quotients.

As before, by taking a homogeneous polynomial G in m variables of degree $d_{\rho} + k$, where $0 \le k \le n$, the universal Gysin formulæ for flag bundles provide us a polynomial Γ in the Chern classes of E such that the equality

$$(\pi_{\rho})_*G(c_1(Q_{\rho,1}),\ldots,c_1(Q_{\rho,m})) = \Gamma(c_{\bullet}(E))$$

between cohomology classes holds.

Moving to the representatives level, we prove the following result (see Theorem 2.10).

Theorem B. We have the equality of differential forms

$$(\pi_{\rho})_*G(\Xi_{\rho,1},\ldots,\Xi_{\rho,m}) = \Gamma(c_{\bullet}(E,h)).$$

In the special case of the projectivized bundle $\mathbb{P}(E)$ of lines (resp. $\mathbb{P}(E^{\vee})$ of hyperplanes) in E, which, in our notation, corresponds to the sequence ρ of dimensions (0, 1, r) (resp. (0, r - 1, r)), Theorem B generalizes previous results of [Mou04; Gul12; Div16].

More precisely, [Mou04, Proposition 6] proves, by explicit calculations, the push-forward identity in the case $G(\Xi_{\rho,\bullet})$ a power of the first Chern form of the line bundle $\mathcal{O}_{\mathbb{P}(E)}(1)$ (resp. $\mathcal{O}_{\mathbb{P}(E^{\vee})}(1)$) equipped with the induced metric. In such case, the push-forward gives the (signed) Segre forms of (E, h), which are special representatives for the Segre classes of E. Alternative proofs of the push-forward formula for projectivized bundles are given in [Gul12, Proposition 3.1] (whose approach is followed in this thesis) and in [Div16, Proposition 1.1] (by pointwise computations).

Theorem B also generalizes the recent result [Fin21, Theorem 3.18] which is a special case of our theorem but for complete flag bundles and where, in our notations, G is taken to be just a monomial with some specific decreasing degrees.

Although, as already mentioned, Theorem B is a particular case of Theorem A, we have chosen here to state them as two separate results by several reasons. Beside the fact that, chronologically, Theorem B came before Theorem A, the proof of the former is obtained by means of explicit computations, while the proof of the latter does not use explicit formulæ. This because, in the end, we do not necessarily need to perform explicitly integration along the fibers to obtain the push-forward formula. The second reason is that the proof of Theorem B is based on the curvature formulæ for universal line bundles over $\mathbb{F}_{\rho}(E)$ provided by Demailly in [Dem88a, Formula (4.9)]. However, another main result of this thesis is the calculation of the Chern curvature (at a point) of any universal vector bundle over $\mathbb{F}_{\rho}(E)$, and the explicit expression of these curvatures (which we provide in Theorem 2.12) are needed to prove Theorem A. The final reason is that we use some of the Gysin formulæ of Theorem B for applications in positivity theory as the following section explains.

Applications: positivity of characteristic forms

The last part of the thesis is devoted to an application of Theorems A and B to a positivity issue, in the same spirit of [Gul12], as follows.

Denote by T_X the holomorphic tangent bundle of X, and suppose that $(E, h) \to X$ is a *Griffiths semipositive* (resp. *Griffiths positive*) vector bundle. This means that for every $x \in X$, $v \in E_x$, $\tau \in T_{X,x}$ the Chern curvature tensor of (E, h) satisfy

$$\langle \Theta(E,h)_x \cdot v, v \rangle_h(\tau,\bar{\tau}) \ge 0$$

(resp. > 0, and = 0 if and only if v or τ is the zero vector).

Recall also that a holomorphic vector bundle $\mathcal{V} \to X$ of rank r is called *ample* if the tautological line bundle $\mathcal{O}_{\mathbb{P}(\mathcal{V}^{\vee})}(1) \to \mathbb{P}(\mathcal{V}^{\vee})$ is ample.

It is well known that a Griffiths positive vector bundle over a compact complex manifold is ample (the converse is not known in general, but it is a conjecture) and that a globally generated vector bundle (*i.e.*, a bundle whose fibers are spanned by the global holomorphic sections) can be equipped with a Hermitian metric which makes it Griffiths semipositive.

It is natural to expect that certain conditions of positivity on the vector bundle (such as ampleness or Griffiths positivity) impose, in turn, the positivity of objects that derive from it. This kind of positivity issues can be placed at both algebraic and differential geometric level and, in both ways, they were considered by Griffiths in the seminal paper [Gri69].

At the differential geometric level, in [Gri69, p. 247] it was asked (and expected) whether the *Schur forms*, and hence their positive linear combinations, are positive differential forms for Griffiths positive vector bundles. See Section 1.4.1 for definitions of the three main notions of positivity for differential forms. See also [HK74; Dem12].

We now give a more detailed description of the Griffiths' question, which is known also as the "Griffiths' conjecture on the positivity of Chern–Weil forms".

Fix $k \in \mathbb{N}$ and denote by $\Lambda(k, r)$ the set of all partitions of k by non-negative

integers $\leq r$, *i.e.*, the set of all $\sigma = (\sigma_1, \ldots, \sigma_k) \in \mathbb{N}^k$ such that

$$r \ge \sigma_1 \ge \cdots \ge \sigma_k \ge 0, \quad |\sigma| = \sum_{j=1}^k \sigma_j = k.$$

To $\sigma \in \Lambda(k, r)$ we associate the *Schur polynomial* $S_{\sigma} \in \mathbb{Z}[c_1, \ldots, c_r]$ of weighted degree 2k (we regard c_s as having degree 2s), which is defined as

$$S_{\sigma}(c_1,\ldots,c_r) := \det \left(c_{\sigma_i+j-i} \right)_{1 \le i,j \le r}$$

By convention, $c_0 = 1$ and $c_s = 0$ if s < 0 or s > r.

As $\sigma \in \Lambda(k, r)$ varies, the Schur polynomials form a basis for the Q-vector space of degree 2k weighted homogeneous polynomials in r variables, and the product of two Schur polynomials is a positive linear combination of Schur polynomials. Hence, Schur polynomials generate a positive convex cone closed under product, which is called *Schur cone*. Following [Gri69] we call it $\Pi(r)$ (however, remark that this is not exactly the positive cone considered by Griffiths, but they coincides a posteriori thanks to the work of [FL83]).

Taking the vector bundle $(E, h) \to X$ as before, we can formally evaluate the Schur polynomial S_{σ} on the Chern forms of (E, h), obtaining a characteristic differential form on X. This is the so called *Schur form* of (E, h) associated to σ ; we denote it by $S_{\sigma}(E, h)$. Clearly, $S_{\sigma}(E, h)$ represents the *Schur class* $S_{\sigma}(E)$, which is the cohomology class of X obtained by formally evaluating the polynomial S_{σ} in the Chern classes of E.

Significant examples of Schur classes (resp. forms) are:

- the Chern classes (resp. forms), corresponding to $\sigma = (k, 0, \dots, 0)$;
- classes (resp. forms) of type $c_1c_{k-1}-c_k$, corresponding to $\sigma = (k-1, 1, 0, \dots, 0)$;
- the signed Segre classes (resp. forms), corresponding to $\sigma = (1, \ldots, 1)$.

To fix the terminology, we call the *Schur cone* of E (resp. of (E, h)) the positive convex cone formed by all the polynomials in $\Pi(r)$ formally evaluated in the Chern classes (resp. forms) of E (resp. of (E, h)). We denote it by $\Pi(E)$ (resp. $\Pi(E, h)$).

Griffiths' conjecture (which we now state with a modification with respect to the original statement in [Gri69]) can be now summarized in the following.

Conjecture ([Gri69]). Given any Griffiths semipositive Hermitian holomorphic vector bundle $(E, h) \to X$ of rank r and a polynomial $P \in \Pi(r)$, then the differential form $P(c_{\bullet}(E, h)) \in \Pi(E, h)$ is positive (in the weak sense of differential forms).

In other words, the conjecture asserts that the cone $\Pi(E, h)$ consists of positive differential forms and that, in particular, the Schur forms $S_{\sigma}(E, h)$'s are positive.

Griffiths' conjecture was thus a differential, pointwise forerunner of the Fulton– Lazarsfeld theorem [FL83], which is a cohomological global statement, that characterizes precisely all numerically positive polynomials for ample vector bundles. The Fulton–Lazarsfeld theorem indeed characterizes them exactly as the positive linear combinations of Schur polynomials, thus extending the previous works [Kle69; BG71; Gie71; UT77]. See [DPS94, Theorem 2.5] for an extension of [FL83, Theorem 3.1] to nef vector bundles on compact Kähler manifolds.

Observe that a cohomology class which can be represented by a positive differential form is numerically positive. Thus, an answer in the affirmative to Griffiths' conjecture would give a stronger Fulton–Lazarsfeld-type statement under the stronger (but conjecturally equivalent) hypothesis of positivity in the sense of Griffiths.

Up to now, very little is known about this question beside the trivial case of (any power of) the first Chern form. Griffiths proved in [Gri69, Appendix to §5.(b)] that the second Chern form of a rank 2 Griffiths positive vector bundle is positive. However, see [Gri69, p. 247], Griffiths himself deemed difficult to adapt his proof to the general case.

In the last few years there has been a renewed interest around Griffiths' question, as important partial results have appeared in the literature. Guler proved in [Gul12, Theorem 1.1] the positivity of the already mentioned signed Segre forms.

Strengthening the hypotheses with stronger notions of positivity, such as (dual) Nakano semipositivity (see for instance [Dem12] for the definitions), the conjecture was proven in [Li21, Proposition 3.1] and in [Fin21, Theorem 1.1]. It is also worth mentioning that Griffiths himself proved his own conjecture in [Gri69, Proof of Theorem D] for globally generated vector bundles. However, the problem still remains open in the general case. See also, for instance, [Div16; Pin18; RT21; Xia22] for further related results.

In Section 3.4.1 we provide an overview on the state of the art of Griffiths' conjecture, focusing, in particular, on the recent progress. Primarily, the purpose of such overview is to clarify which types of positivity (among those in Definition 1.15) have been obtained for the Schur forms of positive vector bundles. Indeed, the terminology used in the literature is currently not always standard, and sometimes may lead to some confusion.

Here, as a consequence of our Theorem B, we are able to establish the positivity of several new positive linear combinations of Schur forms, cf. Section 3.2. Namely, thanks to the curvature formulæ given in [Dem88a, Formula (4.9)] coupled with our universal pointwise push-forward formulæ, we obtain the following result (see Theorem 3.6).

Theorem C. In the notation as above, let $(E, h) \to X$ be Griffiths semipositive. Given a weight $\mathbf{a} \in \mathbb{Z}^m$ such that $a_1 \geq \cdots \geq a_m \geq 0$, we have that the characteristic forms

$$(\pi_{\rho})_*(a_1 \Xi_{\rho,1} + \cdots + a_m \Xi_{\rho,m})^{\wedge (d_{\rho}+k)}$$

are strongly positive (k, k)-forms on X, and moreover positive linear combinations of Schur forms of (E, h).

This confirms Griffiths' conjecture (in even a stronger way) for those differential forms belonging to the sub-cone, which we denote by $\mathcal{F}(E,h)$, of $\Pi(E,h)$ spanned by all the possible wedge products of all the possible push-forwards of type as in Theorem C (see Section 3.2.1 for several concrete examples).

The sub-cone $\mathcal{F}(E,h)$ contains in particular the signed Segre forms, therefore Theorem C generalizes [Gul12, Theorem 1.1]. Observe also that it is in some sense more natural to obtain that these forms are strongly positive rather than merely positive. This is because, as already said, polynomials in $\Pi(r)$ are stable under products and so do strongly positive forms, while a product of positive forms is not necessarily still positive.

As a byproduct of the proof of Theorem C, we also obtain a partial answer to a conjecture raised in [Xia22, Conjecture 1.4]. It asks whether every Schur class $S_{\sigma}(\mathcal{V})$ of an ample vector bundle \mathcal{V} over a projective variety X does admit a positive representative, and [Xia22, Theorem A] gives an affirmative answer if $|\sigma| = \dim X - 1$. Our result (see Theorem 3.8) shows that for each $\mathbf{a} \in \mathbb{Z}^m$ that satisfy the condition $a_1 \geq \cdots \geq a_m \geq 0$, the cohomology classes

$$(\pi_{\rho})_* (a_1 c_1(Q_{\rho,1}) + \dots + a_m c_1(Q_{\rho,m}))^{d_{\rho}+k}$$

belong the Schur cone $\Pi(E)$ and admit a strongly positive representative.

Afterwards, we focus on some differential forms which in general do not belong to the sub-cone $\mathcal{F}(E, h)$.

First, we show how to obtain the positivity of the second Chern form $c_2(E, h)$ in any rank, if (E, h) is Griffiths positive. This can be easily obtained by adapting the above mentioned Griffiths' result on the positivity of $c_2(E, h)$ in rank 2. At the best of our knowledge, this was first observed by Guler in [Gul06], but just as a statement without proof. Here we give full details and, furthermore, we extend such result to the semipositive context in Theorem 3.11.

Subsequently, we focus on the Griffiths' conjecture in rank 3, for which we prove the following result (see Theorem 3.12).

Theorem D. In the notation as above, let $(E,h) \to X$ be Griffiths semipositive of rank 3. Then the Schur form

$$S_{(2,1,0)}(E,h) = c_1(E,h) \wedge c_2(E,h) - c_3(E,h)$$

is a positive (3,3)-forms on X.

The starting point to obtain this result is to consider a variant of Theorem A (which we establish in Formula (2.19)) in order to get the form $S_{(2,1,0)}(E,h)$ as a push-forward from the complete flag bundle $\mathbb{F}(E)$; cf. with Proposition 2.21 where we prove that each Schur form can be obtained as a push-forward. But, at the best of our knowledge, we are not able to deduce the positivity of this form instantly from that push-forward. Hence, our idea is to factorize through the projectivized bundle $\mathbb{P}(E)$. In this way, we find that $S_{(2,1,0)}(E,h)$ can be obtained also by pushing forward a positive form on $\mathbb{P}(E)$, leading us to prove Theorem D.

As a consequence of Theorem D, we find that the following chain of inequalities

$$c_1(E,h)^{\wedge 3} \ge c_1(E,h) \wedge c_2(E,h) \ge c_3(E,h)$$

holds for Griffiths semipositive vector bundles of rank 3 over complex manifolds (see Corollary 3.16). Similar inequalities have appeared, for instance, in [Li21, Theorem 3.2] if h is dual Nakano semipositive, and they give a metric counterpart of some well-known inequalities between Chern numbers valid for nef vector bundles on compact Kähler manifolds (see, [FL83; DPS94; LZ20] and [Li21, Remark 3.3]). In

the setting of Theorem D, if X is compact (not necessarily Kähler) of dimension 3, it follows from the positivity of $S_{(2,1,0)}(E,h)$ that the Chern numbers of E satisfy the relation

$$c_1^{\ 3} \ge c_1 c_2 \ge c_3.$$

Certainly, it is clear that any differential form in the cone $\mathcal{F}(E, h)$, for which we proved the strong positivity in Theorem C, gives certain inequalities between Chern forms/numbers. However, we have preferred here to highlight the chain of inequalities coming from $S_{(2,1,0)}(E, h)$ since it was previously known, with different assumptions on E and X, in works such as [DPS94; LZ20].

In the last part of the thesis (see Section 3.4) we collect some concluding remarks and open questions which may serve for further developments of this topic.

Concluding note

Some of the main results of this thesis are next to publication. More precisely, Theorem B and Theorem C are respectively the main theorem and main application of the author's paper [DF22] joint with S. Diverio. Theorem D is the main result of the paper [Fag22] by the author. Finally, the curvature computations of the universal vector bundles in Chapter 2 and Theorem A are results obtained by the author shortly before the redaction of this thesis. They are currently in preparation in view of a future publication.

Chapter 1 Preliminary notions

In this chapter we introduce the concepts necessary to state the main results of this thesis. Namely, in Section 1.2 we recall the notions of flag bundle ([Dem88a; Dem88b; Ful98]) and universal vector bundles, which are the main objects of our results in Chapter 2. Then, in Section 1.3 we give the definition of Schur polynomial ([FL83; DPS94; Ful97]) and some notions related to Schur functions ([DP17]). Finally, in Section 1.4 we recall some of the positivity notions for differential forms ([HK74; Dem12]) and vector bundles ([Gri69; Laz04; Dem12; Fin21]) needed for our results in Chapter 3. Several further references are cited during the exposition, in particular for the proofs of the results that we recall.

1.1 Introductory notation

Let X be a complex manifold of dimension n.

If $E \to X$ is a complex vector bundle and $0 \le p, q \le n$, then $\mathcal{A}^{p,q}(X, E)$ stands for the space $C^{\infty}(X, \Lambda^{p,q}T_X^{\vee} \otimes E)$ of differential (p, q)-forms on X with values in E. In particular, $\mathcal{A}^{p,q}(X)$ denotes the space $C^{\infty}(X, \Lambda^{p,q}T_X^{\vee})$ of differential (p, q)-forms on X. Similarly, $\mathcal{A}^k(X)$ stands for the space of differential k-forms on X.

In this section we suppose that the vector bundle $E \to X$ is holomorphic of rank r and equipped with a Hermitian metric h. We denote by

$$\Theta(E,h) \in \mathcal{A}^{1,1}(X, \operatorname{End}(E))$$

the Chern curvature tensor of the Hermitian vector bundle (E, h), where

$$\operatorname{End}(E) \cong E^{\vee} \otimes E$$

is the endomorphisms bundle of E and E^{\vee} is the dual bundle of E. Here we prefer to use the symbol \cdot^{\vee} for the dual in order to avoid confusion with the symbol \cdot^* of pull-back.

Recall that, out of the Chern curvature $\Theta(E, h)$ of the Hermitian vector bundle (E, h), one can construct the *Chern forms* of (E, h) on X. They are defined, for $0 \le s \le r$, as

$$c_s(E,h) = \operatorname{tr}_{\operatorname{End}(\Lambda^s E)}\left(\bigwedge^s \frac{i}{2\pi}\Theta(E,h)\right).$$

By the Chern–Weil theory the form $c_s(E,h) \in \mathcal{A}^{s,s}(X)$ is *d*-closed, real and represents the Chern class $c_s(E) \in H^{2s}(X)$ of the vector bundle E.

Fix a point $x_0 \in X$ and local holomorphic coordinates $z = (z_1, \ldots, z_n)$ on an open set of X centered at x_0 . Recall that a local holomorphic frame (e_1, \ldots, e_r) of E centered at x_0 is called a *normal coordinate frame* at x_0 (see, [Dem12, (12.10) Proposition]) if

$$\langle e_{\alpha}(z), e_{\beta}(z) \rangle_h = \delta_{\alpha\beta} - \sum_{1 \le j,k \le n} c_{jk\alpha\beta} z_j \bar{z}_k + \mathcal{O}(|z|^3),$$

where the $c_{ik\alpha\beta}$'s are the coefficients of the Chern curvature tensor

$$\Theta(E,h)_{x_0} = \sum_{1 \le \alpha, \beta \le r} \sum_{1 \le j,k \le n} c_{jk\alpha\beta} \, dz_j \wedge d\bar{z}_k \otimes e_\alpha^{\vee} \otimes e_\beta$$

expressed with respect to the frame (e_1, \ldots, e_r) .

Since $i\Theta(E, h)$ is a (1, 1)-form with values in the bundle Herm(E, h) of Hermitian endomorphisms of E, we have the symmetry relation $\bar{c}_{jk\alpha\beta} = c_{kj\beta\alpha}$.

In this notation, the (β, α) -entry of the matrix associated to $\Theta(E, h)_{x_0}$ with respect to (e_1, \ldots, e_r) in the coordinate open set considered is the (1, 1)-form

$$\Theta_{\beta\alpha} := \sum_{1 \le j,k \le n} c_{jk\alpha\beta} \, dz_j \wedge d\bar{z}_k$$

Push-forward given by integration along fibers

Let $\pi: S \to X$ be a proper holomorphic submersion of complex manifolds with relative dimension s. Then, for $k = 0, \ldots, \dim_{\mathbb{R}} X$ the map

$$\pi_*\colon \mathcal{A}^{2s+k}(S) \to \mathcal{A}^k(X)$$

is the *push-forward* through π , which is given by integration along the fibers of the submersion. Given a differential form η on S, the push-forward can be computed at $x \in X$ by locally splitting the variables of the fiber $\pi^{-1}(x)$ as (x, y) in such a way that

$$\pi_*\eta(x) = \int_{y \in \pi^{-1}(x)} \eta(x, y),$$

where the right hand side stands for the integral of $\eta(x, y)$ performed only on those differentials related to the variable y. For precise definitions, properties and results about push-forwards of differential forms we refer, mainly, to [BT82; Dem12]. We just recall here the so-called *projection formula*, which relates push-forwards and pull-backs. For each differential form α on S and β on X, the following equality

$$\pi_*\left(\alpha \wedge \pi^*\beta\right) = \pi_*\alpha \wedge \beta$$

holds.

We also use the same notation for the map

$$\pi_* \colon H^{2s+k}(S) \to H^k(X)$$

induced in cohomology by integration along fibers of π . These maps are also called *Gysin* homomorphisms (see, for instance, [Ful98]).

1.2 Flag bundles

This section mainly follows [Dem88a; Dem88b]. See also, for instance, [Ful97; Ful98] for further references on flag manifolds and bundles.

Let X be a complex manifold of dimension n and let $E \to X$ be a holomorphic vector bundle of rank r. Given a natural number m > 0, fix a sequence of integers $\rho = (\rho_0, \ldots, \rho_m)$ of the form

$$0 = \rho_0 < \rho_1 < \dots < \rho_l < \dots < \rho_{m-1} < \rho_m = r.$$

We call such a ρ a dimension sequence.

The flag bundle of type ρ associated to E is the holomorphic fiber bundle

$$\pi_{\rho} \colon \mathbb{F}_{\rho}(E) \to X$$

where the fiber over $x \in X$ is the flag manifold $\mathbb{F}_{\rho}(E_x)$, whose points $\mathbf{f}_{x,\rho}$ are the flags of the form

$$\{0_x\} = V_{x,\rho_0} \subset V_{x,\rho_1} \subset \cdots \subset V_{x,\rho_l} \subset \cdots \subset V_{x,\rho_{m-1}} \subset V_{x,\rho_m} = E_x$$

where, for each $0 \leq l \leq m$, dim_C $V_{x,\rho_l} = \rho_l$.

Over $\mathbb{F}_{\rho}(E)$ we have a tautological filtration

$$(0) = U_{\rho,0} \subset U_{\rho,1} \subset \cdots \subset U_{\rho,l} \subset \cdots \subset U_{\rho,m-1} \subset U_{\rho,m} = \pi_{\rho}^* E \tag{1.1}$$

of vector sub-bundles of $\pi_{\rho}^* E$, where, for every $0 \leq l \leq m$, the fiber of $U_{\rho,l}$ over the point $(x, \mathbf{f}_{x,\rho}) \in \mathbb{F}_{\rho}(E)$ is V_{x,ρ_l} . Therefore, the vector bundle $U_{\rho,l} \to \mathbb{F}_{\rho}(E)$ has rank ρ_l . Finally, denote by d_{ρ} the relative dimension of the proper holomorphic submersion π_{ρ} .

Example 1.1. If ρ is the complete sequence $(0, 1, \ldots, r - 1, r)$, then $\mathbb{F}_{\rho}(E)$ is the complete flag bundle associated to E. Observe that this occurs when m = r. In this case, we shall drop the subscript ρ simply writing

$$\pi\colon \mathbb{F}(E)\to X$$

and denoting by d the relative dimension of π . Accordingly, for $0 \leq l \leq r$, the tautological filtration (1.1) is written as

$$(0) = U_0 \subset U_1 \subset \cdots \subset U_l \subset \cdots \subset U_{m-1} \subset U_r = \pi^* E.$$

Now, fix $x \in X$ and set $V := E_x$. By using the description of the flag manifold $\mathbb{F}_{\rho}(V)$ as a homogeneous spaces (see, for instance, [Dem88a; Dem88b; Ful97]), we compute the relative dimension d_{ρ} of the flag bundle $\mathbb{F}_{\rho}(E)$ in the following.

Proposition 1.2. For a dimension sequence $\rho = (\rho_0, \ldots, \rho_m)$, it holds that

$$d_{\rho} = \frac{r^2 - \sum_{j=1}^{m} (\rho_j - \rho_{j-1})^2}{2}.$$
(1.2)

In particular, if ρ is the complete sequence, it holds that

$$d = \frac{r(r-1)}{2}.$$
 (1.3)

Proof. Let $\mathbf{f}_{\rho} \in \mathbb{F}_{\rho}(V)$ be the flag identified by a basis $(\varphi_1, \ldots, \varphi_r)$ of V as

$$\{0_V\} \subset V_{\rho_1} \subset \cdots \subset V_{\rho_l} = \langle \varphi_{r-\rho_l+1}, \varphi_{r-\rho_l+2}, \dots, \varphi_r \rangle \subset \cdots \subset V_{\rho_{m-1}} \subset V.$$

In order to emphasize the basis $(\varphi_1, \ldots, \varphi_r)$ defining the flag \mathbf{f}_{ρ} , it is common to use the notation $[\varphi_1, \ldots, \varphi_r]_{\rho}$.

The general linear group $\operatorname{GL}(r, \mathbb{C})$ acts on the right of $\mathbb{F}_{\rho}(V)$ as follows:

$$([\varphi_1,\ldots,\varphi_r]_{\rho},A)\mapsto \left[\sum_{\lambda=1}^r a_{\lambda 1}\varphi_{\lambda},\ldots,\sum_{\lambda=1}^r a_{\lambda r}\varphi_{\lambda}\right]_{\rho}$$

where the $a_{\lambda\mu}$'s are the entries of the matrix in A acting on the flag \mathbf{f}_{ρ} .

The stabilizer of \mathbf{f}_{ρ} with respect to this action is the parabolic subgroup P_{ρ} of matrices $(a_{\lambda\mu})$ with $a_{\lambda\mu} = 0$ for all λ, μ such that there is an integer $\ell = 1, \ldots, m-1$ with $\lambda \leq r - \rho_{m-\ell} < \mu$. Note that if ρ is the complete sequence then P_{ρ} is the Borel subgroup of lower triangular matrices. Since the right action of $\operatorname{GL}(r, \mathbb{C})$ is transitive on the flag manifold, we have a diffeomorphism $\mathbb{F}_{\rho}(V) \cong \operatorname{GL}(r, \mathbb{C})/P_{\rho}$ and it holds that

$$\dim \mathbb{F}_{\rho}(V) = \dim \operatorname{GL}(r, \mathbb{C}) - \dim P_{\rho}$$

For a matrix $(a_{\lambda\mu}) \in P_{\rho}$, the above condition for which $a_{\lambda\mu} = 0$ gives, for each $j = 1, \ldots, m-1$ and $s = j, \ldots, m-1$, an upper block of zeros in $(a_{\lambda\mu})$ of size $(\rho_j - \rho_{j-1}) \times (\rho_{s+1} - \rho_s)$. In order to compute $d_{\rho} = \dim \mathbb{F}_{\rho}(V)$, we have to multiply the number of rows and columns of each upper block of zeros in $(a_{\lambda\mu})$ and then sum all these products together. Hence,

$$d_{\rho} = \sum_{j=1}^{m-1} \left[(\rho_j - \rho_{j-1}) \left(\sum_{s=j}^{m-1} (\rho_{s+1} - \rho_s) \right) \right]$$

= $\sum_{j=1}^{m} [(\rho_j - \rho_{j-1})(r - \rho_j)]$
= $r \sum_{j=1}^{m} \rho_j - r \sum_{j=1}^{m} \rho_{j-1} - \sum_{j=1}^{m} (\rho_j^2 - \rho_j \rho_{j-1})$
= $r^2 + r \left(\sum_{j=1}^{m-1} \rho_j - \sum_{j=1}^{m} \rho_{j-1} \right) - \frac{1}{2} \left(r^2 + \sum_{j=1}^{m} (\rho_j - \rho_{j-1})^2 \right)$
= $r^2 - r \rho_0 - \frac{r^2}{2} - \frac{1}{2} \sum_{j=1}^{m} (\rho_j - \rho_{j-1})^2$

and we get Formula (1.2). Formula (1.3) follows immediately.

1.2.1 Universal vector bundles

The tautological filtration (1.1) gives $\binom{m+1}{2}$ universal vector bundles over $\mathbb{F}_{\rho}(E)$, which are the quotients $U_{\rho,l}/U_{\rho,\ell}$, for each $0 \leq \ell < l \leq m$. In other words, these are the tautological vector bundles already introduced (we are not considering here the vector bundle of rank 0) and all their possible quotients.

Example 1.3. Set m = 2. If ρ is the sequence (0, 1, r) then $\mathbb{F}_{\rho}(E)$ equals the projective bundle of lines in E, which we denote by $\mathbb{P}(E)$. In this case, the tautological filtration (1.1) consists on one proper sub-bundle only, namely $U_{(0,1,r),1}$, which equals, by definition, the tautological line bundle $\mathcal{O}_{\mathbb{P}(E)}(-1)$. Given that $U_{\rho,2}/U_{\rho,1}$ is the only quotient given by the filtration (1.1), we have that $\mathbb{P}(E)$ has three universal vector bundles. They form the tautological short exact sequence

$$0 \to \underbrace{\mathcal{O}_{\mathbb{P}(E)}(-1)}_{U_{\rho,1}} \hookrightarrow \underbrace{\pi_{\rho}^{*}E}_{U_{\rho,2}} \twoheadrightarrow \underbrace{\pi_{\rho}^{*}E/\mathcal{O}_{\mathbb{P}(E)}(-1)}_{U_{\rho,2}/U_{\rho,1}} \to 0$$
(1.4)

over the projective bundle $\mathbb{P}(E)$.

Remark 1.4. Suppose that E is equipped with a Hermitian metric h. The pullback metric π_{ρ}^*h on π_{ρ}^*E endows the vector bundle $U_{\rho,l}$ with the restriction metric $h_{\rho,l} := \pi_{\rho}^*h\Big|_{U_{\rho,l}}$. Consequently, the quotient bundle $U_{\rho,l}/U_{\rho,\ell}$ is equipped with the Hermitian metric given by the quotient of the metrics $h_{\rho,l}$ and $h_{\rho,\ell}$. When needed, we denote such a Hermitian metric by $H_{\rho,(l,\ell)}$.

Definition 1.5. Let $\rho = (\rho_0, \ldots, \rho_m)$ and $\tau = (\tau_0, \ldots, \tau_{\tilde{m}})$ be two dimension sequences. We say that ρ is greater than or equal to τ , and we write $\rho \geq \tau$, if:

- (a) $m \geq \tilde{m};$
- (b) for each $\tilde{l} \in \{0, \dots, \tilde{m}\}$ there is a index $l \in \{0, \dots, m\}$ for which $\rho_l = \tau_{\tilde{l}}$.
- In addition, if
- (c) $\rho \neq \tau$;

we write $\rho > \tau$.

Given $\rho > \tau$, it is useful to denote by

$$\pi^{\rho}_{\tau} \colon \mathbb{F}_{\rho}(E) \to \mathbb{F}_{\tau}(E)$$

the natural forgetful projection between flag bundles. If $\rho = \tau$, then π_{ρ}^{ρ} is simply the identity map of $\mathbb{F}_{\rho}(E)$.

By construction, $\pi_{\tau} \circ \pi_{\tau}^{\rho} = \pi_{\rho}$, *i.e.*, the diagram

$$\mathbb{F}_{\rho}(E) \xrightarrow{\pi_{\tau}^{\rho}} \mathbb{F}_{\tau}(E) \\
\xrightarrow{\pi_{\rho}} \xrightarrow{\chi} \pi_{\tau} \\
X$$
(1.5)

is commutative.

Moreover, if $\rho \ge \tau \ge \sigma$ then $\pi^{\tau}_{\sigma} \circ \pi^{\rho}_{\tau} = \pi^{\rho}_{\sigma}$, *i.e.*, there is a commutative diagram



of projections between flag bundles over X.

Remark 1.6. Given the projection $\pi_{\tau}^{\rho} \colon \mathbb{F}_{\rho}(E) \to \mathbb{F}_{\tau}(E)$, suppose that there is a index \tilde{l} such that $\rho_{l} = \tau_{\tilde{l}}$. Then, it is straightforward to note that the tautological vector bundle $U_{\rho,l} \to \mathbb{F}_{\rho}(E)$ is the pull-back through π_{τ}^{ρ} of the bundle $U_{\tau,\tilde{l}} \to \mathbb{F}_{\tau}(E)$. Moreover, it is clear that the metric $h_{\rho,l}$ coincides with the pull-back metric $(\pi_{\tau}^{\rho})^{*}(h_{\tau,\tilde{l}})$. Indeed,

$$(\pi_{\tau}^{\rho})^{*}(h_{\tau,\tilde{l}}) = (\pi_{\tau}^{\rho})^{*}\left(\pi_{\tau}^{*}h|_{U_{\tau,\tilde{l}}}\right) = (\pi_{\tau}^{\rho})^{*}\pi_{\tau}^{*}h|_{(\pi_{\tau}^{\rho})^{*}U_{\tau,\tilde{l}}} = \pi_{\rho}^{*}h|_{U_{\rho,l}} = h_{\rho,l}.$$

In addition, for $0 < \ell < l$, if there is a $\tilde{\ell}$ such that $\rho_{\ell} = \tau_{\tilde{\ell}}$, then $U_{\rho,l}/U_{\rho,\ell} \to \mathbb{F}_{\rho}(E)$ is the pull-back through π_{τ}^{ρ} of the quotient $U_{\tau,\tilde{l}}/U_{\rho,\tilde{\ell}} \to \mathbb{F}_{\tau}(E)$. In this case, the Hermitian metric $H_{\rho,(l,\ell)}$ is the pull-back through π_{τ}^{ρ} of the metric $H_{\tau,(\tilde{l},\tilde{\ell})}$.

In particular, we emphasize the following.

Remark 1.7. Choose $m \ge 2$, and let ρ be a dimension sequence. Fix an index $l = 1, \ldots, m-1$ and set $s := \rho_l$. Since $\rho \ge (0, s, r)$, we have of course the forgetful projection

$$\pi_s^{\rho} \colon \mathbb{F}_{\rho}(E) \to \mathbb{G}_s(E)$$

on the Grassmannian bundle $\mathbb{G}_s(E)$ of s-planes in E.

By Remark 1.6, $U_{\rho,l}$ is the pull-back through π_s^{ρ} of the tautological *s*-plane bundle over $\mathbb{G}_s(E)$ denoted by γ_s . Moreover, the Hermitian metric $h_{\rho,l}$ is the pull-back of the obvious restriction metric h_s on γ_s induced by *h*. Similarly, $(\pi_{\rho}^*E/U_{\rho,l}, H_{\rho,(m,l)})$ is the pull-back through π_s^{ρ} of the universal quotient bundle Q_s over $\mathbb{G}_s(E)$, equipped with the quotient metric H_s .

Finally, if m > 2 and $0 < \ell < l < m$, let **s** be the sequence $(0, s_1, s_2, r)$ with $s_1 = \rho_{\ell}, s_2 = \rho_l$. Given that $\rho \ge \mathbf{s}$, we have that $U_{\rho,l}/U_{\rho,\ell}$ is the pull-back through the map $\pi_{\mathbf{s}}^{\mathbf{s}}$ of the unique quotient $Q_{\mathbf{s}} := U_{\mathbf{s},2}/U_{\mathbf{s},1}$ of proper sub-bundles over the flag bundle $\mathbb{F}_{\mathbf{s}}(E)$. As always, $Q_{\mathbf{s}}$ is equipped with the obvious quotient metric which we denote by $H_{\mathbf{s}}$, and $(\pi_{\mathbf{s}}^{\rho})^* H_{\mathbf{s}} = H_{\rho,(l,\ell)}$.

The observations in Remark 1.7 are the starting point for the proof of Theorem 2.12.

Local frames of universal vector bundles

Given a point $x_0 \in X$ and a dimension sequence ρ , fix local holomorphic coordinates z centered at x_0 and a flag $\mathbf{f}_0 \in \mathbb{F}_{\rho}(E_{x_0})$. Let (e_1, \ldots, e_r) be a local normal frame of E at x_0 . As the action of the general linear group is transitive on $\mathbb{F}_{\rho}(E_{x_0})$, we assume that the flag

$$\{0_{x_0}\} \subset \cdots \subset \operatorname{Span}\{e_{r-\rho_l+1}(x_0), e_{r-\rho_l+2}(x_0), \dots, e_r(x_0)\} \subset \cdots \subset E_{x_0}$$

coincides with \mathbf{f}_0 .

The basis $(e_1(z), \ldots, e_r(z))$ gives affine coordinates $\zeta = (\zeta_{\lambda\mu})$ on the fiber $\mathbb{F}_{\rho}(E_z)$, cf. with the proof of Proposition 1.2, where the indices λ and μ satisfy the condition

there exists $\ell = 1, \dots, m-1$ for which $1 \le \lambda \le r - \rho_{m-\ell} < \mu \le r$. (*)

Such coordinates parameterize flags of E_z of type

$$\{0_z\} \subset \cdots \subset V_{z,\rho_l} := \operatorname{Span}\{\epsilon_{r-\rho_l+1}(z,\zeta), \epsilon_{r-\rho_l+2}(z,\zeta), \dots, \epsilon_r(z,\zeta)\} \subset \cdots \subset E_z$$

where, for $1 \leq \alpha \leq r$,

$$\epsilon_{\alpha}(z,\zeta) = e_{\alpha}(z) + \sum \zeta_{\lambda\alpha} e_{\lambda}(z)$$

and the summation is taken over all $1 \leq \lambda < \alpha$ such that, as before, the indices λ and α satisfy Condition (*). Summing up, we have constructed in this way local holomorphic coordinates $(z, \zeta) = (z_1, \ldots, z_n, \zeta_{\lambda\mu})$ on $\mathbb{F}_{\rho}(E)$ centered at (x_0, \mathbf{f}_0) .

By construction, if 0 < l < m, the local sections

$$\epsilon_{\alpha}(z,\zeta), \quad r-\rho_l < \alpha \le r,$$
 (1.6)

form a local holomorphic frame of $U_{\rho,l}$. Moreover, a local holomorphic frame for $\pi_{\rho}^* E/U_{\rho,l}$ is given by the sections

$$\tilde{e}_{\alpha}(z,\zeta) = \text{image of } e_{\alpha}(z) \text{ in } \frac{E_z}{V_{z,\rho_l}}, \quad 1 \le \alpha \le r - \rho_l.$$
(1.7)

Similarly, if $0 < \ell < l < m$, the sections

$$\tilde{\epsilon}_{\alpha}(z,\zeta) = \text{image of } \epsilon_{\alpha}(z,\zeta) \text{ in } \frac{V_{z,\rho_l}}{V_{z,\rho_\ell}}, \quad r - \rho_l < \alpha \le r - \rho_\ell, \tag{1.8}$$

form a local holomorphic frame for $U_{\rho,l}/U_{\rho,\ell}$.

In Section 2.3.1 we use the local frames (1.6), (1.7), (1.8) to compute the Chern curvature tensors in a point of all the $\binom{m+1}{2}$ universal vector bundles of $\mathbb{F}_{\rho}(E)$.

1.2.2 Universal line bundles and their curvature

Over the flag bundle $\mathbb{F}_{\rho}(E)$ one can define canonical line bundles as follows: for $1 \leq j \leq m$ set

$$Q_{\rho,j} := \det(U_{\rho,m-j+1}/U_{\rho,m-j}).$$

For any multi-index $\mathbf{a} = (a_1, \ldots, a_r) \in \mathbb{Z}^r$ satisfying

$$a_{r-\rho_{m-j+1}+1} = a_{r-\rho_{m-j+1}+2} = \dots = a_{r-\rho_{m-j}}, \ 1 \le j \le m, \tag{1.9}$$

set, for $0 \le j \le m$, $s_j := r - \rho_{m-j}$ and define

$$Q_{
ho}^{\mathbf{a}} := Q_{
ho,1}^{\otimes a_{s_1}} \otimes \cdots \otimes Q_{
ho,m}^{\otimes a_{s_m}}$$

Example 1.8. In the notation of Example 1.3, consider the projective bundle $\mathbb{P}(E)$ on X. By taking the determinant of the short exact sequence (1.4), we have that

$$Q_{\rho,1} = \det\left(\pi_{\rho}^{*}E/\mathcal{O}_{\mathbb{P}(E)}(-1)\right) \cong \mathcal{O}_{\mathbb{P}(E)}(1) \otimes \pi_{\rho}^{*}\det E$$

and

$$Q_{\rho,2} = \mathcal{O}_{\mathbb{P}(E)}(-1).$$

Since $\rho = (0, 1, r)$ we get $s_1 = r - 1$ and $s_2 = r$. By Condition (1.9) we can choose any *r*-uple **a** satisfying $a_1 = a_2 = \cdots = a_{r-1}$. Consequently,

$$Q_{\rho}^{\mathbf{a}} = Q_{\rho,1}^{\otimes a_{r-1}} \otimes Q_{\rho,2}^{\otimes a_r} = \mathcal{O}_{\mathbb{P}(E)}(a_{r-1} - a_r) \otimes \left(\pi_{\rho}^* \det E\right)^{\otimes a_{r-1}}.$$

If E is equipped with a Hermitian metric h, then we have an induced metric on all the line bundles $Q_{\rho,j}$'s. For j < m (resp. j = m), this metric is the determinant of the quotient metric $H_{\rho,(m-j+1,m-j)}$ (resp. of the restriction metric $h_{\rho,1}$) mentioned in Remark 1.4. Consequently, we have natural metrics induced on all the line bundles $Q_{\rho}^{\mathbf{a}}$. In order to simplify the notation, it is useful to denote by h all these mentioned metrics by a slight abuse.

For a given multi-index $\mathbf{a} \in \mathbb{Z}^r$ satisfying Condition (1.9), in [Dem88a, Formula (4.9)] the Chern curvature $\Theta(Q^{\mathbf{a}}_{\rho}, h)$ of $(Q^{\mathbf{a}}_{\rho}, h)$ at the point (x_0, \mathbf{f}_0) is computed and reads

$$\Theta(Q_{\rho}^{\mathbf{a}},h)_{(x_0,\mathbf{f}_0)} = \sum_{\lambda} a_{\lambda} \Theta_{\lambda\lambda}(x_0) + \sum_{\lambda,\mu} (a_{\lambda} - a_{\mu}) d\zeta_{\lambda\mu} \wedge d\bar{\zeta}_{\lambda\mu}, \qquad (1.10)$$

where λ and μ in the second summation satisfy Condition (*), in order to have that the $\zeta_{\lambda\mu}$'s are defined. Let

$$\Xi_{\rho}^{\mathbf{a}} := c_1(Q_{\rho}^{\mathbf{a}}, h) = \frac{i}{2\pi} \Theta(Q_{\rho}^{\mathbf{a}}, h) \in \mathcal{A}^{1,1}(\mathbb{F}_{\rho}(E))$$

be the first Chern form of $(Q_{\rho}^{\mathbf{a}}, h)$, which represents the first Chern class $c_1(Q_{\rho}^{\mathbf{a}})$. Remark 1.9. In the particular case where the multi-index is non increasing, which is indeed the case actually considered in [Dem88a, Formula (4.9)], and which shall be of special interest later, Condition (*) on λ and μ in the second summation above is equivalent to require $a_{\lambda} > a_{\mu}$, so that the curvature formula becomes

$$\Theta(Q_{\rho}^{\mathbf{a}},h)_{(x_0,\mathbf{f}_0)} = \sum_{\lambda} a_{\lambda} \Theta_{\lambda\lambda}(x_0) + \sum_{a_{\lambda} > a_{\mu}} (a_{\lambda} - a_{\mu}) d\zeta_{\lambda\mu} \wedge d\bar{\zeta}_{\lambda\mu}.$$
 (1.11)

Observe that $\sum_{a_{\lambda}>a_{\mu}}(a_{\lambda}-a_{\mu})d\zeta_{\lambda\mu}\wedge d\bar{\zeta}_{\lambda\mu}$ gives a positive definite block for the curvature of $(Q_{\rho}^{\mathbf{a}},h)$ if and only if the non increasing multi-index **a** is strictly decreasing at each place where it is allowed by Condition (1.9), *i.e.*

$$a_{s_1} > a_{s_2} > \dots > a_{s_m}.$$
 (1.12)

For $1 \leq j \leq m$, denote by $\mathbf{1}_j$ the element of \mathbb{Z}^r with 1's in the places $s_{j-1} + 1, s_{j-1} + 2, \ldots, s_j$, and 0's elsewhere. By definition, we have that $Q_{\rho}^{\mathbf{1}_j} = Q_{\rho,j}$, and thanks to Formula (1.10) we can recover the curvature of the line bundle $Q_{\rho,j}$ at the point $(x_0, \mathbf{f}_0) \in \mathbb{F}_{\rho}(E)$:

$$\Theta(Q_{\rho,j},h)_{(x_0,\mathbf{f}_0)} = \sum_{\substack{\lambda = s_{j-1}+1 \\ \mu = s_{j-1}+1,\dots,s_j}}^{s_j} \Theta_{\lambda\lambda}(x_0) - \sum_{\substack{\lambda = 1,\dots,s_{j-1} \\ \mu = s_{j-1}+1,\dots,s_j}} d\zeta_{\lambda\mu} \wedge d\bar{\zeta}_{\lambda\mu} + \sum_{\substack{\lambda = s_{j-1}+1,\dots,s_j \\ \mu = s_j+1,\dots,r}} d\zeta_{\lambda\mu} \wedge d\bar{\zeta}_{\lambda\mu}.$$
(1.13)

Finally, let

$$\Xi_{\rho,j} := c_1(Q_{\rho,j},h) = \frac{i}{2\pi} \Theta(Q_{\rho,j},h) \in \mathcal{A}^{1,1}(\mathbb{F}_{\rho}(E))$$

be the first Chern form of $(Q_{\rho,j}, h)$, which represents the first Chern class $c_1(Q_{\rho,j})$.

As before, we shall drop the subscript ρ and write Q_j , Ξ_j and $Q^{\mathbf{a}}$, $\Xi^{\mathbf{a}}$ in case of complete flag bundles.

1.2.3 Splitting of the tangent bundle

Consider the short exact sequence

$$0 \to \ker(d\pi_{\rho}) \hookrightarrow T_{\mathbb{F}_{\rho}(E)} \xrightarrow{d\pi_{\rho}} \pi_{\rho}^* T_X \to 0$$

induced by the differential of $\pi_{\rho} \colon \mathbb{F}_{\rho}(E) \to X$, where $T_{\mathbb{F}_{\rho}(E)}$ and T_X are the tangent bundles of $\mathbb{F}_{\rho}(E)$ and of X respectively. Recall that $\ker(d\pi_{\rho})$ is the *relative tangent* bundle, usually denoted by $T_{\mathbb{F}_{\rho}(E)/X}$.

We now define a natural orthogonal splitting of $T_{\mathbb{F}_{\rho}(E)}$ into a vertical and horizontal part. In order to do this, observe that for any given weight **a** as described at the end of Remark 1.9, by Formula (1.11) the line bundle $Q_{\rho}^{\mathbf{a}}$ is relatively positive. Hence, $\Xi_{\rho}^{\mathbf{a}}$ gives a positive definite Hermitian form whenever restricted to the relative tangent bundle $T_{\mathbb{F}_{\rho}(E)/X}$. Therefore, for any such **a**, we get a corresponding orthogonal decomposition (in the smooth category)

$$T_{\mathbb{F}_{\rho}(E)} = T_{\mathbb{F}_{\rho}(E)/X} \oplus T_{\mathbb{F}_{\rho}(E)/X}^{\perp_{\Xi_{\rho}^{\mathbf{a}}}}.$$

Finally, from the explicit expression of Formula (1.11), we see that such a decomposition is independent of the particular choice of the weight **a**, so that it depends only on h and we can drop any reference to the weight and write

$$T_{\mathbb{F}_{\rho}(E)} = T_{\mathbb{F}_{\rho}(E)/X} \oplus T_{\mathbb{F}_{\rho}(E)/X}^{\perp_{h}}.$$
(1.14)

We denote by

$$p_1: T_{\mathbb{F}_{\rho}(E)} \to T_{\mathbb{F}_{\rho}(E)/X}$$
 and $p_2: T_{\mathbb{F}_{\rho}(E)} \to T_{\mathbb{F}_{\rho}(E)/X}^{\perp_h}$

the natural projections relative to the splitting (1.14).

Remark 1.10. Observe that the above splitting is compatible with $d\pi_{\rho}$ in the following sense. The restriction $d\pi_{\rho}|_{T^{\perp h}_{\mathbb{F}_{\rho}(E)/X}} : T^{\perp h}_{\mathbb{F}_{\rho}(E)/X} \to \pi^{*}_{\rho}T_{X}$ is a smooth isomorphism of complex vector bundles, and moreover, by a direct pointwise computation in the local holomorphic coordinates chosen above, we have $d\pi_{\rho} = d\pi_{\rho} \circ p_{2}$ and $d\pi_{\rho} \circ p_{1} = 0$.

In particular, observe that, for any given point $(x_0, \mathbf{f}_0) \in \mathbb{F}_{\rho}(E)$, given the holomorphic coordinates (z, ζ) centered at (x_0, \mathbf{f}_0) as above, we explicitly have that

$$T_{\mathbb{F}_{\rho}(E)/X,(x_{0},\mathbf{f}_{0})} = \operatorname{Span}\left\{\frac{\partial}{\partial\zeta_{\lambda\mu}}\Big|_{(z,\zeta)=(0,0)}\right\},\$$
$$T_{\mathbb{F}_{\rho}(E)/X,(x_{0},\mathbf{f}_{0})}^{\perp h} = \operatorname{Span}\left\{\frac{\partial}{\partial z_{k}}\Big|_{(z,\zeta)=(0,0)}\right\}.$$

1.3 Schur polynomials

We recall here some notation about Schur polynomials, essentially taken from the exposition in [FL83] (see also [DPS94, § 2] and [Ful97]).

Denote by $\Lambda(k,r)$ the set of all the partitions $\sigma = (\sigma_1, \ldots, \sigma_k)$ in \mathbb{N}^k such that

$$r \ge \sigma_1 \ge \cdots \ge \sigma_k \ge 0, \quad |\sigma| = \sum_{j=1}^k \sigma_j = k.$$

For every $\sigma \in \Lambda(k, r)$ we can define a *Schur polynomial* $S_{\sigma} \in \mathbb{Z}[c_1, \ldots, c_r]$ of weighted degree 2k (we regard c_j as having degree 2j) as

$$S_{\sigma}(c_{1},...,c_{r}) := \det(c_{\sigma_{i}+j-i})_{1 \le i,j \le k}$$

=
$$\det\begin{pmatrix} c_{\sigma_{1}} & c_{\sigma_{1}+1} & \cdots & c_{\sigma_{1}+k-1} \\ c_{\sigma_{2}-1} & c_{\sigma_{2}} & \cdots & c_{\sigma_{2}+k-2} \\ \vdots & \vdots & \ddots & \vdots \\ c_{\sigma_{k}-k+1} & c_{\sigma_{k}-k+2} & \cdots & c_{\sigma_{k}} \end{pmatrix}$$

where, by convention, $c_0 = 1$ and $c_{\ell} = 0$ if $\ell \notin [0, r]$.

The Schur polynomials, as $\sigma \in \Lambda(k, r)$ varies, form a basis for the Q-vector space of degree 2k weighted homogeneous polynomials in r variables. Thus, given such a polynomial P we can write

$$P = \sum_{\sigma \in \Lambda(k,r)} b_{\sigma}(P) S_{\sigma}.$$

The set of all P such that $b_{\sigma}(P) \geq 0$ for every $\sigma \in \Lambda(k, r)$, which is called the set of *positive polynomials*, is of course a positive convex cone, which we call $\Pi(r)$ following [Gri69] (remark that this is not exactly the positive cone considered by Griffiths, but they coincides a *posteriori* thanks to the work of [FL83]). It is well known that any product of Schur polynomials can be written as a linear combination of Schur polynomials with non-negative integral coefficients; the values of these coefficients is given combinatorially by the Littlewood–Richardson rule. Thus, these positive cones are stable under product (cf. with the analogous property for wedge product of strongly positive forms, as observed in Remark 3.7).

Now, if E is a rank r holomorphic vector bundle over a complex manifold X and if σ is a partition in $\Lambda(k, r)$, the *Schur class* of E associated to σ is the cohomology class

$$S_{\sigma}(E) := S_{\sigma}(c_1(E), \dots, c_r(E)) \in H^{2k}(X)$$

$$(1.15)$$

formally obtained by computing S_{σ} on the Chern classes of E.

In the same way, if h is a Hermitian metric on E we can define the *Schur form* of (E, h) associated to σ , formally obtained by computing S_{σ} on the Chern forms of (E, h), and we denote it by $S_{\sigma}(E, h)$. Clearly, the closed differential 2*k*-form $S_{\sigma}(E, h)$ is a special representative for the class $S_{\sigma}(E)$.

Example 1.11. Denote by $s_k(E, h)$ the k-th Segre form of (E, h), which represents the Segre class $s_k(E)$. Given a partition $\sigma \in \Lambda(k, r)$, interesting examples of Schur forms are:

$$S_{\sigma}(E,h) = \begin{cases} c_k(E,h) & \text{if } \sigma = (k,0,\dots,0), \\ c_1(E,h) \wedge c_{k-1}(E,h) - c_k(E,h) & \text{if } \sigma = (k-1,1,0,\dots,0), \\ (-1)^k s_k(E,h) & \text{if } \sigma = (1,\dots,1), \end{cases}$$

and at the level of cohomology classes we have similar equalities.

1.3.1 Generalized Schur classes and forms

Here we want to introduce a family of cohomology classes (resp. differential forms) which are given by dropping the assumption that σ is a partition in $\Lambda(k, r)$. The following notation is taken from [DP17, §4], and will be essential in order to state Formula (2.19).

Definition 1.12. Let $\sigma = (\sigma_1, \ldots, \sigma_k) \in \mathbb{Z}^k$ be a sequence of integers. The generalized Schur class associated to σ is the cohomology class

$$s_{\sigma}(E) := \det(s_{\sigma_i+j-i}(E))_{1 \le i, j \le k}$$

in $H^{2|\sigma|}(X)$, where, as usual, $s_0(E) = 1$ and $s_\ell(E) = 0$ if $\ell \notin [0, n]$.

The relationship between Schur classes and Definition 1.12 is given in the following.

Example 1.13. Let σ be a partition in $\Lambda(k, r)$. It follows from the Jacobi–Trudi identities ([Ful97]) that

$$s_{\sigma}(E) = (-1)^{|\sigma|} S_{\sigma'}(E),$$
 (1.16)

where σ' is the conjugate partition of σ , obtained through the transposition of the Young diagram of σ . For instance, by Formula (1.16) the partition $(1, \ldots, 1)$ gives $(-1)^k c_k(E)$, which is the k-th signed Chern class, while the Segre class $s_k(E)$ is associated to the partition $(k, 0, \ldots, 0)$.

Note that the sign $(-1)^{|\sigma|}$ in Formula (1.16) is due to the fact that the definition of Schur class given in Formula (1.15) involves Chern classes, while Definition 1.12 is given in terms of Segre classes. The reason why we choose the Segre classes' approach in Definition 1.12 is to make our notation uniform with that of [DP17]. This choice is very useful also in Section 3.2.1.

Similarly to Definition 1.12, given $\sigma \in \mathbb{Z}^k$ we call the differential form

$$s_{\sigma}(E,h) := \det(s_{\sigma_i+j-i}(E,h))_{1 \le i,j \le k}$$

the generalized Schur form associated to σ . This terminology is inspired by Formula (1.16) above.

1.4 Positivity notions

In this section we give some notions of positivity for differential forms and for vector bundles, needed to state the results in Chapter 3.

1.4.1 Positivity notions for differential forms

We recall here some basic notions about the positivity of differential forms. The following exposition is taken, mainly, from [HK74] and [Dem12, § III].

Let V be a complex vector space of dimension n and let (e_1, \ldots, e_n) be a basis of V. The notation $(e_1^{\vee}, \ldots, e_n^{\vee})$ stands for the corresponding dual basis of V^{\vee} . For $0 \leq p, q \leq n$, denote by $\Lambda^{p,q}V^{\vee}$ the space of exterior forms of bi-degree (p,q) on V. A form $u \in \Lambda^{p,q}V^{\vee}$ is real if $\bar{u} = u$; since $\bar{u} \in \Lambda^{q,p}V^{\vee}$, it follows that p = q in this case. Let $\Lambda_{\mathbb{R}}^{p,p}V^{\vee}$ denote the space of real (p, p)-forms. **Definition 1.14.** A form $\nu \in \Lambda^{n,n} V^{\vee}$ is called a *positive* volume form if

$$\nu = \tau \ ie_1^{\vee} \wedge \overline{e}_1^{\vee} \wedge \dots \wedge ie_n^{\vee} \wedge \overline{e}_n^{\vee}$$

for some $\tau \in \mathbb{R}, \tau \geq 0$.

Of course, this notion is independent on the choice of the basis (e_1, \ldots, e_n) .

From now on, for $0 \le p \le n$ set q = n - p. Recall that a (q, 0)-form which can be expressed as $\beta_1 \land \cdots \land \beta_q$ for $\beta_1, \ldots, \beta_q \in V^{\lor}$ is called *decomposable*.

Definition 1.15. An exterior form $u \in \Lambda^{p,p}_{\mathbb{R}} V^{\vee}$ is called

- *positive*, if for every $\beta \in \Lambda^{q,0} V^{\vee}$ decomposable, $u \wedge i^{q^2} \beta \wedge \overline{\beta}$ is a positive volume form;
- Hermitian positive, if for every $\beta \in \Lambda^{q,0} V^{\vee}$, $u \wedge i^{q^2} \beta \wedge \overline{\beta}$ is a positive volume form;
- strongly positive, if there are decomposable forms $\alpha_1, \ldots, \alpha_N \in \Lambda^{p,0} V^{\vee}$ such that u can be expressed as $\sum_{s=1}^{N} i^{p^2} \alpha_s \wedge \bar{\alpha}_s$.

Remark 1.16. The terminology used for the positivity of forms is, currently, not always standard in the literature. We have chosen here to follow the terminology of [Dem12, §III], which is a standard reference for positivity of forms. Note however that [Dem12, §III] does not deal with the intermediate notion of positivity (which is called simply *positivity* in [HK74]). The reason why we call it *Hermitian* is due to the fact that $u \in \Lambda_{\mathbb{R}}^{p,p} V^{\vee}$ is Hermitian positive if and only if for each $\beta, \eta \in \Lambda^{q,0} V^{\vee}$, $(\beta, \eta) \mapsto u \wedge i^{q^2} \beta \wedge \bar{\eta}$ gives a positive semidefinite Hermitian form on $\Lambda^{q,0} V^{\vee}$.

Remark 1.17. Of course, with an appropriate modification of Definition 1.15, one can define the *strict* notions of positivity for forms. Mainly, we are dealing here with Griffiths semipositive vector bundles, so it is more natural to consider the three notions given in Definition 1.15, rather than the strict ones. Moreover, note that some authors (see, for instance, [Zhe00; Gul12; Li21]) use the term *non-negative*, or *semi-positive*, (resp. *positive*) instead of *positive* (resp. *strictly positive*).

Example 1.18. Let $\xi \in \Lambda^{p,0}V^{\vee}$, then $i^{p^2}\xi \wedge \overline{\xi}$ is a Hermitian positive form. Indeed, for every $\beta \in \Lambda^{q,0}V^{\vee}$, the wedge product $\xi \wedge \beta$ equals $\lambda e_1^{\vee} \wedge \cdots \wedge e_n^{\vee}$ for some $\lambda \in \mathbb{C}$, hence $i^{p^2}\xi \wedge \overline{\xi} \wedge i^{q^2}\beta \wedge \overline{\beta} = i^{n^2}\xi \wedge \beta \wedge \overline{\xi \wedge \beta}$ is a positive volume form.

Let $\operatorname{WP}^p V^{\vee}$, $\operatorname{HP}^p V^{\vee}$ and $\operatorname{SP}^p V^{\vee}$ denote respectively the closed positive convex cones contained in $\Lambda^{p,p}_{\mathbb{R}}V^{\vee}$ spanned by positive, Hermitian positive and strongly positive forms. The notations WP and SP are taken from [HK74]. It is straightforward to see that, in general,

$$SP^{p}V^{\vee} \subseteq HP^{p}V^{\vee} \subseteq WP^{p}V^{\vee}.$$
(1.17)

Remark 1.19. The two inclusions in (1.17) become equalities when p = 0, 1, n - 1, n. Indeed, if p = 0, n all the positivity notions in Definition 1.15 do coincide. If p = 1, the one-to-one correspondence between Hermitian forms and real (1, 1)-forms shows by a diagonalization argument that every positive (1, 1)-form is strongly positive. By duality, it is also true if p = n - 1. If K is a convex cone in $\Lambda^{q,q}_{\mathbb{R}}V^{\vee}$, then its *dual cone* is

 $\mathbf{K}^* := \{ u \in \Lambda_{\mathbb{R}}^{p,p} V^{\vee} \mid u \wedge v \text{ is a positive volume form } \forall v \in \mathbf{K} \}.$

By Definition 1.15, $WP^p V^{\vee} = (SP^q V^{\vee})^*$ and given that the bidual of a convex cone is equal to its closure, we have that $SP^p V^{\vee} = (WP^q V^{\vee})^*$. This leads to the following characterization of strong positivity.

Proposition 1.20. A form $u \in \Lambda_{\mathbb{R}}^{p,p}V^{\vee}$ is strongly positive if and only if for every positive (q,q)-form $v, u \wedge v$ is a positive volume form.

We also need the following characterization of Hermitian positivity, which follows from the more general [HK74, Theorem 1.2].

Proposition 1.21. A form $u \in \Lambda_{\mathbb{R}}^{p,p}V^{\vee}$ is Hermitian positive if and only if there are $\xi_1, \ldots, \xi_N \in \Lambda^{p,0}V^{\vee}$ such that $u = \sum_{s=1}^N i^{p^2}\xi_s \wedge \bar{\xi}_s$.

From Proposition 1.21, we deduce that the wedge product of two Hermitian positive forms is again Hermitian positive (of course, an analogous property holds for strongly positive forms by definition). Moreover, it is now clear that the dual cone $(\operatorname{HP}^{q} V^{\vee})^{*}$ equals $\operatorname{HP}^{p} V^{\vee}$.

We also want to recall that, sometimes, Proposition 1.21 is given in the literature as a definition; see for instance [Gri69, p. 240] and [Li21, §2].

Remark 1.22. If $2 \leq p \leq n-2$, the two inclusions in (1.17) are strict. To see this, it is sufficient to observe that the Hermitian positive form $i^{p^2}\xi \wedge \bar{\xi}$ is strongly positive if and only if ξ is decomposable (see [Dem12, §III, (1.10) Remark] and [HK74, Proposition 1.5]). Given that, for instance, the (p, 0)-form $(e_1^{\vee} \wedge e_2^{\vee} + e_3^{\vee} \wedge e_4^{\vee}) \wedge e_5^{\vee} \wedge \cdots \wedge e_{p+2}^{\vee}$ is not decomposable, we deduce that $\mathrm{SP}^p V^{\vee} \subsetneq \mathrm{HP}^p V^{\vee}$. Since the duality of cones reverses the inclusions, we get also that $\mathrm{HP}^p V^{\vee} \subsetneq \mathrm{WP}^p V^{\vee}$.

An explicit example of a positive form which is not Hermitian positive can be found in [HK74, p. 50]. As a byproduct, [HK74] constructs a positive form and a Hermitian positive form, for which their wedge product is a negative volume form. Therefore, unlike $SP^p V^{\vee}$ and $HP^p V^{\vee}$, the cone of positive forms is not stable under wedge product (compare this with [BP13]).

It is useful to recall the following characterization of positivity.

Proposition 1.23. A form $u \in \Lambda^{p,p}_{\mathbb{R}} V^{\vee}$ is positive if and only if, equivalently,

- 1. for every vector subspace $W \subseteq V$ with $\dim_{\mathbb{C}} W = p$, the restriction $u|_W$ is a positive volume form on W;
- 2. for every $w_1, \ldots, w_p \in V$, $(-i)^{p^2} u(w_1, \ldots, w_p, \overline{w}_1, \ldots, \overline{w}_p) \ge 0$.

We also recall the following fact (see [Dem12, § III, (1.12) Proposition]).

Proposition 1.24. Let $A: \tilde{V} \to V$ be a complex linear map of complex vector spaces. If $u \in \Lambda^{p,p}_{\mathbb{R}} V^{\vee}$ is (Hermitian, resp. strongly) positive, then the pull-back $A^* u \in \Lambda^{p,p}_{\mathbb{R}} \tilde{V}^{\vee}$ is (Hermitian, resp. strongly) positive.

Of course, all of the definitions and results given in this section can be extended to a complex manifold X. It is sufficient to take $V = T_{X,x}$ for every $x \in X$ and to check that all the concepts expressed here are independent by change of holomorphic coordinates. This follows from Definition 1.14.

Lastly, we recall a well-known property of positive forms on complex manifolds, for which we give the proof.

Proposition 1.25. Let $\pi: S \to X$ be a proper holomorphic submersion of complex manifolds. If η is a (Hermitian, resp. strongly) positive form on S, then the push-forward $\pi_*\eta$ is a (Hermitian, resp. strongly) positive form on X.

Proof. First of all, suppose that η is a positive top degree form on S, and denote by s the relative dimension of π . Let dV_X be a volume form on X and Ω be a positive (1, 1)-form on S. By Remark 1.19 and Proposition 1.24, $\Omega^{\wedge s} \wedge \pi^* dV_X$ is a positive top degree form on S. Hence, there is a non-negative function $f \in C^{\infty}(S)$ such that

$$\eta = f \ \Omega^{\wedge s} \wedge \pi^* dV_X.$$

For every $x \in X$, it follows from the projection formula that

$$\pi_*\eta_x = \left(\pi_*\left(f \ \Omega^{\wedge s}\right) \wedge dV_X\right)_x = \left(\int_{y \in \pi^{-1}(x)} f(x,y) \ \Omega^{\wedge s}_{(x,y)}\right) dV_x.$$

The integral on the right hand side is non-negative since the function f is. Hence $\pi_*\eta$ is a positive top degree form on X.

Set $n = \dim X$. For $0 \le k < n$, suppose now that η is a (Hermitian, resp. strongly) positive (k + s, k + s)-form on S, and take any (Hermitian, resp. positive) strongly positive (n - k, n - k)-form α on X. By the previous point, Definition 1.15, Propositions 1.20 and 1.24, and the projection formula, we have that the top degree form

$$\pi_*\eta \wedge \alpha = \pi_*(\eta \wedge \pi^*\alpha)$$

is positive on X. Therefore $\pi_*\eta$ is a (Hermitian, resp. strongly) positive (k, k)-form on X.

1.4.2 Positivity notions for vector bundles

A Hermitian holomorphic line bundle (L, h) over a complex manifold X is *positive* if the Chern curvature $i\Theta(L, h)$ is a positive (1, 1)-form.

If X is compact, then a line bundle \mathcal{L} over X is *ample* if there is a holomorphic embedding $\iota: X \hookrightarrow \mathbb{P}^N$ for which $\mathcal{L}^{\otimes k} \cong \iota^* \mathcal{O}_{\mathbb{P}^N}(1)$ for some k > 0, hence X is projective.

Since the hyperplane bundle $\mathcal{O}_{\mathbb{P}^N}(1)$ equipped with the standard Fubini–Study metric is positive (its curvature is indeed the Fubini–Study Kähler form on \mathbb{P}^N), then \mathcal{L} endowed with the k-th root of the pull-back metric is a positive line bundle. Conversely, any positive line bundle over a compact complex manifold is ample by the Kodaira embedding theorem [Kod54].

Vector bundles of higher rank have several notions of positivity. We refer, for instance, to [Nak55; BC65; Gri69; Laz04; Dem12; LSY13; Fin21] and the references therein for more details on the theory of positive vector bundles.

In particular, there are different notions of *positive* Hermitian metric which are in general not equivalent. In this thesis, we consider essentially one of these notions which we describe below.

Let $(E, h) \to X$ be a Hermitian holomorphic vector bundle of rank r over a complex manifold of dimension n. At any point $x \in X$ and with respect to a unitary frame (e_1, \ldots, e_r) for E_x , write the Chern curvature as

$$\Theta(E,h)_x = \sum_{1 \le \alpha, \beta \le r} \sum_{1 \le j,k \le n} c_{jk\alpha\beta} \, dz_j \wedge d\bar{z}_k \otimes e_\alpha^{\vee} \otimes e_\beta.$$

To $i\Theta(E,h) \in \mathcal{A}^{1,1}(X, \operatorname{Herm}(E,h))$ we can associate a natural Hermitian form $\theta(E,h)$ on $T_X \otimes E$ which in $x \in X$ is defined as

$$\theta(E,h)_x = \sum_{1 \le \alpha, \beta \le r} \sum_{1 \le j,k \le n} c_{jk\alpha\beta} \left(dz_j \otimes e_\alpha^\vee \right) \otimes \left(\overline{dz_k \otimes e_\beta^\vee} \right).$$

Definition 1.26 ([Gri69]). The vector bundle (E, h) is Griffiths semipositive (resp. Griffiths positive) if for every $x \in X$, $v = \sum v_{\alpha} e_{\alpha} \in E_x$, $\tau = \sum \tau_j \frac{\partial}{\partial z_j} \in T_{X,x}$ we have

$$\langle \Theta(E,h)_x \cdot v, v \rangle_h(\tau, \bar{\tau}) = \sum_{1 \le \alpha, \beta \le r} \sum_{1 \le j, k \le n} c_{jk\alpha\beta} \tau_j \bar{\tau}_k v_\alpha \bar{v}_\beta \ge 0$$

(resp. > 0 and = 0 if and only if v or τ is the zero vector).

In other words, the quadratic form associated to $\theta(E, h)_x$ takes nonnegative (resp. positive) values on non zero tensors $\tau \otimes v \in T_{X,x} \otimes E_x$ for any x.

By requiring that $\theta(E, h)$ is positive (semi)definite as a Hermitian form on $T_X \otimes E$ we get the stronger notion of *Nakano (semi)positivity* for (E, h) (see [Nak55]). Of course, there are the corresponding notions of Griffiths/Nakano (semi)negativity.

Since the Chern curvature tensor of the dual bundle E^{\vee} equipped with the dual metric h^{-1} is given by the opposite of the transpose of $\Theta(E, h)$, we also have the notion of *dual Nakano (semi)positivity* for (E, h), which asks that (E^{\vee}, h^{-1}) is Nakano (semi)negative. See, for instance, [Dem12; Fin21] for more details on (dual) Nakano positivity and for additional concepts of positivity of h, which interpolate between Griffiths (semi)positivity and (dual) Nakano (semi)positivity.

By definition, Nakano (semi)positivity implies Griffiths (semi)positivity. Moreover, by recalling that (E, h) is Griffiths (semi)positive if and only if (E^{\vee}, h^{-1}) is Griffiths (semi)negative, we easily obtain that dual Nakano (semi)positivity implies Griffiths (semi)positivity. However, it is known that the holomorphic tangent bundle $T_{\mathbb{P}^N}$ with the Fubini–Study Kähler form is dual Nakano positive and Nakano semipositive, but $T_{\mathbb{P}^N}$ does not admit a Nakano positive Hermitian metric for any N > 1. See [LSY13] for an example of a Nakano positive and dual Nakano semipositive metric on a vector bundle which does not admit any dual Nakano positive metric. Observe that when dim X = 1 or rk E = 1 Nakano and dual Nakano (semi)positivity coincide with Griffiths (semi)positivity.

Now we recall the corresponding notion of *ampleness* for vector bundles for which we refer to [Laz04].

Definition 1.27 ([Har66]). A holomorphic vector bundle \mathcal{V} over X is called *ample* if the line bundle $\mathcal{O}_{\mathbb{P}(\mathcal{V}^{\vee})}(1)$ is ample over $\mathbb{P}(\mathcal{V}^{\vee})$.

Griffiths positivity is considered as a differential geometric counterpart of ampleness. This because both notions share similar properties. For instance, it is well-known that any quotient of a Griffiths (semi)positive (resp. ample) vector bundle is Griffiths (semi)positive (resp. ample). Also, the direct sum or tensor power of two Griffiths (semi)positive (resp. ample) vector bundles is Griffiths (semi)positive (resp. ample) vector bundles is Griffiths (semi)positive (resp. ample).

In particular, it is well-known ([Gri69; Dem12]) that a Griffiths positive vector bundle over a compact complex manifold is ample. In the notation of Section 1.2, we can see that this also follows from Formula 1.13 with $\rho = (0, r - 1, r)$ and j = 1. Indeed, in this case we have $\mathbb{F}_{\rho}(E) = \mathbb{G}_{r-1}(E) \cong \mathbb{P}(E^{\vee})$ and $U_{\rho,1} = \gamma_{r-1}$. Thus,

$$Q_{\rho,j} = \pi_{\rho}^* E / \gamma_{r-1} \cong \mathcal{O}_{\mathbb{P}(E^{\vee})}(1)$$

and Formula 1.13 proves that $\mathcal{O}_{\mathbb{P}(E^{\vee})}(1)$ is a positive line bundle.

It is a conjecture due to Griffiths [Gri69] that ampleness and Griffiths positivity are indeed equivalent. However, this is still open in general beside the case when X is a curve (see [Ume73; CF90]).

It is natural to ask if certain conditions of positivity on the vector bundle induce, in turn, the positivity of objects that can be defined from it. This kind of positivity issues can be placed at both algebraic (e.g., ampleness) and differential geometric level (e.g., Griffiths positivity). Here we are interested in the latter, and, for instance (see the introduction and Chapter 3), one may ask if the Chern forms coming from a Hermitian vector bundle (E, h) with Griffiths (semi)positive curvature are positive. The answer is trivially affirmative in the case of the first Chern form. Indeed, by definition

$$c_1(E,h) = \frac{i}{2\pi} \operatorname{tr}_{\operatorname{End}(E)} \Theta(E,h)$$

and the diagonal entries of the matrix $i\Theta(E, h)$ are positive (1, 1)-forms, since h is a Griffiths (semi)positive metric.

However, the answer to this question for other (special combinations of) Chern forms is absolutely non trivial, and Chapter 3 is dedicated to the study of this positivity issue.

Chapter 2

Curvature of universal bundles and push-forward formulæ

The aim of this chapter is to give explicit formulæ to compute the push-forward of a polynomial in the Chern forms of universal vector bundles. In this way, we are able to provide a differential geometric version of the Gysin formulæ for flag bundles (see, for instance, [Dam73; Ilo78; KT15; DP17]), which compute the push-forwards at the level of cohomology.

In Section 2.1 we describe the explicit expressions of the Gysin formulæ for flag bundles which we use throughout this thesis. These are the Darondeau–Pragacz formulæ ([DP17]). In Section 2.2 we prove Theorem B, which is crucial in Chapter 3 due to its applications in positivity. In Section 2.3 we compute the curvature of the universal vector bundles over flag bundles. This is the main ingredient for the proof of Theorem A, which is the metric counterpart of the universal Gysin formulæ for flag bundles in cohomology. Finally, in Section 2.4 we give an alternative version of our push-forward formulæ by using monomials in the virtual Chern roots.

From now on, suppose that E is a rank r holomorphic vector bundle over a complex *n*-dimensional manifold X. Let m > 1 be a natural number, and fix a dimension sequence $\rho = (\rho_0, \ldots, \rho_m)$. Let

$$\pi_{\rho} \colon \mathbb{F}_{\rho}(E) \to X$$

be the associated flag bundle, and denote by d_{ρ} its relative dimension.

2.1 Gysin formulæ à la Darondeau–Pragacz

We recall here a formula by Darondeau and Pragacz given in [DP17, Proposition 1.2], which allows to compute the push-forward of a polynomial in the Chern roots of the vector bundle E at the level of cohomology (in fact at the level of Chow rings, but cohomology suffices for our purposes).

First of all, observe that, even if [DP17, Proposition 1.2] is stated for algebraic manifolds over an algebraically closed field \mathbb{K} , when $\mathbb{K} = \mathbb{C}$ it also holds for non necessary algebraic complex manifolds. Indeed, ultimately, their proof only rely upon the push-forward formula for the tautological class on the projectivized bundle in terms of Segre classes, which is definitely valid also for general complex manifolds.

In what follows, for $k \in \mathbb{N}$, we use the same symbol for the push-forward

$$(\pi_{\rho})_* \colon \mathcal{A}^{2(d_{\rho}+k)}(\mathbb{F}_{\rho}(E)) \to \mathcal{A}^{2k}(X)$$

of differential forms and the one induced in cohomology, namely

$$(\pi_{\rho})_* \colon H^{2(d_{\rho}+k)}(\mathbb{F}_{\rho}(E)) \to H^{2k}(X)$$

With the same notation of [DP17], we denote by ξ_1, \ldots, ξ_r the (virtual) Chern roots of $\pi_{\rho}^* E^{\vee}$. The Darondeau–Pragacz formula allows us to compute the push-forward of any cohomology class of the form

$$\tilde{F}(\xi_1,\ldots,\xi_r) \in H^{2(d_{\rho}+k)}(\mathbb{F}_{\rho}(E))$$

in terms of the Segre classes $s_j := s_j(E), 1 \le j \le n$, of the vector bundle E (we are thus implicitly asking here that the polynomial \tilde{F} has the appropriate symmetries, which ensure that $\tilde{F}(\xi_{\bullet})$ is a cohomology class).

More precisely, let t_1, \ldots, t_r be a set of formal variables, then

$$(\pi_{\rho})_* F(\xi_1, \dots, \xi_r) = [t_1^{b_1} \dots t_r^{b_r}] \left(\tilde{F}(t_1, \dots, t_r) \prod_{1 \le i \le r} \left(1 + \sum_{j=1}^n \frac{s_j}{t_i^j} \right) \prod_{1 \le i < j \le r} (t_i - t_j) \right)$$
(2.1)

where, for a monomial \mathfrak{m} and a (Laurent) polynomial P, the notation $[\mathfrak{m}](P)$ stands for the coefficient of \mathfrak{m} in P, and the rule to determine the b_j 's is as follows: for $r - \rho_{\ell} < j \leq r - \rho_{\ell-1}$, say $j = r - \rho_{\ell} + i$ for some $i = 1, \ldots, \rho_{\ell} - \rho_{\ell-1}$, we set $b_j = r - i$.

Formula (2.1) is (one possible instance of) Darondeau–Pragacz formula.

Remark 2.1. Let us call $\Psi(s_1, \ldots, s_n)$ the right-hand side of Formula (2.1). If we consider the Segre classes s_1, \ldots, s_n as formal variables, we can affirm that the polynomial $\Psi(s_1, \ldots, s_n)$ is universal, in the sense that its coefficients depends only upon \tilde{F} and the rank r. Moreover, Ψ is, by construction, weighted homogeneous of degree 2k, since deg $s_j = 2j$.

In the notation of Section 1.3.1, we now recall the determinantal version of Darondeau–Pragacz formula stated in [DP17, Proposition 4.2]. We give it with a modification as follows.

Proposition 2.2. Take the polynomial \tilde{F} as before, and write

$$\tilde{F}(\xi_1,\ldots,\xi_r) = \sum_{|\lambda|=d_\rho+k} a_\lambda \xi_1^{\lambda_1} \ldots \xi_r^{\lambda_r}.$$

If ν is the non-decreasing sequence of integers determined by ρ as:

$$\nu_i = r - \rho_\ell \quad \text{for } r - \rho_\ell < i \le r - \rho_{\ell-1}$$
 (2.2)

(in particular, if m = r the sequence ν is given by $\nu_i = i - 1$), then, in terms of generalized Schur classes, we have

$$(\pi_{\rho})_* \tilde{F}(\xi_1, \dots, \xi_r) = \sum_{|\lambda| = d_{\rho} + k} a_{\lambda} s_{(\lambda - \nu)}(E)$$
(2.3)

where the notation $(\sigma_1, \ldots, \sigma_r) \leftarrow$ stands for $(\sigma_r, \ldots, \sigma_1)$ and the difference of λ and ν is defined componentwise.

Remark 2.3. Proposition 2.2 is stated differently from [DP17, Proposition 4.2]. Indeed, the original statement of Darondeau and Pragacz is given in terms of the Chern roots of $U_{\rho,m-1}^{\vee}$ (which is the greatest proper tautological sub-bundle of $\pi_{\rho}^* E^{\vee}$), while we need a push-forward formula involving all the Chern roots of $\pi_{\rho}^* E^{\vee}$.

What follows is the proof of Proposition 2.2, which is based on [DP17, Proposition 4.2] and [DP17, Proposition 1.2].

Proof of Proposition 2.2. Denote by $s_{1/t}$ the Laurent polynomial $1 + \sum_{j=1}^{n} s_j/t^j$. From Formula (2.1) we have

$$(\pi_{\rho})_* \tilde{F}(\xi_1, \dots, \xi_r) = [t_1^{b_1} \cdots t_r^{b_r}] \left(\sum a_\lambda t_1^{\lambda_1} \cdots t_r^{\lambda_r} \prod_{1 \le i \le r} s_{1/t_i} \prod_{1 \le i < j \le r} (t_i - t_j) \right),$$

where $b_j = 2r - \rho_{\ell} - j$, for $r - \rho_{\ell} < j \le r - \rho_{\ell-1}$.

By the Vandermonde formula

$$\prod_{1 \le i < j \le r} (t_i - t_j) = \det(t_j^{r-i})_{1 \le i,j \le r}$$

we have the following chain of equalities

$$\begin{aligned} &(\pi_{\rho})_* \tilde{F}(\xi_1, \dots, \xi_r) \\ &= \sum a_{\lambda} [t_1^{b_1} \cdots t_r^{b_r}] \left(t_1^{\lambda_1} \cdots t_r^{\lambda_r} \prod_{1 \le i \le r} s_{1/t_i} \det(t_j^{r-i})_{1 \le i,j \le r} \right) \\ &= \sum a_{\lambda} [t_1^{b_1} \cdots t_r^{b_r}] \left(\det \left(t_j^{\lambda_j + r - i} s_{1/t_j} \right)_{1 \le i,j \le r} \right). \end{aligned}$$

Moreover, for $M_{ij} \in H^{\bullet}(X)[t_j, t_j^{-1}]$ where $1 \leq i, j \leq r$, the identity

$$[t_1^{b_1}\cdots t_r^{b_r}]\big(\det(M_{ij})\big) = \det\left([t_j^{b_j}](M_{ij})\right)$$

(see [DP17, Lemma 4.1]) gives

$$(\pi_{\rho})_* \tilde{F}(\xi_1, \dots, \xi_r)$$

= $\sum a_{\lambda} \det \left([t_j^{b_j}] \left(t_j^{\lambda_j + r - i} s_{1/t_j} \right) \right)_{1 \le i, j \le r}$
= $\sum a_{\lambda} \det \left(s_{\lambda_j + r - i - b_j} \right)_{1 \le i, j \le r}.$

Since $s_{\lambda_j+r-i-b_j} = s_{\lambda_j-(r-\rho_\ell)+j-i} = s_{\lambda_j-\nu_j+j-i}$, we obtain a generalized Schur class after the following operations

$$\begin{split} \left(s_{\lambda_{j}-\nu_{j}+j-i}\right) &\mapsto \left(s_{\lambda_{i}-\nu_{i}+i-j}\right) & \text{by } i \leftrightarrow j \\ &\mapsto \left(s_{\lambda_{r-i+1}-\nu_{r-i+1}+(r-i+1)-j}\right) & \text{by } (r-i+1) \rightarrow i \\ &\mapsto \left(s_{\lambda_{r-i+1}-\nu_{r-i+1}+(r-i+1)-(r-j+1)}\right) & \text{by } (r-j+1) \rightarrow j \\ &\mapsto \left(s_{\lambda_{r-i+1}-\nu_{r-i+1}+j-i}\right) \end{split}$$

and we have

$$(\pi_{\rho})_* \tilde{F}(\xi_1, \dots, \xi_r) = \sum_{|\lambda|=d+k} a_\lambda \det \left(s_{\lambda_{r-i+1}-\nu_{r-i+1}+j-i} \right)_{1 \le i,j \le r}$$

which is Formula (2.3).

2.2 Push-forward formulæ for universal line bundles

Suppose that the vector bundle E is now endowed with a Hermitian metric h.

The aim of this section is to prove Theorem B. The main ingredient required is Proposition 2.7 below, whose proof is based on an intrinsic expression of the Chern curvature of the universal line bundles $Q_{\rho,j} \to \mathbb{F}_{\rho}(E)$, which are equipped with the Hermitian metrics induced by h as described in Section 1.2.2.

2.2.1 Curvature intrinsic expression: universal line bundles

Here we rewrite Formula (1.13) more intrinsically. In order to do this, observe that Remark 1.10 shows that, for $1 \le j \le m$, if we define

$$\Xi_{\rho,j}^{\mathrm{vert}} := \Xi_{\rho,j} \circ (p_1 \otimes \overline{p_1})$$

and

$$\Xi_{\rho,j}^{\mathrm{hor}} := \Xi_{\rho,j} \circ (p_2 \otimes \overline{p_2}),$$

which are both sections of $\Lambda^{1,1}T^{\vee}_{\mathbb{F}_{o}(E)}$, we have

$$\Xi_{\rho,j} = \Xi_{\rho,j}^{\text{vert}} + \Xi_{\rho,j}^{\text{hor}}.$$

Moreover, again with the same choice of coordinates, pointwise at (x_0, \mathbf{f}_0) we have

$$\left(\Xi_{\rho,j}^{\mathrm{hor}}\right)_{(x_0,\mathbf{f}_0)} = \frac{i}{2\pi} \sum_{\lambda=s_{j-1}+1}^{s_j} \Theta_{\lambda\lambda}(x_0), \qquad (2.4)$$

this follows from Formula (1.13).

Next, for $x \in X$, let $\mathbf{f} \in \mathbb{F}_{\rho}(E_x)$ be given by a unitary basis (v_1, \ldots, v_r) of E_x . Define a section

$$\theta_j \colon \mathbb{F}_{\rho}(E) \to \Lambda^{1,1} T^{\vee}_{\mathbb{F}_{\rho}(E)}$$

as follows

$$\theta_j(x, \mathbf{f}) = \frac{i}{2\pi} \sum_{\lambda = s_{j-1}+1}^{s_j} \langle \pi^* \Theta(E, h)_{(x, \mathbf{f})} \cdot v_\lambda, v_\lambda \rangle_h.$$

Lemma 2.4. The section θ_j is well defined, i.e. it does not depend upon the choice of a particular representative $\mathbf{v} = (v_1, \ldots, v_r)$ for \mathbf{f} .

Proof. Take a local normal frame (e_1, \ldots, e_r) of E at x such that $\mathbf{e} := (e_1(x), \ldots, e_r(x))$ and \mathbf{v} identify the same flag \mathbf{f} . For $\lambda = 1, \ldots, r$, we have

$$v_{\lambda} = a_{1\lambda}e_1(x) + \dots + a_{r\lambda}e_r(x) \tag{2.5}$$
where $A = (a_{pq})$ is the change of coordinates matrix. Since the unitary bases **e** and **v** give both the same flag, it follows that A is a block matrix with the following form

$$\begin{pmatrix} A_{11} & 0 & 0 \\ 0 & \ddots & 0 \\ 0 & 0 & A_{mm} \end{pmatrix},$$
(2.6)

where the diagonal block A_{jj} is again a unitary matrix of size $s_j - s_{j-1}$; in particular, for $1 \leq \lambda \leq r$ if the entry (λ, λ) hits the block A_{jj} , *i.e.* if $s_{j-1} < \lambda \leq s_j$, Formula (2.5) reads as

$$v_{\lambda} = \sum_{\alpha = s_{j-1}+1}^{s_j} a_{\alpha\lambda} e_{\alpha}(x).$$

Using the local normal frame \mathbf{e} (and omitting its dependence on x to simplify the notation), we get

$$\begin{split} &\frac{i}{2\pi}\sum_{\lambda=s_{j-1}+1}^{s_j} \left\langle \pi^*\Theta(E,h)_{(x,\mathbf{f})} \cdot v_{\lambda}, v_{\lambda} \right\rangle_h \\ &= \frac{i}{2\pi}\sum_{\lambda=s_{j-1}+1}^{s_j} \left\langle \pi^*\Theta(E,h)_{(x,\mathbf{f})} \cdot \left(\sum_{\alpha=s_{j-1}+1}^{s_j} a_{\alpha\lambda}e_{\alpha}\right), \sum_{\tilde{\alpha}=s_{j-1}+1}^{s_j} a_{\tilde{\alpha}\lambda}e_{\tilde{\alpha}} \right\rangle_h \\ &= \frac{i}{2\pi}\sum_{\alpha,\tilde{\alpha}=s_{j-1}+1}^{s_j} \sum_{\lambda=s_{j-1}+1}^{s_j} a_{\alpha\lambda}\overline{a_{\tilde{\alpha}\lambda}} \left\langle \pi^*\Theta(E,h)_{(x,\mathbf{f})} \cdot e_{\alpha}, e_{\tilde{\alpha}} \right\rangle_h \\ &= \frac{i}{2\pi}\sum_{\alpha=s_{j-1}+1}^{s_j} \left\langle \pi^*\Theta(E,h)_{(x,\mathbf{f})} \cdot e_{\alpha}, e_{\alpha} \right\rangle_h, \end{split}$$

and the lemma follows.

Of course, θ_i is smooth, hence it is a form in $\mathcal{A}^{1,1}(\mathbb{F}_{\rho}(E))$.

Lemma 2.5. The equality

$$\theta_j = \Xi_{\rho,j}^{\text{hor}}$$

holds.

Proof. At any given $(x, \mathbf{f}) \in \mathbb{F}_{\rho}(E)$, choose (e_1, \ldots, e_r) to be a local normal frame for E at x such that \mathbf{f} is given by $(e_1(x), \ldots, e_r(x))$, and consider the induced holomorphic coordinates around (x, \mathbf{f}) as in Section 1.2.1.

Since the evaluation of θ_j in (x, \mathbf{f}) does not depend on the choice of the unitary basis defining \mathbf{f} , we have the following chain of equalities

$$\begin{aligned} \theta_j(x, \mathbf{f}) &= \frac{i}{2\pi} \sum_{\lambda=s_{j-1}+1}^{s_j} \left\langle \pi^* \Theta(E, h)_{(x, \mathbf{f})} \cdot e_\lambda(x), e_\lambda(x) \right\rangle_h \\ &= \frac{i}{2\pi} \sum_{\lambda=s_{j-1}+1}^{s_j} \pi^* \Theta_{\lambda\lambda}(x, \mathbf{f}) \\ &= \frac{i}{2\pi} \sum_{\lambda=s_{j-1}+1}^{s_j} \Theta_{\lambda\lambda}(x) \\ &= \left(\Xi_{\rho, j}^{\text{hor}}\right)_{(x, \mathbf{f})}, \end{aligned}$$

where the last equality follows from Formula (2.4).

Now, in order to simplify the notation in what follows, let us relabel $\omega_j := \Xi_{\rho,j}^{\text{vert}}$. The section ω_j is smooth since, for instance, by Lemma 2.5 it equals $\Xi_{\rho,j} - \theta_j$, which are two smooth (1, 1)-forms on $\mathbb{F}_{\rho}(E)$.

Lemma 2.4 and Lemma 2.5 prove the following.

Proposition 2.6. For $1 \le j \le m$, we have the equality

$$\Xi_{\rho,j} = \theta_j + \omega_j. \tag{2.7}$$

In other words, we have shown that the first Chern form of $(Q_{\rho,j}, h)$ can be written as a sum of a horizontal and a vertical part with respect to the natural splitting defined in Section 1.2.3, where the horizontal part contains the information coming from the curvature of (E, h) while the vertical part is the standard curvature of the determinant of the tautological successive quotients on a flag manifold.

2.2.2 Gysin formulæ for the curvature of universal line bundles

We are now in a good position to prove the following main technical proposition.

Proposition 2.7. Let $F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m})$ be a complex homogeneous polynomial in the (1,1)-forms $\Xi_{\rho,1},\ldots,\Xi_{\rho,m}$ on $\mathbb{F}_{\rho}(E)$. Then the push-forward

$$(\pi_{\rho})_*F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m})$$

is given by a universal (weighted) homogeneous polynomial formally evaluated in the Chern forms of (E, h).

In the statement, by *universal* we mean that this polynomial depends only on the shape of F and on the rank of E.

Proof. In order to simplify the notation, denote by π the projection π_{ρ} of the flag bundle $\mathbb{F}_{\rho}(E)$, and by $F(\Xi)$ the differential form $F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m})$. As usual, let d_{ρ} be the relative dimension of the flag bundle. Write $F(\Xi)$ as

$$\sum_{j_1+\dots+j_m=d_\rho+k} a_{(j_1,\dots,j_m)} \Xi^{j_1}_{\rho,1} \wedge \dots \wedge \Xi^{j_m}_{\rho,m}, \quad a_{(j_1,\dots,j_m)} \in \mathbb{C},$$

where we can w.l.o.g. suppose that $0 \le k \le n$, otherwise the push-forward would be identically zero for obvious degree reasons. As we only consider even degree forms, to simplify the notation we omit the symbol \wedge for the wedge product. From now on, we use, where useful, the multi-index notation.

Thanks to Formula (2.7), we can write

$$F(\Xi) = \sum_{|J|=d_{\rho}+k} a_{J} (\theta_{1} + \omega_{1})^{j_{1}} \dots (\theta_{m} + \omega_{m})^{j_{m}}$$

$$= \sum_{|J|=d_{\rho}+k} a_{J} \sum_{b_{1}=0}^{j_{1}} {j_{1} \choose b_{1}} \theta_{1}^{j_{1}-b_{1}} \omega_{1}^{b_{1}} \dots \sum_{b_{m}=0}^{j_{m}} {j_{m} \choose b_{m}} \theta_{m}^{j_{m}-b_{m}} \omega_{m}^{b_{m}}$$

$$= \sum_{|J|=d_{\rho}+k} a_{J} \sum_{b_{1}=0}^{j_{1}} \dots \sum_{b_{m}=0}^{j_{m}} {j_{1} \choose b_{1}} \dots {j_{m} \choose b_{m}} \theta_{1}^{j_{1}-b_{1}} \dots \theta_{m}^{j_{m}-b_{m}} \omega_{1}^{b_{1}} \dots \omega_{m}^{b_{m}}.$$

Since π is a proper submersion, and by definition the push-forward $\pi_*F(\Xi)$ is given by integration along the fibers obtained by locally splitting the variables (x, \mathbf{f}) , at $x \in X$, we have that

$$\pi_* F(\Xi)_x = \int_{\mathbf{f} \in \mathbb{F}_{\rho}(E_x)} F(\Xi)_{(x,\mathbf{f})},$$

where the second term stands, as already mentioned, for the integral of $F(\Xi)_{(x,\mathbf{f})}$ performed only on those differentials related to the variable \mathbf{f} . Therefore, for degree reasons, the only terms which can possibly survive after integration along the fibers are those for which $b_1 + \cdots + b_m = d_\rho$, since the ω_j 's and only the ω_j 's contain the relevant vertical differentials.

For the push-forward we thus obtain

$$\pi_*F(\Xi)_x = \sum_{\substack{|J|=d_\rho+k\\b_1=0,\dots,j_1\\b_1=0,\dots,j_m\\b_1+\dots+b_m=d_\rho}} a_J\binom{j_1}{b_1}\dots\binom{j_m}{b_m} \int_{\mathbb{F}_\rho(E_x)} \underbrace{\theta_1^{j_1-b_1}\dots\theta_m^{j_m-b_m}}_{=:\theta^{J-B}} \underbrace{\omega_1^{b_1}\dots\omega_m^{b_m}}_{=:\omega^B}.$$

What we want to do now is to explicitly write θ^{J-B} at an arbitrary point $(x, \mathbf{f}) \in \mathbb{F}_{\rho}(E_x)$. Let **f** be given by a unitary basis (v_1, \ldots, v_r) of E_x , so that

$$\theta_{\ell}(x, \mathbf{f}) = \frac{i}{2\pi} \sum_{\lambda = s_{\ell-1}+1}^{s_{\ell}} \langle \pi^* \Theta(E, h)_{(x, \mathbf{f})} \cdot v_{\lambda}, v_{\lambda} \rangle_h.$$

Suppose as above we had fixed a local normal frame (e_1, \ldots, e_r) for E centered at $x \in X$, and let $v_{\lambda} = \sum_{\nu} v_{\lambda}^{\nu} e_{\nu}(x)$. Thus, we have

$$\theta_{\ell}(x, \mathbf{f}) = \frac{i}{2\pi} \sum_{\alpha, \beta=1}^{r} \left(\sum_{\lambda=s_{\ell-1}+1}^{s_{\ell}} v_{\lambda}^{\alpha} \bar{v}_{\lambda}^{\beta} \right) \Theta_{\beta\alpha}(x).$$
(2.8)

Then, we get the following expression for θ^{J-B} :

$$\theta^{J-B} = \left(\frac{i}{2\pi}\right)^k \bigwedge_{\ell=1}^m \left(\sum_{\alpha_\ell,\beta_\ell=1}^r \left(\sum_{\lambda_\ell=s_{\ell-1}+1}^{s_\ell} v_{\lambda_\ell}^{\alpha_\ell} \bar{v}_{\lambda_\ell}^{\beta_\ell}\right) \Theta_{\beta_\ell \alpha_\ell}\right)^{j_\ell - b_\ell}$$

$$= \left(\frac{i}{2\pi}\right)^k \sum_{\substack{\alpha_1^1,\beta_1^1,\dots,\alpha_1^{j_1-b_1},\beta_1^{j_1-b_1}=1,\dots,r\\\alpha_m^1,\beta_m^1,\dots,\alpha_m^{j_m-b_m},\beta_m^{j_m-b_m}=1,\dots,r}} BQ_{\beta_1^1\dots\beta_m^{j_m-b_m}}^{\alpha_1^1\dots\alpha_m^{j_m-b_m}} \Theta_{\beta_1^1\alpha_1^1}\dots\Theta_{\beta_m^{j_m-b_m}\alpha_m^{j_m-b_m}},$$

where

$${}^{B}Q_{\beta_{1}^{1}\dots\beta_{m}^{j_{m}-b_{m}}}^{\alpha_{1}^{1}\dots\alpha_{m}^{j_{m}-b_{m}}} = \prod_{\ell=1}^{m} \left[\left(\sum_{\lambda_{\ell}=s_{\ell-1}+1}^{s_{\ell}} v_{\lambda_{\ell}}^{\alpha_{\ell}^{1}} \bar{v}_{\lambda_{\ell}}^{\beta_{\ell}} \right) \dots \left(\sum_{\lambda_{\ell}=s_{\ell-1}+1}^{s_{\ell}} v_{\lambda_{\ell}}^{\alpha_{\ell}^{j_{\ell}-b_{\ell}}} \bar{v}_{\lambda_{\ell}}^{\beta_{\ell}^{j_{\ell}-b_{\ell}}} \right) \right].$$

Remark that the $\Theta_{\beta\alpha}$'s only depend on the point x, while the v_{λ}^{ν} 's can be seen, by a slight abuse of notation, as variables of integration even if they have to be understood modulo the action (cf. with the construction of the matrix A in the proof of Lemma 2.4) of

$$U(s_1 - s_0) \times U(s_2 - s_1) \times \dots \times U(s_m - s_{m-1}) \subset U(\underbrace{s_m - s_0}_{=r})$$

which of course corresponds to the homogeneous presentation of the (possibly incomplete) flag manifold as $U(r)/U(s_1 - s_0) \times \cdots \times U(s_m - s_{m-1})$.

At the end of the day, the integrals $\int_{\mathbb{F}_{\rho}(E_x)} \theta^{J-B} \omega^B$ are given by the following sum:

$$\left(\frac{i}{2\pi}\right)^{k} \sum_{\substack{\alpha_{1}^{1},\ldots,\alpha_{m}^{j_{m}-b_{m}}=1,\ldots,r\\\beta_{1}^{1},\ldots,\beta_{m}^{j_{m}-b_{m}}=1,\ldots,r}} {}^{B} q_{\beta_{1}^{1}\ldots\beta_{m}^{j_{m}-b_{m}}}^{\alpha_{1}^{1}\ldots\alpha_{m}^{j_{m}-b_{m}}} \Theta_{\beta_{1}^{1}\alpha_{1}^{1}}\ldots\Theta_{\beta_{m}^{j_{m}-b_{m}}\alpha_{m}^{j_{m}-b_{m}}},$$

where

$${}^{B}q_{\beta_{1}^{1}\ldots\beta_{m}^{jm-bm}}^{\alpha_{1}^{1}\ldots\alpha_{m}^{jm-bm}} = \int_{\mathbb{F}_{\rho}(E_{x})} {}^{B}Q_{\beta_{1}^{1}\ldots\beta_{m}^{jm-bm}}^{\alpha_{1}^{1}\ldots\alpha_{m}^{jm-bm}} \omega^{B}.$$

The good news is that these coefficients ${}^{B}q_{\beta_{1}^{1}\dots\beta_{m}^{j_{m}-b_{m}}}^{\alpha_{1}^{1}\dots\alpha_{m}^{j_{m}-b_{m}}}$ are universal, in the following sense: they do not depend anymore on the metric h, nor on the point $x \in X$, but only on the multi-index B and on the rank r of E.

Indeed, they might be calculated in the "absolute" case of the flag manifold $\mathbb{F}_{\rho}(\mathbb{C}^r)$, where \mathbb{C}^r is endowed with the standard Euclidean metric, the v_{λ}^{ν} 's are the element of a matrix in U(r) representing the given flag, and where the top form ω^B is nothing else than the corresponding wedge product of the curvature forms of the determinant of the tautological quotients line bundles on $\mathbb{F}_{\rho}(\mathbb{C}^r)$ with respect to the natural metrics induced by the Euclidean metric of \mathbb{C}^r .

Summing up, we have shown that

$$\pi_* F(\Xi)_x = \sum_{\substack{|J|=d_\rho+k \\ b_1=0,\dots,j_1 \\ b_m=0,\dots,j_n \\ b_m=0,\dots,j_m \\ b_1+\dots+b_m=d_\rho}} \sum_{\substack{\alpha_1^1,\dots,\alpha_m^{j_m-b_m}=1,\dots,r \\ \beta_1^1,\dots,\beta_m^{j_m-b_m}=1,\dots,r \\ k_m=0,\dots,j_m \\ \times \left(\frac{i}{2\pi}\Theta_{\beta_1^1}\alpha_1^1\right)\dots\left(\frac{i}{2\pi}\Theta_{\beta_m^{j_m-b_m}}\alpha_m^{j_m-b_m}\right).$$

The above expression is thus given by evaluating a homogeneous polynomial \hat{P} of degree

$$(j_1 - b_1) + (j_2 - b_2) + \dots + (j_m - b_m) = |J| - |B| = k$$

on the entries $\Theta_{\beta\alpha}$'s of the matrix associated to the curvature $\Theta(E, h)_x$ with respect to the frame $(e_1(x), \ldots, e_r(x))$.

This polynomial (*i.e.* its coefficients) is clearly independent of the point $x \in X$. Moreover, \hat{P} is of course also invariant under change of frame at x, because from its very definition in terms of push-forwards it is independent of the local frame chosen to make the computation. It follows, say from Chern–Weil theory (see for instance [GH78, p. 402]), that there exists a weighted homogeneous polynomial P such that globally on X

$$\pi_* F(\Xi) = P(c_1(E,h), \dots, c_r(E,h)), \tag{2.9}$$

where the $c_j(E, h)$'s are the Chern forms of (E, h). By construction, P is universal since it obviously depends only on F and the rank of E.

Remark 2.8. Of course, since Chern forms may be expressed in terms of Segre forms, a completely analogous statement holds with a polynomial whose variables are now the Segre forms $s_i(E,h)$ of (E,h).

We shall see now how to compute the push-forward through the projection π_{ρ} of the differential form given by a complex homogeneous polynomial F in $\Xi_{\rho,1}, \ldots, \Xi_{\rho,m}$.

Remark 2.9. Of course, for our purposes, it shall suffice to use some of the possible polynomials \tilde{F} considered in Formula (2.1) only, namely those whose symmetries are of the form

$$\tilde{F}(\xi_1,\ldots,\xi_r)=F\left(\ldots,-\sum_{\ell=s_{j-1}+1}^{s_j}\xi_\ell,\ldots\right)=F(\ldots,c_1(Q_{\rho,\ell}),\ldots).$$

Observe however, that in the special case of complete flag bundles (m = r), the classes ξ_1, \ldots, ξ_r are not virtual and give actual cohomology classes. Indeed, for $1 \leq j \leq r$, we have the equality $\xi_j = -c_1(Q_j)$. Hence, the only relevant symmetry needed in this case is $\tilde{F} = (-1)^{d+k} F$.

The key result of this section is that, in fact, Darondeau–Pragacz formula (2.1) also holds pointwise at the level of differential forms, in the Hermitian setting, for polynomials \tilde{F} satisfying the further symmetries considered in Remark 2.9.

Theorem 2.10. Let (E, h) be a rank r Hermitian holomorphic vector bundle over a complex manifold X of dimension n, and let F be a complex homogeneous polynomial of degree $d_{\rho} + k$ in m variables. Then, we have the equality

$$(\pi_{\rho})_*F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m})=\Psi(s_1(E,h),\ldots,s_n(E,h)).$$

Clearly, in the statement above, Ψ is the polynomial introduced in Remark 2.1 associated to the polynomial $\tilde{F}(\xi_{\bullet}) = F(c_1(Q_{\rho,\bullet}))$, as in Remark 2.9.

Proof. As in Proposition 2.7 denote by π the projection π_{ρ} to simplify the notation.

If $\eta \in \mathcal{A}^{\bullet}(X)$ is a closed form, then $[\eta]$ stands as usual for the cohomology class in $H^{\bullet}(X)$ represented by η . By Formula (2.1), it holds that

$$[\pi_* F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m})] = \pi_* [F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m})]$$

= $\pi_* \tilde{F}(\xi_1,\ldots,\xi_r)$
= $\Psi(s_1,\ldots,s_n)$
= $[\Psi(s_1(E,h),\ldots,s_n(E,h))].$

Hence, the difference

$$\pi_*F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m}) - \Psi(s_1(E,h),\ldots,s_n(E,h))$$

must be an exact global (k, k)-form on X. Recall that, by Proposition 2.7 and Formula (2.9), $\pi_*F(\Xi_{\rho,1}, \ldots, \Xi_{\rho,m})$ is a universal weighted homogeneous polynomial $P = P(c_{\bullet}(E, h))$ of weighted degree 2k in the Chern forms of (E, h). If we express the Segre forms in terms of the Chern forms, the previous difference can be written as a complex weighted homogeneous polynomial

$$G(c_1(E,h),\ldots,c_r(E,h)) = \sum_{k_1+2k_2+\cdots+rk_r=k} g_{k_1\ldots k_r} c_1(E,h)^{k_1} \wedge \cdots \wedge c_r(E,h)^{k_r}$$

in the Chern forms. Note that G is universal (since P and Ψ are) in the sense that its coefficients $g_{k_1...k_r}$ do not depend upon E, nor X, but only upon r, n, and F. Recall that our aim is to show that G is in fact identically zero: following [Gul12], to achieve this we shall evaluate it on a particular vector bundle on a particular class of manifolds, as follows.

Take X to be any *n*-dimensional *projective* manifold and fix an ample line bundle A on X. Let ω_A be a metric on A with positive curvature. For m_1, \ldots, m_r positive integers, we define the totally split, rank r vector bundle

$$\mathcal{E} := A^{\otimes m_1} \oplus \cdots \oplus A^{\otimes m_r},$$

and denote by $\omega_{\mathcal{E}}$ the natural induced metric on \mathcal{E} by ω_A . Hence,

$$G(c_1(\mathcal{E}, \omega_{\mathcal{E}}), \dots, c_r(\mathcal{E}, \omega_{\mathcal{E}})))$$

$$= \sum_{k_1+2k_2+\dots+rk_r=k} g_{k_1\dots k_r} c_1(\mathcal{E}, \omega_{\mathcal{E}})^{k_1} \wedge \dots \wedge c_r(\mathcal{E}, \omega_{\mathcal{E}})^{k_r}$$

$$= \sum_{k_1+2k_2+\dots+rk_r=k} g_{k_1\dots k_r} \bigwedge_{s=1}^r \left(\sum_{1 \le j_1 < \dots < j_s \le r} m_{j_1} \cdots m_{j_s} c_1(A, \omega_A)^s \right)^{k_s}$$

$$= \left[\sum_{k_1+2k_2+\dots+rk_r=k} g_{k_1\dots k_r} \prod_{s=1}^r \left(\sum_{1 \le j_1 < \dots < j_s \le r} m_{j_1} \cdots m_{j_s} \right)^{k_s} \right] c_1(A, \omega_A)^k.$$

Let T_1, \ldots, T_r be a set of formal variables, and consider the polynomial p defined by

$$p(T_1, \dots, T_r) = \sum_{k_1 + 2k_2 + \dots + rk_r = k} g_{k_1 \dots k_r} \prod_{s=1}^r \left(\sum_{1 \le j_1 < \dots < j_s \le r} T_{j_1} \dots T_{j_s} \right)^{k_s}.$$
 (2.10)

We have by definition that

$$G(c_1(\mathcal{E},\omega_{\mathcal{E}}),\ldots,c_r(\mathcal{E},\omega_{\mathcal{E}})) = p(m_1,\ldots,m_r)c_1(A,\omega_A)^k$$

and, consequently, in cohomology it holds that

$$p(m_1,\ldots,m_r)c_1(A)^k = [G(c_1(\mathcal{E},\omega_{\mathcal{E}}),\ldots,c_r(\mathcal{E},\omega_{\mathcal{E}}))] = 0.$$

Since A is ample, the only possibility is that the polynomial p is zero for every choice of positive integers m_1, \ldots, m_r . But the set of points in \mathbb{C}^r whose coordinates are positive integers is Zariski dense, and thus p must be identically zero. Consequently, having the same coefficients as $p, G \equiv 0$ and this concludes the proof.

Remark 2.11. As already said, Formula (2.1) is not the only possible instance of the Gysin formula given by Darondeau and Pragacz.

For example, we have already seen in Proposition 2.2 a universal Gysin formula (see Formula (2.3)) for flag bundles in terms of generalized Schur classes. Such formula is particularly useful since for instance it explicitly shows that one can obtain all the Schur polynomials in the Chern classes of E as push-forwards from the complete flag bundle (see Proposition 2.21 below for more details).

Clearly, our Theorem 2.10 proves the validity of Formula (2.3) at the level of differential forms for those \tilde{F} as in Remark 2.9. We return on the extension of Formula (2.3) at the forms level in Section 2.4.

2.3 Push-forward formulæ for universal vector bundles

In this section, we prove Theorem A, which is an extension of the pointwise universal Gysin formula in Theorem 2.10 to polynomials in the Chern forms of all the universal vector bundles. The strategy of the proof is similar but, despite Theorem 2.10 where we use the curvature formulæ for the determinants of the successive quotients provided by [Dem88a], we need here curvature formulæ for all universal vector bundles. These formulæ are the core of this section, and we give them in the following.

2.3.1 Curvature of universal vector bundles

Given a dimension sequence $\rho = (\rho_0, \ldots, \rho_m)$, consider the flag bundle $\pi_\rho \colon \mathbb{F}_\rho(E) \to X$. In the notation of Section 1.2, we now compute the Chern curvature tensors

$$\Theta(U_{\rho,l}, h_{\rho,l}) \quad \Theta(\pi_{\rho}^* E/U_{\rho,l}, H_{\rho,(m,l)}) \quad \Theta(U_{\rho,l}/U_{\rho,\ell}, H_{\rho,(l,\ell)})$$

of the universal vector bundles of $\mathbb{F}_{\rho}(E)$ endowed with the Hermitian metrics induced by h as in Remark 1.4.

Theorem 2.12. Choose a point $(x_0, \mathbf{f}_0) \in \mathbb{F}_{\rho}(E)$ and let (e_1, \ldots, e_r) be a local normal frame of (E, h) at x_0 such that $\mathbf{f}_0 = [e_1(x_0), \ldots, e_r(x_0)]_{\rho}$. Fix the local holomorphic coordinates (z, ζ) centered at (x_0, \mathbf{f}_0) introduced in Section 1.2.1. Then:

(i) for any $0 < l \leq m$, the curvature $\Theta(U_{\rho,l}, h_{\rho,l})_{(x_0, \mathbf{f}_0)}$ equals

$$\sum_{r-\rho_l<\alpha,\beta\leq r} \left(\Theta_{\beta\alpha} - \sum_{1\leq\lambda\leq r-\rho_l} d\zeta_{\lambda\alpha} \wedge d\bar{\zeta}_{\lambda\beta}\right) \otimes \epsilon_{\alpha}^{\vee} \otimes \epsilon_{\beta};$$

(ii) for any 0 < l < m, the curvature $\Theta(\pi_{\rho}^* E/U_{\rho,l}, H_{\rho,(m,l)})_{(x_0, f_0)}$ equals

$$\sum_{1 \le \alpha, \beta \le r - \rho_l} \left(\Theta_{\beta \alpha} + \sum_{r - \rho_l < \mu \le r} d\zeta_{\beta \mu} \wedge d\bar{\zeta}_{\alpha \mu} \right) \otimes \tilde{e}_{\alpha}^{\vee} \otimes \tilde{e}_{\beta};$$

(iii) for any $0 < \ell < l < m$, the curvature $\Theta(U_{\rho,l}/U_{\rho,\ell}, H_{\rho,(l,\ell)})_{(x_0,\mathbf{f}_0)}$ equals

$$\sum_{r-\rho_l < \alpha, \beta \le r-\rho_\ell} \left(\Theta_{\beta\alpha} - \sum_{1 \le \lambda \le r-\rho_l} d\zeta_{\lambda\alpha} \wedge d\bar{\zeta}_{\lambda\beta} + \sum_{r-\rho_\ell < \mu \le r} d\zeta_{\beta\mu} \wedge d\bar{\zeta}_{\alpha\mu} \right) \otimes \tilde{\epsilon}_{\alpha}^{\vee} \otimes \tilde{\epsilon}_{\beta}.$$

By a slight abuse of notation, we have identified the (1, 1)-form $\Theta_{\beta\alpha}$ with its pull-back $\pi^*_{\rho}\Theta_{\beta\alpha}$ along π_{ρ} .

Proof of Theorem 2.12 (i). Set $s := \rho_l$. By Remark 1.7 the curvature $\Theta(U_{\rho,l}, h_{\rho,l})$ is just the pull-back through the projection $\pi^{\rho}_{(0,s,r)}$ of the curvature $\Theta(\gamma_s, h_s)$ of the tautological vector bundle $(\gamma_s, h_s) \to \mathbb{G}_s(E)$. Thus, we can suppose that $\mathbb{F}_{\rho}(E) = \mathbb{G}_s(E)$ and that $U_{\rho,l} = \gamma_s$.

A local holomorphic frame for γ_s is given by the sections

$$\epsilon_{\alpha}(z,\zeta) = e_{\alpha}(z) + \sum_{1 \le \lambda \le r-s} \zeta_{\lambda \alpha} e_{\lambda}(z), \quad \text{for } r-s < \alpha \le r.$$

Observe that, for the sequence (0, s, r), Condition (*) on λ and α for the coordinate $\zeta_{\lambda\alpha}$ becomes $1 \leq \lambda \leq r - s < \alpha \leq r$. We want to show that $(\epsilon_{r-s+1}, \ldots, \epsilon_r)$ is also a local normal frame of (γ_s, h_s) at (x_0, \mathbf{f}_0) .

To simplify the notation in what follows, we omit the dependence on the coordinates. The (α, β) -entry of the Hermitian matrix associated to h_s with respect to the coordinates $y := (z, \zeta)$ is

$$\langle \epsilon_{\alpha}, \epsilon_{\beta} \rangle$$

$$= \left\langle e_{\alpha} + \sum_{\lambda} \zeta_{\lambda\alpha} e_{\lambda}, e_{\beta} + \sum_{\mu} \zeta_{\mu\beta} e_{\mu} \right\rangle$$

$$= \left\langle e_{\alpha}, e_{\beta} \right\rangle + \sum_{\mu} \bar{\zeta}_{\mu\beta} \left\langle e_{\alpha}, e_{\mu} \right\rangle + \sum_{\lambda} \zeta_{\lambda\alpha} \left\langle e_{\lambda}, e_{\beta} \right\rangle + \sum_{\lambda,\mu} \zeta_{\lambda\alpha} \bar{\zeta}_{\mu\beta} \left\langle e_{\lambda}, e_{\mu} \right\rangle$$

$$= \delta_{\alpha\beta} - \sum_{j,k} c_{jk\alpha\beta} z_{j} \bar{z}_{k} + \sum_{\mu} \bar{\zeta}_{\mu\beta} \delta_{\alpha\mu} + \sum_{\lambda} \zeta_{\lambda\alpha} \delta_{\lambda\beta} + \sum_{\lambda,\mu} \zeta_{\lambda\alpha} \bar{\zeta}_{\mu\beta} \delta_{\lambda\mu} + O(|y|^{3}).$$

$$(2.11)$$

Now, observe that we can get rid of the terms $\sum \bar{\zeta}_{\mu\beta}\delta_{\alpha\mu}$ and $\sum \zeta_{\lambda\alpha}\delta_{\lambda\beta}$. Indeed, $\sum_{\mu} \bar{\zeta}_{\mu\beta}\delta_{\alpha\mu} = \bar{\zeta}_{\alpha\beta}$ but $\zeta_{\alpha\beta} = 0$ since Condition (*) on α and β contradicts the fact that the index α is greater than r - s. The same holds for β in the term $\sum_{\lambda} \zeta_{\lambda\alpha}\delta_{\lambda\beta}$. We have shown that

$$\langle \epsilon_{\alpha}, \epsilon_{\beta} \rangle = \delta_{\alpha\beta} - \sum_{1 \le j,k \le n} c_{jk\alpha\beta} z_j \bar{z}_k + \sum_{1 \le \lambda \le r-s} \zeta_{\lambda\alpha} \bar{\zeta}_{\lambda\beta} + \mathcal{O}(|y|^3)$$

thus, $(\epsilon_{r-s+1}, \ldots, \epsilon_r)$ is a normal coordinate frame at (x_0, \mathbf{f}_0) .

Therefore, the (β, α) -entry of the curvature matrix $\Theta(\gamma_s, h_s)_{(x_0, \mathbf{f}_0)}$ with respect to the coordinates y is

$$\sum_{1 \le j,k \le n} c_{jk\alpha\beta} dz_j \wedge d\bar{z}_k - \sum_{1 \le \lambda \le r-s} d\zeta_{\lambda\alpha} \wedge d\bar{\zeta}_{\lambda\beta}$$

and we have finished.

Proof of Theorem 2.12 (ii). Set $s := \rho_l$ as before. By Remark 1.7 we can suppose that $(\pi_{\rho}^* E/U_{\rho,l}, H_{\rho,(m,l)})$ is the universal quotient (Q_s, H_s) over $\mathbb{G}_s(E)$.

The local holomorphic coordinates (z, ζ) on $\mathbb{G}_s(E)$ centered at (x_0, \mathbf{f}_0) parameterize the *s*-planes $W \subset E_z$. Therefore, the local holomorphic frame (1.7) for Q_s is

$$\tilde{e}_{\alpha}(z,\zeta) = \text{image of } e_{\alpha}(z) \text{ in } E_z/W, \text{ for } 1 \leq \alpha \leq r-s.$$

Let

$$0 \to \gamma_s \stackrel{\iota}{\hookrightarrow} \pi_\rho^* E \stackrel{p}{\twoheadrightarrow} Q_s \to 0$$

be the tautological exact sequence over $\mathbb{G}_s(E)$, and denote by $p^*: Q_s \to \pi_\rho^* E$ the C^∞ orthogonal splitting of the natural projection p. We have

$$p^{\star} \cdot \tilde{e}_{\alpha}(z,\zeta) = e_{\alpha}(z) + \sum_{r-s < \mu \le r} u_{\alpha\mu}(z,\zeta) \epsilon_{\mu}(z,\zeta),$$

for some smooth functions $u_{\alpha\mu}$. To simplify the notation in what follows, we omit the dependence on the coordinates and we set $y := (z, \zeta)$. For $r - s < \beta$, the following chain of equalities hold

$$\begin{split} 0 &= \langle \tilde{e}_{\alpha}, p \cdot \epsilon_{\beta} \rangle \\ &= \langle p^{*} \cdot \tilde{e}_{\alpha}, \epsilon_{\beta} \rangle \\ &= \left\langle e_{\alpha} + \sum_{r-s < \mu} u_{\alpha\mu} \epsilon_{\mu}, e_{\beta} + \sum_{\nu \leq r-s} \zeta_{\nu\beta} e_{\nu} \right\rangle \\ &= \left\langle e_{\alpha} + \sum_{r-s < \mu} u_{\alpha\mu} e_{\mu} + \sum_{\substack{r-s < \mu \\ \sigma \leq r-s}} u_{\alpha\mu} \zeta_{\sigma\mu} e_{\sigma}, e_{\beta} + \sum_{\nu \leq r-s} \zeta_{\nu\beta} e_{\nu} \right\rangle \\ &= \langle e_{\alpha}, e_{\beta} \rangle + \sum_{\nu \leq r-s} \bar{\zeta}_{\nu\beta} \langle e_{\alpha}, e_{\nu} \rangle + \sum_{\substack{r-s < \mu \\ \sigma \leq r-s}} u_{\alpha\mu} \langle e_{\mu}, e_{\beta} \rangle + \sum_{\substack{r-s < \mu \\ \nu \leq r-s}} u_{\alpha\mu} \zeta_{\sigma\mu} \langle e_{\sigma}, e_{\beta} \rangle \\ &+ \sum_{\substack{r-s < \mu \\ \sigma \leq r-s}} u_{\alpha\mu} \zeta_{\sigma\mu} \langle e_{\sigma}, e_{\beta} \rangle + \sum_{\substack{r-s < \mu \\ \sigma \leq r-s}} u_{\alpha\mu} \zeta_{\sigma\mu} \bar{\zeta}_{\nu\beta} \langle e_{\sigma}, e_{\nu} \rangle \\ &= \delta_{\alpha\beta} - \sum_{j,k} c_{jk\alpha\beta} z_{j} \bar{z}_{k} + \bar{\zeta}_{\alpha\beta} + u_{\alpha\beta} \\ &+ \sum_{r-s < \mu} u_{\alpha\mu} \left(-\sum_{j,k} c_{jk\mu\beta} z_{j} \bar{z}_{k} + \sum_{\nu \leq r-s} \zeta_{\nu\mu} \bar{\zeta}_{\nu\beta} \right) + O(|y|^{3}). \end{split}$$

Therefore, $u_{\alpha\beta} = -\bar{\zeta}_{\alpha\beta} + O(|y|^2)$ and consequently,

$$p^{\star} \tilde{e}_{\alpha} = e_{\alpha} - \sum_{r-s < \mu} \left(\bar{\zeta}_{\alpha\mu} + \mathcal{O}(|y|^2) \right) \epsilon_{\mu}.$$

If $b \in \mathcal{A}^{1,0}(\mathbb{G}_s(E), \operatorname{Hom}(\gamma_s, Q_s))$ denotes the second fundamental form of γ_s in $\pi_{\rho}^* E$ and $b^* \in \mathcal{A}^{0,1}(\mathbb{G}_s(E), \operatorname{Hom}(Q_s, \gamma_s))$ is the adjoint of b, then $-\iota \circ b^* = \bar{\partial}p^*$. Hence,

$$-\iota \circ b^{\star} \cdot \tilde{e}_{\alpha}|_{(x_0, \mathbf{f}_0)} = \bar{\partial} p^{\star} \tilde{e}_{\alpha}|_{(0, 0)} = -\sum_{r-s < \mu} d\bar{\zeta}_{\alpha \mu} \otimes \epsilon_{\mu}.$$

We can write

$$b^{\star}_{(x_0,\mathbf{f}_0)} = \sum_{\substack{\alpha \leq r-s \\ r-s < \mu}} d\bar{\zeta}_{\alpha\mu} \otimes \tilde{e}^{\vee}_{\alpha} \otimes \epsilon_{\mu}, \quad b_{(x_0,\mathbf{f}_0)} = \sum_{\substack{\beta \leq r-s \\ r-s < \lambda}} d\zeta_{\beta\lambda} \otimes \epsilon^{\vee}_{\lambda} \otimes \tilde{e}_{\beta}.$$

Therefore, the curvature $\Theta(Q_s, H_s)_{(x_0, \mathbf{f}_0)}$ with respect to the coordinates y is

$$\left(\Theta(\pi_{\rho}^{*}E, \pi_{\rho}^{*}h) \big|_{Q_{s}} + b \wedge b^{*} \right)_{(x_{0}, \mathbf{f}_{0})}$$

$$= \sum_{\alpha, \beta} \left(\sum_{1 \leq j,k \leq n} c_{jk\alpha\beta} dz_{j} \wedge d\bar{z}_{k} + \sum_{r-s < \mu \leq r} d\zeta_{\beta\mu} \wedge d\bar{\zeta}_{\alpha\mu} \right) \otimes \tilde{e}_{\alpha}^{\vee} \otimes \tilde{e}_{\beta}$$

and we have done.

Proof of Theorem 2.12 (iii). We are left only with the quotient of two proper tautological sub-bundles. Again by Remark 1.7 it is sufficient to compute the curvature of $(Q_{\mathbf{s}}, H_{\mathbf{s}})$ in a point of $\mathbb{F}_{\mathbf{s}}(E)$, where we have defined the sequence \mathbf{s} to be $(0, s_1 = \rho_{\ell}, s_2 = \rho_l, r)$.

The local holomorphic coordinates (z, ζ) on $\mathbb{F}_{\mathbf{s}}(E)$ centered at (x_0, \mathbf{f}_0) parameterize the flags $0 \subset V_{z,s_1} \subset V_{z,s_2} \subset E_z$. Thus, the local holomorphic frame (1.8) for $Q_{\mathbf{s}}$ is given by the sections

$$\tilde{\epsilon}_{\alpha}(z,\zeta) = \text{image of } \epsilon_{\alpha}(z,\zeta) \text{ in } V_{z,s_2}/V_{z,s_1}, \text{ for } r-s_2 < \alpha \leq r-s_1.$$

Let

$$0 \to U_{\mathbf{s},1} \stackrel{\iota}{\hookrightarrow} U_{\mathbf{s},2} \stackrel{p}{\twoheadrightarrow} Q_{\mathbf{s}} \to 0$$

be a tautological exact sequence over $\mathbb{F}_{\mathbf{s}}(E)$, and let $p^* \colon Q_{\mathbf{s}} \to U_{\mathbf{s},2}$ be the C^{∞} orthogonal splitting of p. As before,

$$p^{\star} \cdot \tilde{\epsilon}_{\alpha}(z,\zeta) = \epsilon_{\alpha}(z,\zeta) + \sum_{r-s_1 < \mu \le r} u_{\alpha\mu}(z,\zeta) \epsilon_{\mu}(z,\zeta),$$

for some smooth functions $u_{\alpha\mu}$. Set $y := (z, \zeta)$. For $r - s_1 < \beta \leq r$, by using the computations made in (2.11), we have

$$\begin{aligned} 0 &= \langle \tilde{\epsilon}_{\alpha}, p \cdot \epsilon_{\beta} \rangle \\ &= \langle p^{\star} \cdot \tilde{\epsilon}_{\alpha}, \epsilon_{\beta} \rangle \\ &= \left\langle \epsilon_{\alpha} + \sum_{r-s_{1} < \mu} u_{\alpha\mu} \epsilon_{\mu}, \epsilon_{\beta} \right\rangle \\ &= \left\langle \epsilon_{\alpha}, \epsilon_{\beta} \right\rangle + \sum_{r-s_{1} < \mu} u_{\alpha\mu} \left\langle \epsilon_{\mu}, \epsilon_{\beta} \right\rangle \\ &= \delta_{\alpha\beta} - \sum_{j,k} c_{jk\alpha\beta} z_{j} \bar{z}_{k} + \bar{\zeta}_{\alpha\beta} + \sum_{\lambda \leq r-s_{2}} \zeta_{\lambda\alpha} \bar{\zeta}_{\lambda\beta} \\ &+ u_{\alpha\beta} + \sum_{r-s_{1} < \mu} u_{\alpha\mu} \left(-\sum_{j,k} c_{jk\mu\beta} z_{j} \bar{z}_{k} + \sum_{\nu \leq r-s_{1}} \zeta_{\nu\mu} \bar{\zeta}_{\nu\beta} \right) + \mathcal{O}(|y|^{3}). \end{aligned}$$

Note that, unlike what happens in (2.11), the coordinate $\zeta_{\alpha\beta}$ is now well-defined. Indeed, the inequalities $\alpha \leq r - s_1 < \beta$ are compatible with Condition (*). However, the coordinate $\zeta_{\mu\beta}$ coming from the product $\langle \epsilon_{\mu}, \epsilon_{\beta} \rangle$ is zero. This because both μ and β are strictly greater than $r - s_1$. Thus, it can not happen that $\mu \leq r - s_2 < \beta$, consequently Condition (*) fails for $\zeta_{\mu\beta}$.

Therefore, $u_{\alpha\beta} = -\bar{\zeta}_{\alpha\beta} + O(|y|^2)$ and

$$p^{\star} \tilde{\epsilon}_{\alpha} = \epsilon_{\alpha} - \sum_{r-s_1 < \mu} \left(\bar{\zeta}_{\alpha\mu} + \mathcal{O}(|y|^2) \right) \epsilon_{\mu}.$$

Denoting by $b \in \mathcal{A}^{1,0}(\mathbb{F}_{\mathbf{s}}(E), \operatorname{Hom}(U_{\mathbf{s},1}, Q_{\mathbf{s}}))$ the second fundamental form of $U_{\mathbf{s},1}$ in $U_{\mathbf{s},2}$ and by $b^* \in \mathcal{A}^{0,1}(\mathbb{F}_{\mathbf{s}}(E), \operatorname{Hom}(Q_{\mathbf{s}}, U_{\mathbf{s},1}))$ its adjoint, then

$$-\iota \circ b^{\star} \cdot \tilde{\epsilon}_{\alpha}|_{(x_0, \mathbf{f}_0)} = \bar{\partial} p^{\star} \tilde{\epsilon}_{\alpha}|_{(0, 0)} = -\sum_{r-s_1 < \mu} d\bar{\zeta}_{\alpha \mu} \otimes \epsilon_{\mu}.$$

Therefore, the curvature $\Theta(Q_{\mathbf{s}}, H_{\mathbf{s}})_{(x_0, \mathbf{f}_0)}$ with respect to the coordinates y is

$$\left(\Theta(U_{\mathbf{s},2},h_{\mathbf{s},2}) \big|_{Q_{\mathbf{s}}} + b \wedge b^{\star} \right)_{(x_0,\mathbf{f}_0)}$$

= $\sum_{\alpha,\beta} \left(\Theta_{\beta\alpha} - \sum_{1 \le \lambda \le r - s_2} d\zeta_{\lambda\alpha} \wedge d\bar{\zeta}_{\lambda\beta} + \sum_{r-s_1 < \mu \le r} d\zeta_{\beta\mu} \wedge d\bar{\zeta}_{\alpha\mu} \right) \otimes \tilde{\epsilon}_{\alpha}^{\vee} \otimes \tilde{\epsilon}_{\beta}$

where the equality follows from the expressions

$$b^{\star}_{(x_0,\mathbf{f}_0)} = \sum_{\substack{r-s_2 < \alpha \le r-s_1\\r-s_1 < \mu}} d\bar{\zeta}_{\alpha\mu} \otimes \tilde{\epsilon}^{\vee}_{\alpha} \otimes \epsilon_{\mu}, \quad b_{(x_0,\mathbf{f}_0)} = \sum_{\substack{r-s_2 < \beta \le r-s_1\\r-s_1 < \lambda}} d\zeta_{\beta\lambda} \otimes \epsilon^{\vee}_{\lambda} \otimes \tilde{\epsilon}_{\beta}$$

and from the previous part (i).

Parts (ii) and (iii) of the proof above follow the computations for the curvature of the tautological and universal quotient bundles of the Grassmann manifolds given in [Dem12, § 16.C].

Recall that, for $\rho = (0, r - 1, r)$, [Mou04, Formula (2.1)] provides the curvature in a point of the universal quotient $T = \pi_{\rho}^* E^{\vee} / \mathcal{O}_{\mathbb{P}(E^{\vee})}(-1)$ of the projectivized bundle $\mathbb{P}(E^{\vee})$, computed with respect to the local frame given by the isomorphism

$$T \cong T_{\mathbb{P}(E^{\vee})/X} \otimes \mathcal{O}_{\mathbb{P}(E^{\vee})}(-1).$$

2.3.2 Curvature intrinsic expression: universal vector bundles

In the spirit of Proposition 2.6, we want to express intrinsically the curvature of the universal vector bundles. To do this we use the formulæ provided in our Theorem 2.12.

From now on, fix two indices ℓ and l such that $0 \leq \ell < l \leq m$. In order to simplify the notation, we denote by \mathcal{E} the universal vector bundle $U_{\rho,l}/U_{\rho,\ell}$ of $\mathbb{F}_{\rho}(E)$. Finally, let H be the Hermitian metric naturally induced by h on \mathcal{E} , as mentioned in Remark 1.4.

As pointed out in Section 1.2.3, we have a natural (smooth) orthogonal splitting of the tangent bundle $T_{\mathbb{F}_{\rho}(E)}$ as the direct sum of the relative tangent bundle $T_{\mathbb{F}_{\rho}(E)/X}$ plus a horizontal part $T_{\mathbb{F}_{\rho}(E)/X}^{\perp_h}$, which depends only on h. Therefore, we can write the curvature tensor $\Theta(\mathcal{E}, H)$ as the sum of a "vertical" tensor

$$\Theta^{\text{vert}}_{(\mathcal{E},H)} \in C^{\infty} \left(\mathbb{F}_{\rho}(E), \Lambda^{1,1} T^{\vee}_{\mathbb{F}_{\rho}(E)/X} \otimes \text{End}(\mathcal{E}) \right)$$

plus a "horizontal" tensor

$$\Theta_{(\mathcal{E},H)}^{\mathrm{hor}} \in C^{\infty}\left(\mathbb{F}_{\rho}(E), \Lambda^{1,1}(T_{\mathbb{F}_{\rho}(E)/X}^{\perp_{h}})^{\vee} \otimes \mathrm{End}(\mathcal{E})\right).$$

Our aim is to generalize what was done in Section 2.2.1 for the curvature of universal line bundles, giving an explicit expression of the tensor $\Theta_{(\mathcal{E},H)}^{\text{hor}}$.

In the notation of Section 1.2.3, we have that

$$\Theta_{(\mathcal{E},H)}^{\text{hor}} = \Theta(\mathcal{E},H) \circ (p_2 \otimes \bar{p}_2)$$
(2.12)

where the right hand side of Formula (2.12) is a slightly improper notation which means that we are composing the (1, 1)-form part of $\Theta(\mathcal{E}, H)$ with $p_2 \otimes \bar{p}_2$, where p_2 is the projection on $T_{\mathbb{F}_{\rho}(E)/X}^{\perp_h}$.

Fix a point $(x_0, \mathbf{f}_0) \in \mathbb{F}_{\rho}(E)$. With respect to the coordinates (z, ζ) introduced in Section 1.2.1, Theorem 2.12 and Formula (2.12) give the equality

$$\Theta_{(\mathcal{E},H)}^{\text{hor}}(x_0,\mathbf{f}_0) = \sum_{r-\rho_l < \alpha, \beta \le r-\rho_\ell} \Theta_{\beta\alpha} \otimes e_{\alpha}^{\vee}(x_0) \otimes e_{\beta}(x_0).$$
(2.13)

Observe that, whichever the local frame of \mathcal{E} , that is (1.6), (1.7) or (1.8), we can suppose that it coincides with $(e_{r-\rho_l+1}(x_0), \ldots, e_{r-\rho_\ell}(x_0))$ if evaluated in (x_0, \mathbf{f}_0) , since (e_1, \ldots, e_r) is a local normal frame at x_0 .

Now, we define a section

$$\theta_{(\ell,l)} \colon \mathbb{F}_{\rho}(E) \to \Lambda^{1,1}T^{\vee}_{\mathbb{F}_{\rho}(E)} \otimes \operatorname{End}(\mathcal{E})$$

as follows. For $x \in X$, let $\mathbf{f} \in \mathbb{F}_{\rho}(E_x)$ be given by a unitary basis (v_1, \ldots, v_r) of E_x . We set

$$\theta_{(\ell,l)}(x,\mathbf{f}) = \frac{i}{2\pi} \sum_{r-\rho_l < \lambda, \mu \le r-\rho_\ell} \langle \pi_\rho^* \Theta(E,h)_{(x,\mathbf{f})} \cdot v_\lambda, v_\mu \rangle_h \otimes v_\lambda^\vee \otimes v_\mu.$$
(2.14)

Lemma 2.13. The section $\theta_{(\ell,l)}$ is well defined, i.e. it does not depend upon the choice of a particular representative $\mathbf{v} = (v_1, \ldots, v_r)$ for \mathbf{f} .

Proof. Take a local normal frame (e_1, \ldots, e_r) of E at x such that $\mathbf{e} := (e_1(x), \ldots, e_r(x))$ and \mathbf{v} identify the same flag \mathbf{f} . As already observed in Lemma 2.4, we know that if the index λ is such that

$$s_{m-l} = r - \rho_l < \lambda \le r - \rho_\ell = s_{m-\ell}$$

then we can write

$$v_{\lambda} = \sum_{s_{m-l} < \alpha \le s_{m-\ell}} a_{\alpha\lambda} e_{\alpha}(x),$$

where the coefficients (a_{pq}) are the entries of the diagonal block matrix A as in (2.6).

Using the local normal frame \mathbf{e} (and omitting its dependence on x to simplify the notation), we get

$$\begin{split} &\frac{i}{2\pi}\sum_{\lambda,\mu}\langle \pi_{\rho}^{*}\Theta(E,h)_{(x,\mathbf{f})}\cdot v_{\lambda},v_{\mu}\rangle_{h}\otimes v_{\lambda}^{\vee}\otimes v_{\mu}\\ &=\frac{i}{2\pi}\sum_{\lambda,\mu}\left\langle \pi_{\rho}^{*}\Theta(E,h)_{(x,\mathbf{f})}\cdot \left(\sum_{\alpha}a_{\alpha\lambda}e_{\alpha}\right),\sum_{\beta}a_{\beta\mu}e_{\beta}\right\rangle_{h}\otimes \left(\sum_{\tilde{\alpha}}a_{\tilde{\alpha}\lambda}e_{\tilde{\alpha}}\right)^{\vee}\otimes\sum_{\tilde{\beta}}a_{\tilde{\beta}\mu}e_{\tilde{\beta}}\\ &=\frac{i}{2\pi}\sum_{\alpha,\tilde{\alpha}}\sum_{\lambda}a_{\alpha\lambda}\overline{a_{\tilde{\alpha}\lambda}}\sum_{\tilde{\beta},\beta}\sum_{\mu}a_{\tilde{\beta}\mu}\overline{a_{\beta\mu}}\left\langle \pi_{\rho}^{*}\Theta(E,h)_{(x,\mathbf{f})}\cdot e_{\alpha},e_{\beta}\right\rangle_{h}\otimes e_{\tilde{\alpha}}^{\vee}\otimes e_{\tilde{\beta}}\\ &=\frac{i}{2\pi}\sum_{\alpha,\beta}\left\langle \pi_{\rho}^{*}\Theta(E,h)_{(x,\mathbf{f})}\cdot e_{\alpha},e_{\beta}\right\rangle_{h}\otimes e_{\alpha}^{\vee}\otimes e_{\beta}, \end{split}$$

where the indices of the summations are such that

$$\lambda, \mu, \alpha, \beta, \tilde{\alpha}, \beta = s_{m-l} + 1, \dots, s_{m-\ell}$$

Therefore, the lemma follows.

Observe also that, by definition, the section $\theta_{(\ell,l)}$ is smooth, hence it belongs to the space $\mathcal{A}^{1,1}(\mathbb{F}_{\rho}(E), \operatorname{End}(\mathcal{E}))$.

Now we show that the section $\theta_{(\ell,l)}$ coincides in each point of the flag bundle with the *h*-horizontal part of the Chern curvature of (\mathcal{E}, H) .

Lemma 2.14. The equality

$$\theta_{(\ell,l)} = \frac{i}{2\pi} \Theta_{(\mathcal{E},H)}^{\mathrm{hor}}$$

holds.

Proof. At any given $(x, \mathbf{f}) \in \mathbb{F}_{\rho}(E)$, choose (e_1, \ldots, e_r) to be a local normal frame for E at x such that \mathbf{f} is given by $(e_1(x), \ldots, e_r(x))$, and consider the induced holomorphic coordinates around (x, \mathbf{f}) as in Section 1.2.1.

Since the evaluation of $\theta_{(\ell,l)}$ in (x, \mathbf{f}) does not depend on the choice of the unitary basis defining \mathbf{f} , we have the following chain of equalities

$$\begin{aligned} \theta_{(\ell,l)}(x,\mathbf{f}) &= \frac{i}{2\pi} \sum_{r-\rho_l < \lambda, \mu \le r-\rho_\ell} \left\langle \pi_{\rho}^* \Theta(E,h)_{(x,\mathbf{f})} \cdot e_{\lambda}(x), e_{\mu}(x) \right\rangle_h \otimes e_{\lambda}(x)^{\vee} \otimes e_{\mu}(x) \\ &= \frac{i}{2\pi} \sum_{r-\rho_l < \lambda, \mu \le r-\rho_\ell} \pi_{\rho}^* \Theta_{\mu\lambda}(x,\mathbf{f}) \otimes e_{\lambda}(x)^{\vee} \otimes e_{\mu}(x) \\ &= \frac{i}{2\pi} \sum_{r-\rho_l < \lambda, \mu \le r-\rho_\ell} \Theta_{\mu\lambda}(x) \otimes e_{\lambda}(x)^{\vee} \otimes e_{\mu}(x) \\ &= \frac{i}{2\pi} \Theta_{(\mathcal{E},H)}^{\text{hor}}(x,\mathbf{f}), \end{aligned}$$

where the last equality follows from Formula (2.13).

Similarly to what was done in Section 2.2.1 we relabel $\omega_{(\ell,l)} := \frac{i}{2\pi} \Theta_{(\mathcal{E},H)}^{\text{vert}}$. The section $\omega_{(\ell,l)}$ is smooth as it equals the difference $\frac{i}{2\pi} \Theta(\mathcal{E},H) - \theta_{(\ell,l)}$.

Therefore, by Lemma 2.13 and Lemma 2.14 we have proved the following.

Proposition 2.15. For $0 \le l < l \le m$, we have the equality

$$\frac{i}{2\pi}\Theta(\mathcal{E},H) = \theta_{(\ell,l)} + \omega_{(\ell,l)}.$$
(2.15)

Observe that Formula (2.15) generalizes Formula (2.7).

2.3.3 Gysin formulæ for the curvature of universal vector bundles

Set $N := \binom{m+1}{2}$, and let $(\mathcal{E}_1, H_1), \ldots, (\mathcal{E}_N, H_N)$ be an enumeration of all the universal vector bundles, of ranks r_1, \ldots, r_N respectively, over $\mathbb{F}_{\rho}(E)$. Let F be a complex weighted homogeneous polynomial of degree $2(d_{\rho} + k)$ in $r_1 + \cdots + r_N$ variables.

We state the following technical proposition which generalizes Proposition 2.7.

Proposition 2.16. Let $F(c_{\bullet}(\mathcal{E}_1, H_1), \ldots, c_{\bullet}(\mathcal{E}_N, H_N))$ be a complex homogeneous polynomial in the Chern forms of the universal vector bundles on $\mathbb{F}_{\rho}(E)$. Then the push-forward

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N))$$

is given by a universal (weighted) homogeneous polynomial formally evaluated in the Chern forms of (E, h).

Remark 2.17. In order to prove the proposition we follow the steps in the proof of Proposition 2.7, by using the results for the universal vector bundles obtained in Sections 2.3.1 and 2.3.2. However, unlike Proposition 2.7 where we give the explicit computations needed for the proof, here we provide general remarks on the key concepts that allow us to prove the theorem, since, in the end, the computations are similar and explicit formulæ are not strictly necessary.

Proof of Proposition 2.16. In the notation of Section 2.3.2, fix for a moment two indices $0 \leq \ell < l \leq m$ which identify a universal vector bundle \mathcal{E} over $\mathbb{F}_{\rho}(E)$, equipped with the Hermitian metric H induced by h.

By Formula (2.15) we know that the curvature of (\mathcal{E}, H) can be written as the sum of a horizontal tensor plus a vertical one (in the sense of Section 2.3.2). We denote them by θ and ω respectively, by ignoring the indices.

For $j = 1, ..., \text{rk } \mathcal{E}$, recall the well-known equality from linear algebra

$$c_{j}(\mathcal{E}, H) = \operatorname{tr}_{\operatorname{End}(\Lambda^{j}\mathcal{E})} \left(\bigwedge^{j} \frac{i}{2\pi} \Theta(\mathcal{E}, H) \right)$$
$$= \frac{1}{j!} \det \begin{pmatrix} \operatorname{tr}(\theta + \omega) & j - 1 & 0 & \cdots \\ \operatorname{tr}(\theta + \omega)^{2} & \operatorname{tr}(\theta + \omega) & j - 2 & \cdots \\ \vdots & \vdots & \ddots & \vdots \\ \operatorname{tr}(\theta + \omega)^{j-1} & \operatorname{tr}(\theta + \omega)^{j-2} & \cdots & 1 \\ \operatorname{tr}(\theta + \omega)^{j} & \operatorname{tr}(\theta + \omega)^{j-1} & \cdots & \operatorname{tr}(\theta + \omega) \end{pmatrix}$$

where we have used Formula (2.15).

Now, fix a point $x \in X$ and let **f** be any flag in the fiber $\mathbb{F}_{\rho}(E_x)$ given by a unitary basis (v_1, \ldots, v_r) of E_x . Fixed a local normal frame (e_1, \ldots, e_r) for Ecentered at $x \in X$, we write $v_{\lambda} = \sum_{\nu} v_{\lambda}^{\nu} e_{\nu}(x)$, where the coordinates v_{λ}^{ν} 's are exactly as described in Proposition 2.7. Moreover, we can write θ (resp. ω) in matrix form with respect to the frame $(e_1(x), \ldots, e_r(x))$ as (θ_{pq}) (resp. ω_{uv}), where $p, q, u, v = 1, \ldots, \operatorname{rk} \mathcal{E}$.

Therefore, it is clear that each entry $\operatorname{tr}(\theta + \omega)^j$ is a polynomial in θ_{pq} 's and ω_{tu} 's, whose coefficients depend only on the rank of \mathcal{E} and of the indices $0 \leq \ell < l \leq m$ defining the bundle \mathcal{E} . Consequently, by the chain of equalities above, $c_j(\mathcal{E}, H)$ is a polynomial of type $P^j(\theta_{pq}, \omega_{tu})$ whose coefficients are universal.

Now, Formula 2.14 gives us an explicit expression of θ for which, w.r.t. the frame $(e_1(x), \ldots, e_r(x))$, we get

$$\theta_{pq}(x, \mathbf{f}) = \frac{i}{2\pi} \sum_{\alpha, \beta=1}^{r} Q_{\alpha\beta}(v, \bar{v}) \Theta_{\beta\alpha}(x), \qquad (2.16)$$

where we have denoted by v the vector of coordinates v_{λ}^{ν} 's. Observe that the polynomials $Q_{\alpha\beta}$'s have universal coefficients by Formula 2.14.

In order to compute the push-forward, by linearity we can of course suppose that the given polynomial F is just a monomial. Thus, it is sufficient to show that the push-forward through π_{ρ} of

$$\bigwedge_{s=1}^{N} \bigwedge_{j_s=1}^{r_s} c_{j_s}(\mathcal{E}_s, H_s)^{\wedge \ell_{j_s}} = \bigwedge_{s=1}^{N} \bigwedge_{j_s=1}^{r_s} P_s^{j_s}(\theta_{pq}^s, \omega_{tu}^s)^{\wedge \ell_{j_s}}$$
(2.17)

is a polynomial on the entries $\Theta_{\beta\alpha}$'s of the matrix associated to the curvature $\Theta(E,h)_x$ with respect to the frame (e_1,\ldots,e_r) .

We have observed in Formula (2.16) that θ_{pq}^s has universal polynomials in v and \bar{v} as coefficients. Therefore, coupling this with the universality of the P^{j} 's, after performing all the wedge powers and products in the right hand side of Formula (2.17), we can rewrite the latter as a polynomial A in the $\Theta_{\beta\alpha}$'s and ω_{tu}^s 's. Moreover, since all the polynomials involved in the right hand side of Formula (2.17) are universal, and we are performing wedge powers and products of them, we deduce that the coefficients of A are universal polynomials in v and \bar{v} .

As already pointed out in Proposition 2.7, recall that the $\Theta_{\beta\alpha}$'s only depend on the point x, while the v_{λ}^{ν} 's can be seen as variables of integration. Moreover, only the ω_{tu}^{s} 's contain the vertical differentials which can be integrated along the fibers of the submersion π_{ρ} .

By applying $(\pi_{\rho})_*$ to Formula (2.17), we have expressed

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N))$$

as a polynomial in the $\Theta_{\beta\alpha}$'s, whose coefficients are integrals over $\mathbb{F}_{\rho}(E_x)$ of universal polynomials in the variables v_{λ}^{ν} 's, and the volume forms are products of the ω_{tu}^s 's. By the same reasons as in Proposition 2.7, these integrals can be computed over the flag manifold $\mathbb{F}_{\rho}(\mathbb{C}^r)$, by applying a unitary transformation from (E_x, h_x) onto \mathbb{C}^r equipped with the standard metric. This concludes the proof. \Box

Now we see how to explicitly compute the push-forward

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N)) \in \mathcal{A}^{k,k}(X).$$

At the cohomology level, we formally evaluate F in the Chern classes of $\mathcal{E}_1, \ldots, \mathcal{E}_N$. This gives a cohomology class $F(c_{\bullet}(\mathcal{E}_1), \ldots, c_{\bullet}(\mathcal{E}_N))$ in the group $H^{2(d_{\rho}+k)}(\mathbb{F}_{\rho}(E))$, which we write as a polynomial \tilde{F} in terms of the Chern roots ξ_1, \ldots, ξ_r of $\pi_{\rho}^* E^{\vee}$. Thus, we have an equality

$$F(c_{\bullet}(\mathcal{E}_1),\ldots,c_{\bullet}(\mathcal{E}_N)) = \tilde{F}(\xi_1,\ldots,\xi_r).$$

Darondeau–Pragacz formula (2.1) gives us a universal weighted homogeneous polynomial Ψ of degree 2k such that

$$(\pi_{\rho})_* \tilde{F}(\xi_1, \dots, \xi_r) = \Psi(s_1(E), \dots, s_n(E)).$$

By expressing the Segre classes in terms of the Chern classes, we get a polynomial Φ such that $\Phi(c_{\bullet}(E)) = \Psi(s_{\bullet}(E))$. We thus have the equation

$$(\pi_{\rho})_* F(c_{\bullet}(\mathcal{E}_1), \dots, c_{\bullet}(\mathcal{E}_N)) = \Phi(c_1(E), \dots, c_r(E)).$$

$$(2.18)$$

Observe that the equation above can be recovered also by using Formula 2.3.

The next result, which is the main theorem of this section, completely translates Formula (2.18) at the level of differential forms and generalizes Theorem 2.10.

Theorem 2.18. Let (E, h) be a rank r Hermitian holomorphic vector bundle over a complex manifold X of dimension n, and let F be a complex homogeneous polynomial of degree $d_{\rho} + k$ in $r_1 + \cdots + r_N$ variables. Then, we have the equality

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N)) = \Phi(c_1(E,h),\ldots,c_r(E,h)).$$

Proof. The difference

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N)) - \Phi(c_{\bullet}(E,h))$$

is of course an exact global (k, k)-form on X. By Proposition 2.16,

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N))$$

is a universal homogeneous polynomial of weighted degree 2k in the Chern forms of (E, h). Hence, the previous difference can be written as a complex weighted homogeneous polynomial G in the Chern forms of (E, h), whose coefficients depend only upon r, n and F.

From now on, the proof is exactly the same as the proof of Theorem 2.10. Indeed, to show that $G(c_{\bullet}(E,h))$ is identically zero, we just need that it is exact and that G is a polynomial whose coefficients are universal.

Explicit computations which use Theorem 2.18 will be given in Section 3.2.1.

Remark 2.19. Theorem 2.10 is a particular case of Theorem 2.18. Indeed, Theorem 2.10 considers only polynomials in the first Chern forms of the successive quotients

$$\pi_{\rho}^{*E} / U_{\rho,m-1}, \dots, U_{\rho,2} / U_{\rho,1}, U_{\rho,1}$$

while Theorem 2.18 allows to compute the push-forward of any polynomial in the Chern forms of all the possible universal vector bundles.

2.4 Alternative expression of the push-forward formulæ

In the notation of Proposition 2.2, we now see how to give an analogue of Formula 2.3 to the differential forms level. As always, suppose that $(E, h) \to X$ is a Hermitian holomorphic vector bundle of rank r.

Given a homogeneous polynomial F as in Theorem 2.18, we want to give an explicit expression of the push-forward

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1,H_1),\ldots,c_{\bullet}(\mathcal{E}_N,H_N))$$

in terms of generalized Schur forms of (E, h).

Clearly, at the cohomology level we have that

$$F(c_{\bullet}(\mathcal{E}_1),\ldots,c_{\bullet}(\mathcal{E}_N)) \in H^{2(d_{\rho}+k)}(\mathbb{F}_{\rho}(E)).$$

Therefore, there exists a polynomial in the virtual Chern roots of $\pi_{\rho}^* E^{\vee}$ called

$$\tilde{F}(\xi_1,\ldots,\xi_r) = \sum_{|\lambda|=d_\rho+k} b_\lambda \xi_1^{\lambda_1}\cdots\xi_r^{\lambda_r},$$

which has the appropriate symmetries for which

$$\ddot{F}(\xi_1,\ldots,\xi_r)=F(c_{\bullet}(\mathcal{E}_1),\ldots,c_{\bullet}(\mathcal{E}_N)),$$

i.e., the polynomial $\tilde{F}(\xi_1, \ldots, \xi_r)$ can be considered a cohomology class of $\mathbb{F}_{\rho}(E)$.

In particular, if m = r the Chern roots of $\pi^* E^{\vee}$ are not virtual, cf. with Remark 2.9.

Since the explicit expression of the push-forward in cohomology is obviously independent of the method we use to compute it, from Formula 2.3 and Theorem 2.18 it follows the identity

$$(\pi_{\rho})_*F(c_{\bullet}(\mathcal{E}_1, H_1), \dots, c_{\bullet}(\mathcal{E}_N, H_N)) = \sum_{|\lambda|=d_{\rho}+k} b_{\lambda} s_{(\lambda-\nu)} (E, h),$$
(2.19)

where ν is given as in Proposition 2.2 and the b_{λ} 's are the coefficients of \tilde{F} . Obviously, the right hand side of Formula (2.19) is the form $\Phi(c_{\bullet}(E, h))$ given by Theorem 2.18.

In the next result, by means of Formula (2.19) we prove that each Schur form can be obtained as a push-forward from some flag bundle.

Let σ be a partition in $\Lambda(k, r)$. By the Jacobi–Trudi identities we know that

$$S_{\sigma}(E,h) = (-1)^{|\sigma|} s_{\sigma'}(E,h)$$
(2.20)

where σ' is the conjugate partition of σ , cf. with Example 1.13.

Remark 2.20. Observe that we can canonically associate to the conjugate partition $\sigma' \in \Lambda(k,r) \subset \mathbb{N}^k$ an element $\tilde{\sigma} = (\tilde{\sigma}_1, \dots, \tilde{\sigma}_r)$ of \mathbb{N}^r as follows. If

- k < r, then we set $\tilde{\sigma} = (\sigma'_1, \dots, \sigma'_k, \sigma'_{k+1} = 0, \dots, \sigma'_r = 0);$
- k = r, then $\tilde{\sigma}$ is σ' ;
- k > r, then, since $r \ge \sigma_1 \ge \cdots \ge \sigma_k \ge 0$, by definition σ' must be of the form $(\sigma'_1, \ldots, \sigma'_r, 0, \ldots, 0)$. Therefore we set $\tilde{\sigma} = (\sigma'_1, \ldots, \sigma'_r)$.

In all of the three cases above, it follows from the Definition 1.12 that $s_{\tilde{\sigma}}(E,h) = s_{\sigma'}(E,h)$.

The next proposition shows that the Schur form $S_{\sigma}(E, h)$ can be obtained as a push-forward from the complete flag bundle associated to E. We follow the notation introduced in Remark 2.20.

Proposition 2.21. Let σ be a partition in $\Lambda(k,r)$ and let $\lambda \in \mathbb{Z}^r$ be the sequence whose *j*-th element is $\lambda_j = \tilde{\sigma}_{r-j+1} + j - 1$. Then

$$\pi_*[(-1)^{|\lambda|+|\sigma|}\Xi_1^{\wedge\tilde{\sigma}_r}\wedge\cdots\wedge\Xi_r^{\wedge(\tilde{\sigma}_1+r-1)}] = S_{\sigma}(E,h)$$
(2.21)

where π is the natural projection of the complete flag bundle $\mathbb{F}(E)$, and the Ξ_j 's are the first Chern forms of its tautological successive quotients Q_j 's.

Proof. The sequence λ is defined in such a way that $\tilde{\sigma} = (\lambda - \nu)^{\leftarrow}$, where $\nu = (0, 1, \ldots, r-1)$ is as in Formula (2.2). Therefore, by Formula (2.19) it follows that

$$\pi_*\left[(-\Xi_1)^{\lambda_1} \wedge \dots \wedge (-\Xi_r)^{\lambda_r}\right] = s_{(\lambda-\nu)} (E,h) = (-1)^{|\sigma|} S_{\sigma}(E,h)$$

where the last equality follows from Formula (2.20) and Remark 2.20.

Observe that Proposition 2.21 is an alternative expression of the Jacobi–Trudi identity for differential forms given in [Fin21, Theorem 3.18]. Therefore, these two results are both a special case of Theorem 2.18.

Chapter 3

Positivity of characteristic forms for positive vector bundles

In this chapter, by means of the universal Gysin formulæ obtained in Chapter 2, we show the (strong) positivity of several characteristic differential forms built from the Chern curvature of Griffiths semipositive vector bundles. This gives a partial (but in a sense stronger) confirmation to a conjecture proposed by Griffiths in the late sixties, which has raised interest in the past (see, for instance, [Gri69; FL83; DPS94]) as well as in recent years (see, for instance, [Gul12; Fin21; Li21]).

In Section 3.1 we describe the above mentioned Griffiths' conjecture ([Gri69]), which concerns the positivity of the characteristic forms coming from positive vector bundles. In Section 3.2 we prove Theorem C and in Section 3.3 we prove Theorem D. These two results provide characteristic forms whose positivity was not previously known. Finally, Section 3.4 collects several results on Griffiths' conjecture in other literature, and some concluding remarks and open questions which may serve for further developments of this topic.

3.1 About Griffiths' conjecture on positive polynomials

As we saw in the introduction, Griffiths conjectured (and proved partially) in [Gri69] that given any rank r Hermitian holomorphic positive vector bundle on a projective manifold, the polynomials belonging to $\Pi(r)$ whenever evaluated on its Chern classes have to return a positive number once integrated over any subvariety of the correct dimension. A full proof of this conjecture is given in [FL83] in the more general setting of ample vector bundles (see also [DPS94] for the even more general context of E nef and X compact Kähler).

Actually, in [FL83] a more universal problem is considered and settled, *i.e.* to characterize precisely the *numerically positive polynomials for ample vector bundles* of rank r. These are defined to be those weighted homogeneous polynomials say of degree 2n that whenever evaluated on the Chern classes of any rank r ample vector bundle \mathcal{V} over an irreducible projective variety of dimension n give a positive number. Once again, the characterization is that numerically positive polynomials

for ample vector bundles of rank r are exactly the non zero positive polynomials.

Remark 3.1. It is observed in [FL83, Remark (1)] that if a weighted homogeneous polynomial P of degree 2n is not positive, *i.e.* it does not belong to $\Pi(r)$, then there exists a *smooth* projective manifold of dimension n, and an ample vector bundle of rank r over it, such that when one evaluates this polynomial in its Chern classes and integrates over the manifold, one gets a negative number. Moreover, such vector bundle is constructed as a quotient of a direct sum of very ample line bundles.

The upshot is that if we want to show that a weighted homogeneous polynomial is positive it suffices to show that it is a numerically positive polynomial for ample vector bundles over *smooth* projective manifolds. This will be useful later during the proof of Theorem 3.6.

Now, let $E \to X$ be a holomorphic vector bundle endowed with a Griffiths (semi)positive Hermitian metric h.

It is then natural to ask whether in this Hermitian setting the Fulton–Lazarsfeld– Demailly–Peternell–Schneider theorem holds pointwise for Chern forms, and this is also a question raised by Griffiths in the same paper.

Recall that in Definition 1.15 we have given the three main notions of positivity for differential (k, k)-forms.

Question 1 ([Gri69]). Given a Griffiths (semi)positive Hermitian holomorphic vector bundle (E, h), is it true that the positive polynomials evaluated on the Chern forms of E give rise to positive forms?

Remark 3.2. Coming back to Remark 3.1, we see that given P a weighted homogeneous polynomial of degree 2n which is not positive, there exists a rank r holomorphic Hermitian vector bundle (E, h) over a smooth projective manifold X of dimension n whose Chern curvature is Griffiths (as well as dual Nakano) positive and such that the corresponding characteristic form obtained by computing P in the Chern forms of (E, h) is not a positive (volume) form.

This is because one can endow E with the quotient metric of a positively curved direct sum metric on the direct sum of the very ample line bundles in question. Such a metric, begin a quotient of a positively curved (in any sense) metric, is both Griffiths and dual Nakano positive (but not Nakano positive, in general). The corresponding volume form $P(c_{\bullet}(E, h))$ has negative total mass, and hence must be negative somewhere.

This means that, even in the pointwise Hermitianized case considered by Griffiths, the cone of positive polynomials is the largest possible for which one can hope such a result.

Griffiths, in *loc. cit.*, answered in the affirmative to this question in the special case of the second Chern form of a rank 2 Griffiths positive holomorphic vector bundle (for the first Chern form the answer is trivially yes). Remark that this question gives, under the stronger hypothesis of Griffiths positivity, a stronger answer than its cohomological version stated earlier, since —as observed— a positive polynomial in the Chern form is a special representative in cohomology of the corresponding positive polynomial in the Chern classes.

In the last recent years there have been several partial results towards a fully affirmative answer to Griffiths' question. First, [Gul12] (see also [Div16] for a more

direct proof of the main technical result needed, as well as [Mou04] for similar, and somehow more general, computations) proved that the answer is affirmative in the special case of signed Segre classes.

Then, [Li21] proved the full statement but under the stronger assumption of Bott–Chern nonnegativity for (E, h) (see Section 3.4.1). We refer to *ibid.* for the definition of this variant of Hermitian positivity, which has been observed to be indeed equivalent to dual Nakano semipositivity in [Fin21].

Other related interesting results about finding positive representatives (not necessarily coming from the given positively curved metric) of the Schur polynomials in the Chern classes are obtained in [Pin18; Xia22].

For an even more recent result in the case of (dual) Nakano positive vector bundles, see [Fin21].

3.2 A family of strongly positive characteristic forms

Here, we are concerned with the original Question 1, which is still very much open: one can construct indeed (local, say over a ball) examples of Hermitian holomorphic vector bundles which are Griffiths positive but not Nakano nor dual Nakano positive, see for instance [Fin21, Proposition 2.9]. Our Theorem 2.10 allows us indeed to confirm the strong positivity of quite a few new positive combinations of Schur polynomials in the Chern forms, as follows.

Let (E, h) be a Griffiths semipositive vector bundle of rank r over a complex manifold X of dimension n. Consider the flag bundle $\pi_{\rho} \colon \mathbb{F}_{\rho}(E) \to X$ and, for $\mathbf{a} \in \mathbb{N}^r$ satisfying Condition (1.9), let $Q_{\rho}^{\mathbf{a}} \to \mathbb{F}_{\rho}(E)$ be the line bundle introduced in Section 1.2.2.

The first observation is contained in the following.

Proposition 3.3 ([Dem88a, Lemma 3.7 (a), Formula (4.9)]). If

$$\mathbf{a} = (a_1, \dots, a_r) \in \mathbb{N}^r$$

is non increasing, then $Q_{\rho}^{\mathbf{a}} \to \mathbb{F}_{\rho}(E)$ endowed with the natural Hermitian metric induced by h is a semipositive line bundle.

In the notation of Section 1.2.2, this means precisely that the Chern curvature $\Xi_{\rho}^{\mathbf{a}} = c_1(Q_{\rho}^{\mathbf{a}}, h)$ is a closed positive (1, 1)-form.

Let us recall (see Remark 1.19) that strongly positive forms are positive, but the converse is not true in general. However, strongly positive (k, k)-forms and positive (k, k)-forms do coincide for k = 0, 1, n - 1, n. Thus, $\Xi_{\rho}^{\mathbf{a}}$ is also a strongly positive (1, 1)-form.

Since the wedge product of strongly positive forms is again strongly positive (see Definition 1.15), then all wedge powers of $\Xi_{\rho}^{\mathbf{a}}$ are again strongly positive.

Now, by Proposition 1.25 we have that the push-forward of a closed strongly positive form under a proper holomorphic submersion is again a closed strongly positive form. Thus, we obtain immediately the next proposition.

Proposition 3.4. If $(E,h) \to X$ is a Griffiths semipositive vector bundle, then the closed forms

 $(\pi_{\rho})_*(\Xi_{\rho}^{\mathbf{a}})^{\wedge(d_{\rho}+k)}, \quad a_1 \ge a_2 \ge \cdots \ge a_r \ge 0,$

where d_{ρ} is the relative dimension and k is a non negative integer, are closed strongly positive (k, k)-forms.

Remark 3.5. If it happens that the chain of inequalities in the above statement is not strictly decreasing where prescribed by Condition (1.12), then the push-forward is identically zero. This is because in this case the curvature $\Xi_{\rho}^{\mathbf{a}}$ has some vertical zero eigenvalue in each fiber, thanks to Formula (1.11). Therefore, the vertical top form against which we integrate to obtain the push-forward is identically zero being, modulo a factor, the determinant of the vertical part of the curvature. So, in what follows, we can consider without loss of generality only weights $\mathbf{a} \in \mathbb{N}^r$ satisfying Condition (1.9) and such that $a_{s_1} > a_{s_2} > \cdots > a_{s_m}$.

Recall that we have denoted by ξ_1, \ldots, ξ_r the Chern roots of $\pi_{\rho}^* E^{\vee}$. Now, we come to the main result of this section.

Theorem 3.6. Let (E, h) be a Griffiths semipositive Hermitian holomorphic vector bundle of rank r over a complex manifold X of dimension n.

For every $\mathbf{a} \in \mathbb{N}^r$ satisfying Conditions (1.9) and (1.12), and for $0 \leq k \leq n$, the differential form

$$(\pi_{\rho})_* (\Xi_{\rho}^{\mathbf{a}})^{\wedge (d_{\rho}+k)}$$

is a closed strongly positive (k, k)-form on X belonging to the positive convex cone $\Pi(E, h)$ spanned by the Schur forms of (E, h).

Moreover, given the polynomial

$$P(t_1,\ldots,t_r) = \left(-\sum_{j=1}^m \sum_{\lambda=s_{j-1}+1}^{s_j} a_{s_j} t_\lambda\right)^{d_\rho+k},$$

the explicit expression of the push-forward can be obtained either:

- by formally evaluating in the Segre forms of (E,h) the right hand side of Formula (2.1) with $\tilde{F} = P$;

This theorem covers in particular Guler's work [Gul12], which concerned pushforwards from the projectivized bundle.

Remark 3.7. Observe that it is in some sense more natural to obtain that these forms are strongly positive rather than merely positive. This is because, as said earlier, positive polynomials are stable under products and so do strongly positive forms, while a product of positive forms is not necessarily still positive.

Proof. By definition, since $\Xi_{\rho}^{\mathbf{a}} = c_1(Q_{\rho}^{\mathbf{a}}, h)$ we have that

$$(\Xi_{\rho}^{\mathbf{a}})^{\wedge (d_{\rho}+k)} = (a_{s_1}\Xi_{\rho,1} + \dots + a_{s_m}\Xi_{\rho,m})^{\wedge (d_{\rho}+k)},$$

and thanks to Theorem 2.10 and Formula (2.19) applied to

$$F(\Xi_{\rho,1},\ldots,\Xi_{\rho,m}) = (a_{s_1}\Xi_{\rho,1}+\cdots+a_{s_m}\Xi_{\rho,m})^{\wedge(d_\rho+k)}$$

we get, respectively, the two explicit expressions claimed at the end of the statement.

The closedness and strong positivity of $(\pi_{\rho})_*(\Xi_{\rho}^{\mathbf{a}})^{\wedge (d_{\rho}+k)}$ are the content of Proposition 3.4.

By Remark 2.8, there exists a unique homogeneous polynomial $\Psi_{\mathbf{a}}^k$ of weighted degree 2k such that

$$(\pi_{\rho})_*(\Xi_{\rho}^{\mathbf{a}})^{\wedge(d_{\rho}+k)} = \Psi_{\mathbf{a}}^k(s_1(E,h),\ldots,s_n(E,h)).$$

We now want to show that $(\pi_{\rho})_*(\Xi_{\rho}^{\mathbf{a}})^{\wedge(d_{\rho}+k)}$ can be written as a positive linear combination of Schur forms. To do this, let $\Phi_{\mathbf{a}}^k$ be the *unique* polynomial such that

$$\Psi_{\mathbf{a}}^k(s_1(E,h),\ldots,s_n(E,h)) = \Phi_{\mathbf{a}}^k(c_1(E,h),\ldots,c_r(E,h)).$$

Observe that obviously the polynomial $\Phi_{\mathbf{a}}^k$ does not depend on the particular vector bundle considered, nor on the particular given base manifold, as usual. What we want to show will then follow from the Fulton–Lazarsfeld theorem [FL83] if we can prove that $\Phi_{\mathbf{a}}^k$ is a numerically positive polynomial for ample vector bundles of rank r over smooth projective manifolds, see Remark 3.1.

So, take any rank r ample vector bundle \mathcal{V} over a k-dimensional projective manifold Z. By [Dem88b, Lemma 4.1], the corresponding line bundle $Q_{\rho}^{\mathbf{a}}$ over $\mathbb{F}_{\rho}(\mathcal{V})$ is ample, and it can be therefore endowed with a smooth Hermitian metric $h_{\mathcal{V},\mathbf{a}}$ whose Chern curvature $i\Theta(Q_{\rho}^{\mathbf{a}}, h_{\mathcal{V},\mathbf{a}})$ is *strictly* positive, *i.e.* a Kähler form. But then, $(\pi_{\rho})_*c_1(Q_{\rho}^{\mathbf{a}}, h_{\mathcal{V},\mathbf{a}})^{\wedge(d_{\rho}+k)}$, is a closed positive *nowhere zero* (k, k)-form representing the cohomology class $\Phi_{\mathbf{a}}^k(c_1(\mathcal{V}), \ldots, c_r(\mathcal{V}))$. In particular, being represented by a non zero positive (k, k)-form, we have that

$$\int_{Z} \Phi_{\mathbf{a}}^{k}(c_{1}(\mathcal{V}),\ldots,c_{r}(\mathcal{V})) > 0,$$

as desired.

As a byproduct of the proof above one immediately obtains the following statement for ample vector bundles in the same spirit of [Pin18; Xia22].

Theorem 3.8. Let $E \to X$ be an ample vector bundle of rank r over a projective manifold. For every $\mathbf{a} \in \mathbb{N}^r$ satisfying Conditions (1.9) and (1.12), and for every $k = 0, \ldots, n = \dim X$, the (k, k)-cohomology classes $(\pi_{\rho})_* c_1(Q_{\rho}^{\mathbf{a}})^{d_{\rho}+k}$ contain a closed strongly positive form and belong to $\Pi(E)$.

We obtain thus a partial affirmative answer to Griffiths' question for the polynomials in the Chern forms of (E, h) belonging to the positive convex sub-cone $\mathcal{F}(E, h) \subset \Pi(E, h)$ spanned by all possible wedge products of all possible pushforwards $(\pi_{\rho})_* c_1(Q_{\rho}^{\mathbf{a}}, h)^{\wedge (d_{\rho}+k)}$, for $k = 0, \ldots, n$, as the weights $\mathbf{a} \in \mathbb{N}^r$ vary in the appropriate range prescribed by Conditions (1.9) and (1.12).

This sub-cone contains in particular the signed Segre forms, which arise in the case of projectivized bundle.

Remark 3.9. Even if it is possible to obtain every Schur form as a push-forward, see Remark 2.11 and Proposition 2.21, this is not enough to get here that every Schur form of a Griffiths semipositive vector bundle is positive. This is because the

curvature computations for the tautological bundles over the compete flag bundle do not permit to conclude that the relevant monomials whose push-forward give the desired Schur forms are positive.

The next examples are intended to give some flavor of which kind of new positive forms, in particular besides signed Segre forms, we are able to obtain with our methods.

3.2.1 Examples

In this section we give several examples of differential forms whose (strong) positivity is due to Theorem 3.6 and was not previously known in general. The explicit forms of some of them are obtained by implementing Formula (2.1) in PARI/GP as illustrated in Appendix A.

As we have seen, for (E, h) a rank r Griffiths (semi)positive vector bundle, the forms $(\pi_{\rho})_*c_1(Q_{\rho}^{\mathbf{a}}, h)^{\wedge(d_{\rho}+k)}$ are strongly positive. We want to highlight here some among them that cannot be shown to be positive only using results in the literature preceding the present work, at the best of our knowledge.

To this aim, observe that we already knew that the $(-1)^k s_k(E,h)$'s, *i.e.*, the signed Segre forms, are (strongly) positive for Griffiths (semi)positive vector bundles thanks to [Gul12, Theorem 1.1] (even though the strong positivity was not explicitly observed there). Also, as already noted, the product of positive forms, all of them strongly positive (resp. all except possibly one) is strongly positive (resp. positive).

This understood, in order to check for which forms we get new information about their positivity, we shall express $(\pi_{\rho})_* c_1(Q_{\rho}^{\mathbf{a}}, h)^{\wedge (d_{\rho}+k)}$ as a polynomial in the (signed) Segre forms of (E, h).

In what follows, in order to simplify the notation, we denote by c_1, \ldots, c_r the Chern forms of (E, h) and by s_1, \ldots, s_n the Segre forms of (E, h). The symbol S_{σ} stands for the Schur form of (E, h) associated to the partition σ . Moreover, we omit the symbol \wedge for the wedge product of forms.

Push-forwards from Grassmannian bundles

Denote by ρ the sequence (0, r - d, r). Then, $\mathbb{F}_{\rho}(E)$ is the Grassmannian bundle $\mathbb{G}_{r-d}(E)$ of (r-d)-planes in E. Let $\pi : \mathbb{G}_{r-d}(E) \to X$ be the projection, and denote by Q the universal quotient bundle of rank d on $\mathbb{G}_{r-d}(E)$ equipped with the quotient metric. In our notation, the class $c_1(Q)$ equals $c_1(Q_{\rho,1})$.

Therefore, for $N \ge d(r-d)$ the metric counterpart of the Darondeau–Pragacz push-forward formula reads

$$\pi_* c_1(Q,h)^N = \sum_{|\lambda|=N-d(r-d)} f^{\lambda+\varepsilon} \det\left((-1)^{\lambda_i+j-i} s_{\lambda_i+j-i}\right)_{1 \le i,j \le d}$$
(3.1)

where $\lambda = (\lambda_1, \ldots, \lambda_d)$ is a partition and $|\lambda|$ is its total weight, ε stands for the *d*-uple $(r-d)^d = (r-d, \ldots, r-d)$ and $f^{\lambda+\varepsilon}$ is the number of standard Young tableaux with shape $\lambda + \varepsilon$ (we have used here the more explicit version computed in the particular case of Grassmannian bundles in [KT15, Theorem 0.1]). Now, as explained for instance in [KT15], we have that

$$f^{\lambda+\varepsilon} = \frac{N! \prod_{1 \le i < j \le d} (\lambda_i - \lambda_j - i + j)}{\prod_{1 \le i \le d} (r + \lambda_i - i)!}.$$

Note that when d = 1 the bundle $\mathbb{G}_{r-1}(E)$ can be identified with $\mathbb{P}(E^{\vee})$, consequently $Q \cong \mathcal{O}_{\mathbb{P}(E^{\vee})}(1)$, and Formula (3.1) becomes

$$\pi_* c_1(\mathcal{O}_{\mathbb{P}(E^{\vee})}(1), h)^N = (-1)^{N-r+1} s_{N-r+1},$$

which is the push-forward formula by [Mou04; Gul12; Div16] giving the positivity of signed Segre forms.

It is noteworthy (see Example 1.8) to observe that already for the projectivized bundle of lines $\mathbb{P}(E)$ corresponding to the partition (0, 1, r), if we push-forward powers of $c_1(Q)$, where $Q = \pi^* E / \mathcal{O}_{\mathbb{P}(E)}(-1)$, we are now able to get forms whose positivity was not previously known.

In rank 3 (if r = 2 we have that $\mathbb{P}(E) \cong \mathbb{P}(E^{\vee})$ and there is nothing more to add) we see for instance, by using Formula (3.1), that

$$\pi_* c_1(Q,h)^5 = 4c_1^3 - 3c_1c_2 - c_3 = s_3 - 5s_1s_2,$$

and the positivity of $s_3 - 5s_1s_2$ was not previously known given that s_3 is negative.

Analogously, for the same reasons, if r = 4, the positivity of

$$\pi_* c_1(Q,h)^6 = 10c_1^3 - 4c_1c_2 - c_3 = s_3 - 6s_1s_2 - 5s_1^3$$

was not previously known.

The simplest example of a Grassmannian bundle which is not a projectivized bundle is $\mathbb{G}_2(E)$ for E of rank 4. Also in this case we get something new, as follows. By Formula (3.1), the push-forward via π of $c_1(Q, h)^N$ is given by

$$\begin{cases} 2 & \text{for } N = 4, \\ 5c_1 & \text{for } N = 5, \\ 9c_1^2 - 4c_2 & \text{for } N = 6, \\ 14(c_1^3 - c_1c_2) & \text{for } N = 7, \\ 2(10c_1^4 - 16c_1^2c_2 - c_1c_3 + 3c_2^2 + 4c_4) & \text{for } N = 8. \end{cases}$$

When rewritten in terms of signed Segre forms, we obtain

$$\begin{cases} 2 & \text{for } N = 4, \\ 5(-s_1) & \text{for } N = 5, \\ 5s_1^2 + 4s_2 & \text{for } N = 6, \\ 14(-s_1)s_2 & \text{for } N = 7, \\ 2(7s_1s_3 + 7s_2^2 - 4s_4) & \text{for } N = 8, \end{cases}$$

so that the positivity of the last form could not be previously deduced, since $-s_4$ is negative.

Push-forwards from complete flag bundles: the case of rank 3

The general formulæ for the push-forwards of $c_1(Q^{(a,b,c)},h)^{3+k}$ in terms of Schur forms, up to degree 3, are:

$$\begin{split} k &= 0 & \rightsquigarrow & 3(a^2b - ab^2 - a^2c + ac^2 + b^2c - bc^2), \\ k &= 1 & \rightsquigarrow & 4(a^3b - ab^3 - a^3c + ac^3 + cb^3 - bc^3)S_{(1)}, \\ k &= 2 & \rightsquigarrow & 10(a^3b^2 - a^2b^3 - a^3c^2 + a^2c^3 + b^3c^2 - b^2c^3)S_{(2,0)} \\ & + 5(a^4b - ab^4 - a^4c + ac^4 + b^4c - bc^4)S_{(1,1)}, \\ k &= 3 & \rightsquigarrow & 60(a^3b^2c - a^2b^3c - a^3bc^2 + a^2bc^3 + ab^3c^2 - ab^2c^3)S_{(3,0,0)} \\ & + 15(a^4b^2 - a^2b^4 - a^4c^2 + a^2c^4 + b^4c^2 - b^2c^4)S_{(2,1,0)} \\ & + 6(a^5b - ab^5 - a^5c + ac^5 + b^5c - bc^5)S_{(1,1,1)}. \end{split}$$

Clearly, our method produces positive forms until k reaches n, but already from these first cases we see how the complexity rapidly increases. Thanks to Theorem 3.6 we can say that all of the above listed forms belong to the positive cone for every $a \ge b \ge c \ge 0$.

For a concrete example, if (a, b, c) = (3, 2, 0) we obtain in terms of Segre forms:

$$\pi_* c_1(Q^{(3,2,0)}, h)^6$$

= 2700 S_(2,1,0) + 2340 S_(1,1,1)
= 180(-15s_1s_2 + 2s_3),

and the positivity of this form was not already known since s_3 is negative.

Note that, for instance, if (a, b, c) = (2, 1, 0) the positivity of

$$\pi_* c_1(Q^{(2,1,0)}, h)^6$$

= 180 S_(2,1,0) + 180 S_(1,1,1)
= -180s_1s_2

was instead previously known since $-s_1s_2$ is the wedge product of strongly positive forms.

Push-forwards from complete flag bundles: the case of rank 4

Here, we prefer to emphasize the different behavior in two special cases instead of giving the general formulæ for $a \ge b \ge c \ge d \ge 0$.

For (a, b, c, d) = (3, 2, 1, 0), we obtain

$$\pi_* c_1(Q^{(3,2,1,0)}, h)^9 = 90720(-s_1^3 - 2s_1s_2),$$

$$\pi_* c_1(Q^{(3,2,1,0)}, h)^{10} = 5040(216s_1^2s_2 + 7s_1s_3 + 39s_2^2 - 4s_4)$$

Note that the positivity of the last form was not previously known since $-s_4$ is negative, while we already knew that $-s_1^3 - 2s_1s_2$ is positive.

Now set (a, b, c, d) = (4, 3, 2, 0). In this case, we get

$$\pi_* c_1(Q^{(4,3,2,0)}, h)^9 = 181440(-8s_1^3 - 12s_1s_2 + s_3),$$

$$\pi_* c_1(Q^{(4,3,2,0)}, h)^{10} = 40320(648s_1^2s_2 - 124s_1s_3 + 42s_2^2 + 13s_4).$$

The positivity of both of these forms was not already known again because s_3 is negative.

3.3 Other positive characteristic forms

In this section we present a variant of the approach in Section 1.4.1, in order to obtain positivity of some Schur forms which, in general, are not in the cone $\mathcal{F}(E, h)$.

More specifically, we show the positivity of the Schur form c_2 in every rank and $c_1c_2 - c_3$ in rank 3 for Griffiths (semi)positive Hermitian holomorphic vector bundles.

In what follows, we proceed to observe that the positivity of the second Chern form is implied by the Griffiths *semi* positivity of the curvature tensor. Such argument is central in the proof of the positivity of $c_1c_2 - c_3$.

Let (E, h) be a Griffiths positive Hermitian holomorphic vector bundle of rank 2 over a complex manifold X. Thanks to [Gri69, Appendix to §5.(b)], we know that the second Chern form $c_2(E, h)$ is positive. The argument used by Griffiths in the proof consists in the following. First, by Proposition 1.23, one can assume that X is a complex surface. Of course, this assumption is not allowed if one wants to show the Hermitian (or strong) positivity of $c_2(E, h)$ (cf. with Remarks 1.19 and 1.22). However, in rank 2 the curvature is a 2 × 2 matrix of (1, 1)-forms, thus

$$c_2(E,h) = -\frac{1}{4\pi^2} \det \Theta(E,h).$$

Given that the Hermitian forms associated to the diagonal entries of the matrix $i\Theta(E, h)$ are positive definite, one can perform a simultaneous diagonalization of such Hermitian forms. After the diagonalization, using that dim X = 2, to show that $-\det \Theta(E, h)$ is positive it is sufficient to apply the Schwarz inequality coupled with the definition of Griffiths positivity.

Now, suppose that (E, h) is Griffiths semipositive. Once fixed a strictly positive (1, 1)-form ω on X and $\varepsilon > 0$, we can apply the above argument to the Herm(E, h)-valued (1, 1)-form

$$i\Theta(E,h) + \varepsilon\omega \otimes \mathrm{Id}_E$$

Passing to the limit for $\varepsilon \to 0$ we get the following.

Lemma 3.10. Given $(E,h) \to X$ Griffiths semipositive of rank 2, the differential form $c_2(E,h)$ is positive.

3.3.1 Positivity of c_2 in any rank

Let $(E,h) \to X$ be a Hermitian holomorphic vector bundle of rank r, and denote by $\Theta_{\alpha\beta}$ the entries of the curvature matrix. In general, we have that

$$c_{2}(E,h) = \operatorname{tr}_{\operatorname{End}(\Lambda^{2}E)}\left(\bigwedge^{2} \frac{i}{2\pi} \Theta(E,h)\right)$$

= $-\frac{1}{4\pi^{2}} \sum_{1 \leq \alpha < \beta \leq r} \left(\Theta_{\alpha\alpha} \wedge \Theta_{\beta\beta} - \Theta_{\alpha\beta} \wedge \Theta_{\beta\alpha}\right).$ (3.2)

If (E, h) is Griffiths positive, every diagonal entry $\Theta_{\alpha\alpha}$ of $\Theta(E, h)$ gives a positive definite Hermitian form. Therefore, it is possible to apply [Gri69, Appendix to §5.(b)] to all the summands in the last member of Formula (3.2). This because each summand is by definition the determinant of a 2 × 2 principal sub-matrix of the curvature matrix. From this follows the positivity of $c_2(E, h)$ in any rank.

In the semipositive case, the positivity of $c_2(E, h)$ in any rank is obtained by applying Lemma 3.10 (instead of the above argument) to all the summands in the last member of Formula (3.2). This proves the following.

Theorem 3.11. Let (E, h) be a Griffiths semipositive Hermitian holomorphic vector bundle over a complex manifold. Then the second Chern form

$$S_{(2,0)}(E,h) = c_2(E,h)$$

is a positive (2,2)-form.

3.3.2 Positivity of $c_1c_2 - c_3$ in rank 3

From now on, suppose that r = 3. Let $p := \pi_{(0,1,3)} \colon \mathbb{P}(E) \to X$ be the projective bundle of lines in E with associated tautological short exact sequence

$$0 \to \mathcal{O}_{\mathbb{P}(E)}(-1) \hookrightarrow p^*E \twoheadrightarrow Q := p^*E/\mathcal{O}_{\mathbb{P}(E)}(-1) \to 0$$

over $\mathbb{P}(E)$. Note that the quotient bundle $Q \to \mathbb{P}(E)$ is Griffiths semipositive of rank 2 with respect to the natural quotient metric induced by (E, h) (and denoted again by h). Consequently, the (1, 1)-form $c_1(Q, h)$ is strongly positive and, by Lemma 3.10, $c_2(Q, h)$ is positive.

We are now ready to prove the following result.

Theorem 3.12. Let (E, h) be a Griffiths semipositive Hermitian holomorphic vector bundle of rank 3 over a complex manifold. Then the Schur form

$$S_{(2,1,0)}(E,h) = c_1(E,h) \wedge c_2(E,h) - c_3(E,h)$$

is a positive (3,3)-form.

Remark 3.13. The main idea behind the next proof was developed for the first time by the author in an alternative proof of the positivity of $c_2(E, h)$ in rank 3. Since the positivity of $c_2(E, h)$ for Griffiths semipositive vector bundles is the content of Theorem 3.11, we omit such an alternative proof.

Proof of Theorem 3.12. Let $\pi \colon \mathbb{F}(E) \to X$ be the complete flag bundle associated to $E \to X$.

Following the notation of Section 2.4, let ξ_1, ξ_2, ξ_3 be the Chern roots of $\pi^* E^{\vee}$. They are given by the tautological filtration (1.1) (where r = m = 3), for which $\pi^* E$ splits (non canonically) as a differentiable vector bundle over $\mathbb{F}(E)$ as $Q_1 \oplus Q_2 \oplus Q_3$.

By the commutativity of the following diagram of projections

$$\mathbb{F}(E) \xrightarrow{q := \pi_{(0,1,3)}^{(0,1,2,3)}} \mathbb{P}(E)$$

$$\pi \xrightarrow{q} p$$

$$(3.3)$$

(cf. with the commutative diagram (1.5) above) we see that ξ_1 and ξ_2 are also the Chern roots of q^*Q^{\vee} , where $Q \to \mathbb{P}(E)$ is the rank 2 tautological quotient bundle introduced before. Indeed, since $\mathbb{F}(E)$ coincides with $\mathbb{F}(Q) = \mathbb{P}(Q)$,

$$q^*Q = q^*\left(p^*E/\mathcal{O}_{\mathbb{P}(E)}(-1)\right) = \pi^*E/U_1$$

and the associated graded bundle of the filtration $(0) \subset U_2/U_1 \subset q^*Q$ is

$$U_2/U_1 \oplus q^*Q/(U_2/U_1) = U_2/U_1 \oplus \pi^*E/U_2 = Q_2 \oplus Q_1.$$

By Formula (2.19) applied to $\pi: \mathbb{F}(E) \to X$ and where $F(\Xi_1, \Xi_2, \Xi_3)$ is taken to be the monomial $\Xi_1^{\wedge 4} \wedge \Xi_2^{\wedge 2} \wedge \Xi_3^{\wedge 0}$, we have the equality

$$\pi_*[\Xi_1^{\wedge 4} \wedge \Xi_2^{\wedge 2} \wedge \Xi_3^{\wedge 0}] = s_{(-2,1,4)}(E,h) = S_{(2,1,0)}(E,h).$$
(3.4)

Observe however that, a priori, we do not conclude anything about the positivity of $S_{(2,1,0)}(E,h)$ using only Equation (3.4), since, for instance, the line bundle Q_2 is not positive in general. Therefore, we apply again Formula (2.19), but to the flag bundle $q: \mathbb{P}(Q) \to \mathbb{P}(E)$, getting

$$q_*[\Xi_1^{\wedge 4} \wedge \Xi_2^{\wedge 2}] = s_{(1,4)}(Q,h) = c_1(Q,h) \wedge c_2(Q,h)^{\wedge 2}, \tag{3.5}$$

where the last equality holds given that $c_3(Q,h) \equiv 0$ and $c_4(Q,h) \equiv 0$.

Hence, the chain of equalities

$$p_* \left[c_1(Q,h) \wedge c_2(Q,h)^{\wedge 2} \right] = p_* q_* [\Xi_1^{\wedge 4} \wedge \Xi_2^{\wedge 2}]$$

= $\pi_* [\Xi_1^{\wedge 4} \wedge \Xi_2^{\wedge 2} \wedge \Xi_3^{\wedge 0}]$ by commutativity of (3.3)
= $S_{(2,1,0)}(E,h)$ by Formula (3.4)

follows, by applying p_* to both members of Formula (3.5).

Since $Q \to \mathbb{P}(E)$ is Griffiths semipositive, by Lemma 3.10 we know that $c_2(Q, h)$ is a positive form on $\mathbb{P}(E)$. Thanks to [BP13, Theorem 1], we have that the square of a positive (2, 2)-form is also positive. Hence $c_1(Q, h) \wedge c_2(Q, h)^{\wedge 2}$ is positive, given that it is the wedge product of a strongly positive and of a positive form. Consequently, $S_{(2,1,0)}(E, h)$ must be a positive form on X, since it is the push-forward of a positive differential form.

Remark 3.14. For dimension and bi-degree reasons (see Remark 1.19) we underline that if dim $X \leq 3$, then $c_2(E, h)$ is strongly positive, and if dim $X \leq 4$, then $S_{(2,1,0)}(E, h)$ is strongly positive, too.

Remark 3.15. It is interesting to observe that Theorem 3.12 can be deduced also by Theorem 2.18. More precisely, the general push-forward formula obtained in Theorem 2.18 gives us the equality

$$p_*\left[c_1(Q,h) \wedge c_2(Q,h)^{\wedge 2}\right] = S_{(2,1,0)}(E,h)$$

without the need to use the map π_* of the complete flag bundle. However, the previous strategy of proof has provided us the right (positive) differential form on $\mathbb{P}(E)$ to be pushed-forward to obtain the positivity of $S_{(2,1,0)}(E,h)$.

From the (strong) positivity of the second Segre form

$$s_2(E,h) = c_1(E,h)^{\wedge 2} - c_2(E,h)$$

(see [Gul12, Theorem 1.1] and Theorem 3.6), and from Theorem 3.12 we deduce the following.

Corollary 3.16. If (E, h) is Griffiths semipositive of rank 3 over a complex manifold, then the following pointwise inequalities hold

$$c_1(E,h)^{\wedge 3} \ge c_1(E,h) \wedge c_2(E,h) \ge c_3(E,h).$$
 (3.6)

In addition, if X is compact and 3-dimensional, the following chain of inequalities between Chern numbers of E also holds

$$\int_{X} c_1(E)^3 \ge \int_{X} c_1(E) c_2(E) \ge \int_{X} c_3(E).$$
(3.7)

As already mentioned in the introduction, the inequalities in (3.6) can be deduced by [Li21, Theorem 3.2] only if h is dual Nakano semipositive. Thus, Corollary 3.16 is in some sense a generalization of [Li21, Theorem 3.2] in rank 3.

Remark 3.17. It is well-known that a vector bundle admitting a Griffiths semipositive metric is nef (see [DPS94] for a definition). Recall that the Chern numbers of a nef vector bundle on a compact *n*-dimensional Kähler manifold are bounded above by the Chern number c_1^n ([DPS94, Corollary 2.6]) and below by the Euler number c_n ([LZ20, Theorem 2.9]). See also [Li21, Remark 3.3]. Therefore, assuming that Xis also Kähler the first and the second inequalities in (3.7) are a particular case of [DPS94, Corollary 2.6] and [LZ20, Theorem 2.9] respectively. Moreover, observe that the inequalities in (3.7) follow from [DPS94, Theorem 2.5] if X is Kähler.

Remark 3.18. It seems difficult to apply the strategy in the proof of Theorem 3.12 to prove the positivity of $c_3(E, h)$ in rank 3. Indeed, in order to factor through the quotient bundle, we need to push-forward only monomials of the form $\Xi_1^{\wedge\lambda_1} \wedge \Xi_2^{\wedge\lambda_2}$. But, by Formula (2.19) we have the equality $\pi_*[\Xi_1^{\wedge3} \wedge \Xi_2^{\wedge2} \wedge \Xi_3] = c_3(E, h)$ that involves Ξ_3 , which is the pull-back through q of the Chern curvature of $\mathcal{O}_{\mathbb{P}(E)}(-1)$.

However, one can try to adapt the same ideas presented in Theorem 3.12 to vector bundles of rank higher than 3. For example, in rank 4 we could use the tautological quotient Q of the Grassmann bundle $\mathbb{G}_2(E)$ in order to get push-forward formulæ for some Schur forms. The problem here is that the monomials we push-forward contain $c_2(Q, h)^{\wedge \ell}$, $\ell > 2$. Therefore, as pointed out for instance in [BP13], we do not know a priori if these powers are positive.

3.4 State of the art, remarks and open questions

In this concluding section we suppose, as always, that (E, h) is a rank r Hermitian holomorphic vector bundle over a complex manifold X.

3.4.1 A comparison on the positivity of Schur forms

We compare here the different notions of positivity (see Definition 1.15) of Schur forms that appear in [BC65; Gri69; Gul12; Li21; Fin21], which are the works mainly related to Theorem C and Theorem D.

Recall that we do not deal with the strict notions of positivity (see Remark 1.17). For this, the following exposition can be made slightly more precise, although this does not affect the purpose of the comparison.

Positivity in [BC65]

In [BC65, Definition 5.1] a notion of positivity for elements in $\mathcal{A}^{p,p}(X, \operatorname{End}(E))$ is given. In particular, for p = 1, such notion includes the so-called Bott–Chern nonnegativity (this terminology is due to [Li21], see below), which requires that the curvature of (E, h) can be expressed, locally, as $A \wedge \overline{A}^t$, where A is a matrix of (1,0)-forms of appropriate size. Although [BC65] predates Griffiths' conjecture, from [BC65, Lemma 5.3] one can deduce the Hermitian positivity of the top Chern form of a Bott–Chern nonnegative vector bundle.

Positivity in [Gri69]

The positivity notion considered in [Gri69, p. 240] is, by Proposition 1.21, Hermitian positivity. As already mentioned, Griffiths' conjecture first appears in [Gri69, p. 247], where it is conjectured that for a Griffiths positive vector bundle the cone of positive polynomials in the Chern forms (*i.e.*, the Schur cone) consists of Hermitian positive differential forms. The full conjecture is verified in [Gri69, p. 246] for globally generated vector bundles; for more details we refer to [Gri69, Proof of Theorem D].

Moreover, by the characterization of Proposition 1.23, Griffiths proves in [Gri69, Appendix to §5.(b)] the positivity, but not the Hermitian positivity, of $c_2(E, h)$, for (E, h) Griffiths positive of rank 2.

Positivity in [Gul12]

The main result of [Gul12] states that the signed Segre forms of a Griffiths positive vector bundle are positive. Actually, [Gul12, Theorem 1.1] implicitly proves that such forms are strongly positive, even though the strong positivity is not explicitly observed therein.

Positivity in [Li21]

The two positivity notions considered by [Li21] are, by Propositions 1.23 and 1.21, positivity and Hermitian positivity, although in [Li21, §2] they are called *nonnegativity* and *strong nonnegativity* respectively. In the same spirit of [Gri69, Proof of Theorem D], [Li21, Proposition 3.1] extends the above mentioned result on globally generated vector bundles to the larger family of Bott–Chern nonnegative (see [Li21, Definition 2.1]) vector bundles, showing that the Schur forms of these bundles are Hermitian positive.

Positivity in [Fin21]

All the three notions of Definition 1.15 are addressed in [Fin21], although with the terminology of [HK74]. By assuming Nakano, or dual Nakano, (semi)positivity [Fin21, Theorem 1.1] shows that Schur forms are Hermitian positive.

It is also worth to recall that [Fin21] relates, in a very interesting way, Bott-Chern notion of positivity with the dual Nakano one. More precisely, [Fin21, Theorem 2.14] states that a Hermitian vector bundle $(E, h) \to X$ is dual Nakano semipositive if and only if for every $x \in X$ there is a vector space V and a matrix of (1, 0)forms $A \in T_{X,x}^{\vee} \otimes \operatorname{Hom}(V, E_x)$ such that $\Theta(E, h)_x = A \wedge \overline{A}^t$. In order to have a similar characterization for dual Nakano positivity, we observe that it is sufficient to require, in addition, the invertibility of the operator $\widetilde{A} \in \operatorname{Hom}(T_{X,x} \otimes E_x^{\vee}, V^{\vee})$ naturally associated to A. Therefore, both [Li21] and [Fin21] obtain the full Griffiths' conjecture for dual Nakano (semi)positive bundles, but with different methods.

Finally, we recall that [Fin21, Theorem 1.3] establishes the equivalence between Griffiths' conjecture and an open question concerning *mixed discriminants*, see [Fin21, Open problem].

Positivity in this thesis

Given a Griffiths semipositive vector bundle $(E, h) \to X$, Theorem 3.6 shows the strong positivity of the family $\mathcal{F}(E, h)$ of differential forms in the Schur cone $\Pi(E, h)$. Such family form a sub-cone which, in the notation of Section 3.2, is spanned by all possible wedge products of all possible push-forwards

$$(\pi_{\rho})_* (a_1 \Xi_{\rho,1} + \dots + a_m \Xi_{\rho,m})^{\wedge (d_{\rho} + k)}$$
 (3.8)

as $k \geq 0$, $\rho = (\rho_0, \ldots, \rho_m)$ and the sequence of integers $a_1 \geq \cdots \geq a_m \geq 0$ vary. As already mentioned, observe that to $\rho = (0, r - 1, r)$ corresponds the bundle $\pi_{\rho} \colon \mathbb{P}(E^{\vee}) \to X$ of hyperplanes in E. Hence $\Xi_{\rho,1} = c_1(\mathcal{O}_{\mathbb{P}(E^{\vee})}(1), h)$ and for $(a_1, a_2) = (1, 0)$ Guler's result [Gul12, Theorem 1.1] on the signed Segre forms is recovered. Moreover, Theorems 3.11 and 3.12 show the positivity of $c_2(E, h)$ in any rank and $S_{(2,1,0)}(E, h)$ in rank 3 respectively.

Currently, the positivity of other (positive combinations of) Schur forms of Griffiths semipositive bundles is not known.

3.4.2 Concluding remarks

If we assume (dual) Nakano semipositivity, then the Schur cone consists of Hermitian positive differential forms. Hence, it is natural to ask the following question.

Question 2. Beside those as in Expression (3.8) (and wedge products of them), are there in the Schur cone other strongly positive differential forms for (dual) Nakano semipositive vector bundles?

Of course, the best possible result would be that all the Schur cone consists of strongly positive differential forms.

Assume now that the vector bundle is Griffiths semipositive. Although it is not explicitly mentioned in Griffiths' conjecture (as stated in the literature), by the results listed in Section 3.4.1 it seems interesting to ask what is the most natural notion of positivity, among those in Definition 1.15, that we can expect to hold for the Schur forms of the vector bundle. Certainly, the conjecture is still open even requiring the weakest notion of positivity, although, as already said, Theorem 3.6 shows us that the strong positivity naturally appears in this context. In addition, it is in a sense more natural to require the strong, or Hermitian, positivity of the Schur forms (see Remark 3.7). Indeed, as mentioned in Section 1.3, positive polynomials (*i.e.*, those belonging to the Schur cone) are stable under product and so do Hermitian and strongly positive forms; while the wedge product of two positive forms is not necessarily positive (see [Dem12, §III, (1.11) Proposition] and [BP13]).

Summing up, one may ask an analogue of Question 2 for Griffiths semipositive vector bundles, wondering if, outside the cone $\mathcal{F}(E, h)$ spanned by the wedge products of the push-forwards (3.8), there are other Hermitian or strongly positive Schur forms. If they were all (at least) Hermitian positive one would have an affirmative answer to the original Griffiths' conjecture as stated in [Gri69, p. 247].

However, this question is currently open. For instance, beside the cases listed in Remark 3.14, we do not deduce the Hermitian positivity of $c_2(E,h)$ (resp. of $S_{(2,1,0)}(E,h)$) by the proof of Theorem 3.11 (resp. of Theorem 3.12).

3.4.3 Further related questions

Let \mathcal{V} be an ample vector bundle over a projective manifold X.

As pointed out in [Pin18; Xia22], one can study some variants of Griffiths' conjecture. For instance, [Xia22, Conjecture 1.4] asks whether every Schur class $S_{\sigma}(\mathcal{V})$ does admit a positive representative. Such question fits between Fulton–Lazarsfeld theorem and Griffiths' conjecture. Thanks to [Xia22, Theorem A] we know that the answer to this question is affirmative if $|\sigma| = \dim X - 1$. Another partial affirmative answer to [Xia22, Conjecture 1.4] is given by our Theorem 3.8, where it is shown that the positive combinations of Schur classes given by pushforwards as in Expression (3.8) contain a strongly positive form. However, the general problem is open. Moreover, as observed in [Xia22, Remark 1.5], given a smooth representative $\eta \in S_{\sigma}(\mathcal{V})$ it is not clear how to find a Hermitian metric h on \mathcal{V} such that $\eta = S_{\sigma}(\mathcal{V}, h)$. By virtue of these remarks, one may wonder the following.

Question 3. Given an ample vector bundle $\mathcal{V} \to X$, is it true that for any partition $\sigma \in \Lambda(k, r)$, there exists a Hermitian metric h_{σ} on \mathcal{V} such that $S_{\sigma}(\mathcal{V}, h_{\sigma})$ is (Hermitian/strongly) positive?

If dim X = 2 and \mathcal{V} is semistable with respect to some polarization, [Pin18, Theorem 1.1] provides an affirmative answer to Question 3, finding a Hermitian metric on \mathcal{V} whose Schur forms are positive. However, it is not clear if such metric is Griffiths positive: see [Pin18, p. 633], and also [Fin21, Section 4.2] for related considerations. Clearly, an affirmative answer to Question 3 would imply [Xia22, Conjecture 1.4].

We would finally remark that affirmative answers to [Xia22, Conjecture 1.4] or Question 3 do not imply Griffiths' conjecture, given that the latter requires the positivity of the Schur forms with respect to one *given* Griffiths (semi)positive Hermitian metric on the vector bundle.
Appendix A PARI/GP computations

Here we report the codes implemented on PARI/GP CALCULATOR Version 2.11.4 which provide some of the explicit examples of strongly positive differential forms given in Section 3.2.1. These codes are implementations of the right hand side of Formula 2.1 in case of complete flag bundles.

Although appropriate simple modifications of the following codes provide all the differential forms obtainable from Theorem C in case of complete flag bundles, the computation complexity rapidly increases. For instance, our computers reached the general formulæ in the case of rank 4 in a long time.

Note that all the outputs of the following codes are in terms of Segre forms s_1, \ldots, s_n . To obtain outputs in terms of Schur forms we have implemented change of basis formulæ up to degree 4, which we give at the end of this section.

Push-forwards from complete flag bundles: r = 3, n = 3:

```
t=[t1,t2,t3]; /*formal variables*/
s=[s1,s2,s3]; /*Segre forms of (E,h) on X*/
q=prod(i=1,3, 1+sum(k=1,3, (1/((t[i])^k))*s[k]));
v=(t[1]-t[2])*(t[1]-t[3])*(t[2]-t[3]); /*Vandermonde polynomial*/
Y=q*v;
w=[a,b,c]; /*exponents of the universal line bundle*/
/*the following must be implemented on the same line*/
```

```
for(d=3,6,
C=((w[1]*(-t[1])+w[2]*(-t[2])+w[3]*(-t[3]))^d)*Y;
for(j=1,3, C=polcoeff(C,2,t[j]));
print(C))
```

Push-forwards from complete flag bundles: r = 4, n = 4:

```
t=[t1,t2,t3,t4]; /*formal variables*/
s=[s1,s2,s3,s4]; /*Segre forms of (E,h) on X*/
q=prod(i=1,4, 1+sum(k=1,4, (1/((t[i])^k))*s[k]));
v=prod(i=1,3, prod(j=i+1,4, t[i]-t[j])); /*Vandermonde polynomial*/
Y=q*v;
```

```
w=[a,b,c,d]; /*exponents of the universal line bundle*/
/*the following must be implemented on the same line*/
for(d=6,10,
C=((w[1]*(-t[1])+w[2]*(-t[2])+w[3]*(-t[3])+w[4]*(-t[4]))^d)*Y;
for(j=1,4, C=polcoeff(C,3,t[j]));
print(C))
```

Remark A.1. Of course, the implementations above work for any chosen dimension n simply by modifying the array **s** of the Segre forms and the polynomial **q** accordingly. Consequently, by modifying the upper limit of **d** (which is the relative dimension of the complete flag bundle) in the last line of the code we get n + 1 outputs.

Change of basis from Segre forms to Schur forms

Given C a weighted homogeneous polynomial of degree d in the Segre forms of (E, h), we want an array S containing all the coefficients of C in terms of the basis given by Schur forms. In order to do this, the code we have implemented is the following.

```
/*d=1*/
S=[-polcoeff(C,1,s1)]
/*d=2*/
S=[polcoeff(C,2,s1),polcoeff(C,2,s1)+polcoeff(C,1,s2)]
/*d=3*/
/*the following must be implemented on the same line*/
S=
Γ
-polcoeff(C,3,s1),
-2*polcoeff(C,3,s1)-polcoeff(polcoeff(C,1,s1),1,s2),
-polcoeff(C,3,s1)-polcoeff(polcoeff(C,1,s1),1,s2)-polcoeff(C,1,s3)
٦
/*d=4*/
/*the following must be implemented on the same line*/
S=
Γ
polcoeff(C,4,s1),
3*polcoeff(C,4,s1)+polcoeff(polcoeff(C,2,s1),1,s2),
2*polcoeff(C,4,s1)+polcoeff(polcoeff(C,2,s1),1,s2)+polcoeff(C,2,s2),
3*polcoeff(C,4,s1)+2*polcoeff(polcoeff(C,2,s1),1,s2)+
polcoeff(polcoeff(C,1,s1),1,s3)+polcoeff(C,2,s2),
polcoeff(C,4,s1)+polcoeff(polcoeff(C,2,s1),1,s2)+
polcoeff(polcoeff(C,1,s1),1,s3)+polcoeff(C,2,s2)+polcoeff(C,1,s4)
]
```

 $\mathbf{58}$

Bibliography

- [AC87] Ersan Akyildiz and James B. Carrell. "An algebraic formula for the Gysin homomorphism from G/B to G/P". In: Illinois J. Math. 31.2 (1987), pp. 312–320. ISSN: 0019-2082. URL: http://projecteuclid.org/euclid.ijm/1255989322.
- [BC65] Raoul Bott and S. S. Chern. "Hermitian vector bundles and the equidistribution of the zeroes of their holomorphic sections". In: Acta Math. 114 (1965), pp. 71–112. ISSN: 0001-5962. DOI: 10.1007/BF02391818. URL: https://doi.org/10.1007/BF02391818.
- [BG71] Spencer Bloch and David Gieseker. "The positivity of the Chern classes of an ample vector bundle". In: *Invent. Math.* 12 (1971), pp. 112–117. ISSN: 0020-9910. DOI: 10.1007/BF01404655. URL: https://doi.org/10.1007/BF01404655.
- [BP13] Zbigniew Błocki and Szymon Pliś. "Squares of positive (p, p)-forms". In: C. R. Math. Acad. Sci. Paris 351.1-2 (2013), pp. 27-32. ISSN: 1631-073X. DOI: 10.1016/j.crma.2013.01.009. URL: https://doi.org/10.1016/ j.crma.2013.01.009.
- [Bri96] Michel Brion. "The push-forward and Todd class of flag bundles". In: *Parameter spaces (Warsaw, 1994)*. Vol. 36. Banach Center Publ. Polish Acad. Sci. Inst. Math., Warsaw, 1996, pp. 45–50.
- [BS12] Gergely Bérczi and András Szenes. "Thom polynomials of Morin singularities". In: Ann. of Math. (2) 175.2 (2012), pp. 567–629. ISSN: 0003-486X. DOI: 10.4007/annals.2012.175.2.4. URL: https://doi.org/10. 4007/annals.2012.175.2.4.
- [BT82] Raoul Bott and Loring W. Tu. Differential forms in algebraic topology. Vol. 82. Graduate Texts in Mathematics. Springer-Verlag, New York-Berlin, 1982, pp. xiv+331. ISBN: 0-387-90613-4.
- [CF90] F. Campana and H. Flenner. "A characterization of ample vector bundles on a curve". In: *Math. Ann.* 287.4 (1990), pp. 571–575. ISSN: 0025-5831. DOI: 10.1007/BF01446914. URL: https://doi.org/10.1007/ BF01446914.
- [Dam73] James Damon. "The Gysin homomorphism for flag bundles". In: Amer. J. Math. 95 (1973), pp. 643–659. ISSN: 0002-9327. DOI: 10.2307/2373733.
 URL: https://doi.org/10.2307/2373733.

[Dem12]	Jean-Pierre Demailly. Complex Analytic and Differential Geometry. Version of 2012. URL: https://www-fourier.ujf-grenoble.fr/ ~demailly/manuscripts/agbook.pdf.
[Dem88a]	Jean-Pierre Demailly. "Vanishing theorems for tensor powers of a positive vector bundle". In: <i>Geometry and analysis on manifolds (Katata/Kyoto, 1987)</i> . Vol. 1339. Lecture Notes in Math. Springer, Berlin, 1988, pp. 86–105. DOI: 10.1007/BFb0083049. URL: https://doi.org/10.1007/BFb0083049.
[Dem88b]	Jean-Pierre Demailly. "Vanishing theorems for tensor powers of an ample vector bundle". In: <i>Invent. Math.</i> 91.1 (1988), pp. 203–220. ISSN: 0020-9910. DOI: 10.1007/BF01404918. URL: https://doi.org/10.1007/BF01404918.
[DF22]	Simone Diverio and Filippo Fagioli. "Pointwise universal Gysin formulæ and applications towards Griffiths' conjecture". In: Ann. Sc. Norm. Super. Pisa Cl. Sci. (2022). in press. DOI: 10.2422/2036-2145.202011_021. URL: https://doi.org/10.2422/2036-2145.202011_021.
[Div16]	Simone Diverio. "Segre forms and Kobayashi-Lübke inequality". In: <i>Math. Z.</i> 283.3-4 (2016), pp. 1033–1047. ISSN: 0025-5874. DOI: 10.1007/s00209-016-1632-y. URL: https://doi.org/10.1007/s00209-016-1632-y.
[DP17]	Lionel Darondeau and Piotr Pragacz. "Universal Gysin formulas for flag bundles". In: <i>Internat. J. Math.</i> 28.11 (2017), 1750077, 23 pp. ISSN: 0129-167X. DOI: 10.1142/S0129167X1750077X. URL: https://doi.org/10.1142/S0129167X1750077X.
[DPS94]	Jean-Pierre Demailly, Thomas Peternell, and Michael Schneider. "Com- pact complex manifolds with numerically effective tangent bundles". In: <i>J. Algebraic Geom.</i> 3.2 (1994), pp. 295–345. ISSN: 1056-3911.
[Fag22]	Filippo Fagioli. "A note on Griffiths' conjecture about the positivity of Chern-Weil forms". In: <i>Differ. Geom. Appl.</i> 81 (2022), Paper No. 101848. ISSN: 0926-2245. DOI: 10.1016/j.difgeo.2022.101848. URL: https://doi.org/10.1016/j.difgeo.2022.101848.
[Fin21]	Siarhei Finski. On characteristic forms of positive vector bundles, mixed discriminants and pushforward identities. 2021. arXiv: 2009.13107 [math.AG].
[FL83]	William Fulton and Robert Lazarsfeld. "Positive polynomials for ample vector bundles". In: Ann. of Math. (2) 118.1 (1983), pp. 35–60. ISSN: 0003-486X. DOI: 10.2307/2006953. URL: https://doi.org/10.2307/2006953.
[FP98]	William Fulton and Piotr Pragacz. Schubert varieties and degeneracy loci. Vol. 1689. Lecture Notes in Mathematics. Appendix J by the authors in collaboration with I. Ciocan-Fontanine. Springer-Verlag, Berlin, 1998, pp. xii+148. ISBN: 3-540-64538-1. DOI: 10.1007/BFb0096380. URL: https://doi.org/10.1007/BFb0096380.

[Ful97]	William Fulton. Young Tableaux, With Applications to Representation
	Theory and Geometry. Vol. 35. London Mathematical Society Student
	Texts. Cambridge University Press, Cambridge, 1997, pp. x+260. ISBN:
	0-521-56144-2; 0-521-56724-6.

- [Ful98] William Fulton. Intersection theory. Second. Vol. 2. Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Springer-Verlag, Berlin, 1998, pp. xiv+470. ISBN: 3-540-62046-X; 0-387-98549-2. DOI: 10.1007/978-1-4612-1700-8. URL: https://doi.org/10.1007/978-1-4612-1700-8.
- [GH78] Phillip Griffiths and Joseph Harris. Principles of algebraic geometry.
 Pure and Applied Mathematics. Wiley-Interscience [John Wiley & Sons], New York, 1978, pp. xii+813. ISBN: 0-471-32792-1.
- [Gie71] David Gieseker. "p-ample bundles and their Chern classes". In: Nagoya Math. J. 43 (1971), pp. 91–116. ISSN: 0027-7630. URL: http://projecteuclid. org/euclid.nmj/1118798367.
- [Gri69] Phillip A. Griffiths. "Hermitian differential geometry, Chern classes, and positive vector bundles". In: *Global Analysis (Papers in Honor of K. Kodaira)*. Univ. Tokyo Press, Tokyo, 1969, pp. 185–251.
- [Gul06] Dincer Guler. Chern forms of positive vector bundles. Thesis (Ph.D.)-The Ohio State University. ProQuest LLC, Ann Arbor, MI, 2006, p. 51. ISBN: 978-0542-77897-1. URL: http://gateway.proquest.com/openurl? url_ver=Z39.88-2004&rft_val_fmt=info:ofi/fmt:kev:mtx: dissertation&res_dat=xri:pqdiss&rft_dat=xri:pqdiss:3226339.
- [Gul12] Dincer Guler. "On Segre Forms of Positive Vector Bundles". In: Canad. Math. Bull. 55.1 (2012), pp. 108–113. ISSN: 0008-4395. DOI: 10.4153/CMB-2011-100-6.
 2011-100-6. URL: https://doi.org/10.4153/CMB-2011-100-6.
- [Har66] Robin Hartshorne. "Ample vector bundles". In: Inst. Hautes Études Sci. Publ. Math. 29 (1966), pp. 63-94. ISSN: 0073-8301. URL: http: //www.numdam.org/item?id=PMIHES_1966__29__63_0.
- [HK74] Reese Harvey and A. W. Knapp. "Positive (p, p) forms, Wirtinger's inequality, and currents". In: Value distribution theory (Proc. Tulane Univ. Program, Tulane Univ., New Orleans, La., 1972–1973), Part A. 1974, pp. 43–62.
- [Ilo78] Samuel A. Ilori. "A generalization of the Gysin homomorphism for flag bundles". In: Amer. J. Math. 100.3 (1978), pp. 621–630. ISSN: 0002-9327. DOI: 10.2307/2373843. URL: https://doi.org/10.2307/2373843.
- [JLP81] T. Józefiak, A. Lascoux, and P. Pragacz. "Classes of determinantal varieties associated with symmetric and skew-symmetric matrices". In: *Izv. Akad. Nauk SSSR Ser. Mat.* 45.3 (1981), pp. 662–673. ISSN: 0373-2436.
- [Kle69] Steven L. Kleiman. "Ample vector bundles on algebraic surfaces". In: *Proc. Amer. Math. Soc.* 21 (1969), pp. 673–676. ISSN: 0002-9939. DOI: 10.2307/2036445. URL: https://doi.org/10.2307/2036445.

[Kod54]	K. Kodaira. "On Kähler varieties of restricted type (an intrinsic char- acterization of algebraic varieties)". In: Ann. of Math. (2) 60 (1954), pp. 28–48. ISSN: 0003-486X. DOI: 10.2307/1969701. URL: https://doi. org/10.2307/1969701.
[KT15]	Hajime Kaji and Tomohide Terasoma. "Degree formula for Grassmann bundles". In: <i>J. Pure Appl. Algebra</i> 219.12 (2015), pp. 5426–5428. ISSN: 0022-4049. DOI: 10.1016/j.jpaa.2015.05.024. URL: https://doi. org/10.1016/j.jpaa.2015.05.024.
[Laz04]	Robert Lazarsfeld. <i>Positivity in algebraic geometry. II.</i> Vol. 49. Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics [Results in Mathematics and Related Areas. 3rd Series. A Series of Modern Surveys in Mathematics]. Positivity for vector bundles, and multiplier ideals. Springer-Verlag, Berlin, 2004, pp. xviii+385. ISBN: 3-540-22534-X. DOI: 10.1007/978-3-642-18808-4. URL: https://doi.org/10.1007/978-3-642-18808-4.
[Li21]	Ping Li. "Nonnegative Hermitian vector bundles and Chern numbers". In: <i>Mathematische Annalen</i> 380.1-2 (Mar. 2021), pp. 21–41. ISSN: 1432- 1807. DOI: 10.1007/s00208-020-01974-x. URL: https://doi.org/10. 1007/s00208-020-01974-x.
[LSY13]	Kefeng Liu, Xiaofeng Sun, and Xiaokui Yang. "Positivity and vanishing theorems for ample vector bundles". In: <i>J. Algebraic Geom.</i> 22.2 (2013), pp. 303–331. ISSN: 1056-3911. DOI: 10.1090/S1056-3911-2012-00588-8. URL: https://doi.org/10.1090/S1056-3911-2012-00588-8.
[LZ20]	Ping Li and Fangyang Zheng. "Chern Class Inequalities on Polarized Manifolds and Nef Vector Bundles". In: <i>International Mathematics Research Notices</i> (Dec. 2020). rnaa317. ISSN: 1073-7928. DOI: 10.1093/ imrn/rnaa317. eprint: https://academic.oup.com/imrn/advance- article-pdf/doi/10.1093/imrn/rnaa317/34753560/rnaa317.pdf. URL: https://doi.org/10.1093/imrn/rnaa317.
[Man98]	Laurent Manivel. Fonctions symétriques, polynômes de Schubert et lieux de dégénérescence. Vol. 3. Cours Spécialisés [Specialized Courses]. Société Mathématique de France, Paris, 1998, pp. vi+179. ISBN: 2-85629-066-3.
[Mou04]	Christophe Mourougane. "Computations of Bott-Chern classes on $\mathbb{P}(E)$ ". In: <i>Duke Math. J.</i> 124.2 (2004), pp. 389–420. ISSN: 0012-7094. DOI: 10.1215/S0012-7094-04-12425-X. URL: https://doi.org/10.1215/S0012-7094-04-12425-X.
[Nak55]	Shigeo Nakano. "On complex analytic vector bundles". In: <i>J. Math. Soc. Japan</i> 7 (1955), pp. 1–12. ISSN: 0025-5645. DOI: 10.2969/jmsj/00710001. URL: https://doi.org/10.2969/jmsj/00710001.
[Pin18]	Vamsi Pritham Pingali. "Representability of Chern-Weil forms". In: Math. Z. 288.1-2 (2018), pp. 629–641. ISSN: 0025-5874. DOI: 10.1007/ s00209-017-1903-2. URL: https://doi.org/10.1007/s00209-017- 1903-2.

- [PR97] P. Pragacz and J. Ratajski. "Formulas for Lagrangian and orthogonal degeneracy loci; Q-polynomial approach". In: Compositio Math. 107.1 (1997), pp. 11–87. ISSN: 0010-437X. DOI: 10.1023/A:1000182205320. URL: https://doi.org/10.1023/A:1000182205320.
- [Qui69] Daniel Quillen. "On the formal group laws of unoriented and complex cobordism theory". In: Bull. Amer. Math. Soc. 75 (1969), pp. 1293–1298. ISSN: 0002-9904. DOI: 10.1090/S0002-9904-1969-12401-8. URL: https://doi.org/10.1090/S0002-9904-1969-12401-8.
- [RT21] Julius Ross and Matei Toma. *Hodge-Riemann bilinear relations for Schur* classes of ample vector bundles. 2021. arXiv: 1905.13636 [math.AG].
- [Tu17] Loring W. Tu. "Computing the Gysin map using fixed points". In: Algebraic geometry and number theory. Vol. 321. Progr. Math. Birkhäuser / Springer, Cham, 2017, pp. 135–160.
- [Ume73] Hiroshi Umemura. "Some results in the theory of vector bundles". In: Nagoya Math. J. 52 (1973), pp. 97–128. ISSN: 0027-7630. URL: http: //projecteuclid.org/euclid.nmj/1118794880.
- [UT77] Sampei Usui and Hiroshi Tango. "On numerical positivity of ample vector bundles with additional condition". In: J. Math. Kyoto Univ. 17.1 (1977), pp. 151–164. ISSN: 0023-608X. DOI: 10.1215/kjm/1250522817. URL: https://doi.org/10.1215/kjm/1250522817.
- [Xia22] Jian Xiao. "On the positivity of high-degree Schur classes of an ample vector bundle". In: Sci. China Math. 65.1 (2022), pp. 51–62. ISSN: 1674-7283. DOI: 10.1007/s11425-020-1868-7. URL: https://doi.org/10.1007/s11425-020-1868-7.
- [Zhe00] Fangyang Zheng. Complex Differential Geometry. Vol. 18. AMS/IP Studies in Advanced Mathematics. American Mathematical Society, Providence, RI; International Press, Boston, MA, 2000, pp. xii+264. ISBN: 0-8218-2163-6. DOI: 10.1090/amsip/018. URL: https://doi.org/ 10.1090/amsip/018.
- [Zie18] Magdalena Zielenkiewicz. "Pushing-forward Schur classes using iterated residues at infinity". In: Schubert varieties, equivariant cohomology and characteristic classes—IMPANGA 15. EMS Ser. Congr. Rep. Eur. Math. Soc., Zürich, 2018, pp. 331–345.